

LSHTM Research Online

Estebanez Estebanez, Pilar; (2000) HIV risk behaviour in female injecting drug users in Madrid, Spain. PhD thesis, London School of Hygiene & Tropical Medicine. DOI: https://doi.org/10.17037/PUBS.01567778

Downloaded from: https://researchonline.lshtm.ac.uk/id/eprint/1567778/

DOI: https://doi.org/10.17037/PUBS.01567778

Usage Guidelines:

Please refer to usage guidelines at https://researchonline.lshtm.ac.uk/policies.html or alternatively contact researchonline@lshtm.ac.uk.

Available under license. To note, 3rd party material is not necessarily covered under this license: http://creativecommons.org/licenses/by-nc-nd/3.0/

HIV Risk Behavior in Female Injecting Drug Users in Madrid, Spain

**

Pilar Estébanez Estébanez

London School of Hygiene and Tropical Medicine

> Doctoral Thesis May 1998

Supervisor Dr. Klim McPherson

Abstract

A cross-sectional study was performed to explore the behaviour of female intravenous drug users (IDUs) in Madrid, Spain, to determine the relationships between behaviour and a number of dependent variables. We considered:

1.Sociodemographic characteristics Drug-taking behaviour, sexual behavior (including prostitution), reproductive health, and utilisation of services targeted at female IDUs.

2. Direct and indirect determinants of HIV in the study population

3. Determinants of potential risk factors for HIV

4. Attitudes among women towards safer sexual practices

5. The effects of different kinds of drug treatment on risk behavior

The limitations inherent in a cross-sectional study design, such as the time sequence of events, were addressed appropriately in the questionnaire. Because of the potential for selection bias, we implemented sampling techniques (such as sampling from multiple sites) to correct for this. Finally, in order to explore at different stages how the different variables might have a direct and indirect effect on the dependent variables and how the effect may be mediated through other variables in the causal pathway, we built a number of hierarchical models.

The foremost conclusion that can be drawn is the significance of low socioeconomic status. Furthermore, the high-risk population is homogenous and represents an already marginalised population. The result of regression analysis showed the most important and most predictive variable (OR=12.2) to be having a regular partner who is HIV-positive. Other significant factors were: age 35 or over; no fixed address; history of sharing needles; and co-infection with an STD. Early age of first pregnancy (younger than 18) was a direct determinant of prostitution and needle sharing. We also detected a link between sexual risk behaviour and drug-taking risk behaviour. The public health implications of these findings seems obvious. Women have a range of special needs, including reproductive needs. Efforts directed towards peer education for men and women are likely to be successful since the person delivering the intervention is someone the target population can identify with.

Acknowledgements

**

I would like to thank the following persons who were of invaluable help to me in the preparation and research of this thesis.

To my supervising professor, **Dr. Klim McPherson**, Professor at the University of London, thank you for the confidence you placed in me and in my work, for your continual encouragement, and your constant help in navigating my way through my second language.

To **Nancy Russell** and **Jonathan Grant**, who served as my advisors and editorial assistants throughout the project. Mark Twain wrote, "A powerful agent is the right word." Thank you for your insightful help in finding all the right words through these 200 pages.

To **Dr. Isabel Cifuentes** and **Ángeles Dopaccio**, who were my partners "in the trenches." Your unflagging efforts in compiling data and conducting interviews made this thesis possible and touched the lives of others in the process.

To María Dolores Aguilar and Victoria Zunzunegi, who helped turn the unruly results that field work always yields into the numbers and statistics that enable us to better understand the health concerns at hand and to envision policies to address the problems and questions these statistics helped formulate.

And last, but certainly not list, a thank you from my heart to my family, especially my two daughters, and to my friends, who share my life and who shared, too, in the process of creating this thesis with their continual encouragement, understanding, and love.

Table of Contents

1

Chapter One Background Information, the HIV/ AIDS Epidemic and Literature Review

28

<u>Chapter Two</u> Aim and Objectives of the Study

31

<u>Chapter Three</u> Methodology and Data Collection

46

Chapter Four Characteristics of Female IDUs

83

Chapter Five Determinants of HIV amongst Female IDUs

97

<u>Chapter Six</u> HIV, Drugs, and Prostitution: An Examination of the Determinants of Prostitution amongst Female IDUs

114

<u>Chapter Seven</u> Determinants of Previous Imprisonment amongst Female IDUs

130

<u>Chapter Eight</u> Determinants of Sharing Needles amongst Female IDUs

144

<u>Chapter Nine</u> Determinants of Never Used Condoms amongst Female IDUs

158

<u>Chapter Ten</u> Determinants of Never Received Treatment amongst Female IDUs

174

Chapter Eleven Conclusion and Policy Recommendations

> 184 Bibliography

List of Tables

Table 1.1 Notified AIDS cases in Europe for adolescents and adults by year of diagnosis a transmission group, reported by 31 December 1997, with adjustment for delayer notification.	
Table 1.2 Number of notified AIDS cases in Europe for adults and adolescents by year of diagnosis and sex, reported by 31 December 1997, with adjustment for delayed notification.	
Table 1.3 Accumulated AIDS cases in Spain by men and women and male:female ratio, without adjustment for delayed notification.	8
Table 4.1 Sociodemographic characteristics of the study population.	50
Table 4.2 Drug-use habits.	55
Table 4.3 Sexual relationship with regular partner.	60
Table 4.4 Sexual relationship with casual partner.	62
Table 4.5 Condom use with regular partner and sharing injecting equipment.	64
Table 4. 6 Condom use with casual partner and sharing injecting equipment.	65
Table 4.7 Sexual relationship with clients.	67

Table 4.8 Condom use with clients and sharing injecting equipment.	70
Table 4.9 Selected variables describing reproductive and sexual health.	75
Table 4.10Type of health services used by pooled female IDUs.	79
Table 4.11 Type of drug services used by pooled female IDUs.	80
Table 4.12 Type of social services used by pooled female IDUs.	81
Table 5.1 Selected variables describing HIV prevalence by country.	91
Table 5.2 HIV prevalence by selected variables.	92
Table 5.3 Hierarchical logistic regression of HIV on risk factors.	96
Table 6.1 Descriptive analysis of prostitution.	103
Table 6.2 Proportion of women reporting clients by selected variables.	104
Table 6.3 Hierarchical logistic regression of prostitution on risk factors.	113
Table 7.1 Proportion of women previously imprisoned by selected variables.	120

Table 7.2 Hierarchical logistic regression of previous imprisonment on risk factors.	129
Table 8.1 Proportion of women sharing needles by selected variables.	137
Table 8.2 Hierarchical logistic regression of sharing needles last 6 months on risk factors	14 3
Table 9.1 Proportion of women who never used condoms by selected variables.	149
Table 9.2 Hierarchical logistic regression of never used condom on risk factors.	157
Table 10.1 Proportion of women reporting never had drug treatment by selected variables.	169
Table 10.2 Hierarchical logistic regression of ever had drug treatment on risk factors.	173

Hierarchical logistic regression of ever had drug treatment on risk factors.

Chapter One

*

Background Information, the HIV/AIDS Epidemic, and Literature Review

1. Introduction

The relation of gender to health, gender to drugs and gender to human immunodeficiency virus (HIV) infection is an issue that probably depends greatly on the way society is organised. Although organisational structure may differ widely between countries, it is likely that many societies are characterised by major inequalities for women. Given this assumption, the principal objective of this study is to examine risk behaviour in female intravenous drug users (IDUs). The current chapter introduces some of the concepts pertinent to the thesis. Section Two describes some background information that led to the implementation of this study. Section Three presents data regarding the HIV and auto-immune deficiency syndrome (AIDS) epidemic and drug use in both Europe and Spain. Section Four reviews the literature on gender-specific risk factors associated with HIV/AIDS and IDU.

Women make up an increasing proportion of persons infected with HIV and AIDS. In some African countries, men and women have been infected in approximately equal numbers since the beginning of the epidemic. In North America and Europe, however, women originally made up a small fraction of those infected, but the female-male ratio has shown an ever increasing trend in recent years. In industrialised countries women now comprise the fastest growing group of persons infected with HIV and AIDS; a situation that has multiple repercussions including, for example, transmission to unborn children.

The main reason for the increase in heterosexual transmission of HIV among women is that they are more vulnerable to HIV than men. This vulnerability is not only linked to a greater anatomical susceptibility to HIV but also to social, cultural, economic and political factors. For example, one of the major risk factors associated with heterosexual HIV infection is the presence of sexually transmitted diseases (STDs). STDs are frequently asymptomatic in women, thus females may not seek treatment as often or in as timely a manner as men. In fact, it is only recently that the scientific community has begun to consider the possibility of gender bias in medicine caused by making incorrect assumptions about similarities between men's and women's health situations and risks or vice versa, about differences where there are none (Ruiz and Verbrugge, 1997). It is proposed that these incorrect assumptions have led to inequalities in knowledge about women's health problems, resulting in inequalities in treatment and prevention of diseases. Other factors also put women at a disadvantage in comparison to men, especially in developing countries, where women may have less access to information and health services, as they are economically dependent on men and they have little power in their sexual relationships with men.

The continuous increase in the number of cases of HIV and AIDS in female injecting drug-users (IDUs) is a worrying feature of the current situation. Moreover, it is thought that most HIV-positive women in western countries have been infected by sharing contaminated injection equipment. Until recently, investigations on drug-dependent women occurred infrequently. Data on men and women were usually merged as though there were no significant differences between

genders. Possibly for this reason, studies have not found gender-specific risks of contracting HIV infection, despite there being important lifestyle differences that make women more likely to engage in risk behaviour than men. Many of these lifestyle differences between men and women stem from the different roles played by men and women. For example, women may lack decision-making power in their sexual relationships. While male drugs users typically resort to drug dealing and robbery to finance their drug habits, it is often assumed that female drug users exchange sex for money or drugs. This risk behaviour could, for example, be one factor in determining women's increased vulnerability to heterosexual transmission. In addition, female IDUs have male sexual partners who are also IDUs, a fact which has great implications for HIV transmission, not only through the possibility of sharing injecting equipment, but also due to a women's unequal power in making decisions about sex.

Despite these problems, it is surprising that studies regarding HIV transmission and control have traditionally paid little attention to women. Many of the problems in HIV control strategies stem from a lack of knowledge about the epidemic in women, the development of fewer prevention measures aimed at this group and a lack of interest in the research about reproductive issues.

The lack of research interest in women has also occurred in the drug addiction field. Despite the central role played by female IDUs in the spread of the epidemic, there is also a relative paucity of research on gender-specific HIV risk behaviours. Gender is a key variable to be taken into account in attempting to understand any behaviour. An understanding to what extent women may differ in their risks from their better-studied male counterparts should enhance our ability to formulate better policies of prevention and control. Most research on IDUs' risk behaviour either has been conducted among male subjects or has been carried out with samples in which the proportion of women were so small that they were not considered as a separate group in the analysis. In general, gender has not been treated as a variable, thus it has been assumed that IDUs' behaviour and risk are gender neutral.

In fact, there are several reasons to think that female IDUs may have a higher possibility of acquiring the HIV infection than male IDUs. Some studies have shown female drug injectors almost never shoot drugs alone and that women almost always share needles when they inject drugs. Most women drug users are strongly influenced by men in their drug behaviours. For example, they often need someone else to inject the drug and this is usually the male who injects his partner after first injecting himself.

The role played by contraceptive methods such as the pill or intrauterine devices in enhancing HIV transmission is controversial. Despite the great impact of issues surrounding contraception on society, it is still unclear whether or not oral contraception might facilitate HIV transmission.

In order to understand women's drug behaviour, it is necessary to take into account certain social, cultural and economic characteristics of today's world. Such factors as job scarcity, social isolation and marginalisation could influence drug abuse, especially among women. Women are more frequently passive and dependent, have lower self-esteem, and may suffer from greater social marginality than male IDUs. Consequently this gender-specific behaviour can only be understood in the social and cultural context in which it takes place.

On the other hand, it has been suggested that patterns of entering into treatment may also differ between men and women, with male subjects predominating in both treatment clinics and research studies. Thus, to date, the possible special needs of women addicts have not been adequately assessed, and treatment programs have, to that extent, been hampered in their efforts to treat women.

Furthermore, a large proportion of prison entrants are IDUs who have committed crimes because of their need to obtain income to buy illegal drugs. The relation between HIV and prisons depends on the proportion of drug users among the prison population, and the prevalence of HIV among IDUs and risk practices among them. A number of risk behaviours for HIV transmission occur in prison including the injection of illicit drugs, unprotected sexual intercourse and tattooing with inadequately disinfected equipment. These factors potentially increase the risk of HIV transmission among women in prison.

In summary, risk factors for HIV infection may differ substantially between men and women, which implies that the epidemiology of HIV in IDUs may be gender-specific. Given the extent of HIV/AIDS, and the fact that there are not, at present, effective vaccines or therapies to fight the disease, there is an urgent need to know specific risk factors in different populations in order to assess and develop appropriate preventive and health education programs.

So far several questions have been posed, but few answered. Further research is particularly needed into the many problems that have not yet been addressed but concern the links between women, drug use and HIV. The present study is part of a European multicentre cross-sectional study on the HIV risk behaviour of IDU women. The purpose of the Spanish study was to answer some of those questions which have great implications on the public health of Spain. In total, 304 female IDUs recruited in Madrid (Spain) were interviewed about their drug use and sexual behaviour, their reproductive and gynaecological history, their experience of preventive interventions and their attitudes toward services (including health, drug dependency and social services). The results of this research, the knowledge of risk behaviour and the evaluation of the use of services are expected to lead to a better understanding of drug-addicted women, their needs and the type of intervention necessary to begin to meet those needs. Such findings could facilitate the design of more appropriate services for women and aid in the control of the HIV epidemic.

2. The HIV/AIDS Epidemic and Drug Use in Europe and Spain

2.1. The HIV/AIDS epidemic

By December 1997, an accumulated total of 11.7 million deaths caused by AIDS cases have been reported globally (UNAIDS and WHO, 1997). In 1997 alone 2.3 million AIDS related deaths were reported (one fifth of the total). This data, however, probably underestimates the true extent of the pandemic. For example, the World Health Organisation (WHO), estimates that there are 30.6 million people world-wide living with HIV/AIDS, of which 12.1 million are women and 1.1 million are children (under age 15). Furthermore it is estimated that there are 16,000 new cases daily. Of course, it should be noted that the grand majority of these cases (more than 90 percent) occur in the developing world.

In the WHO European region, the accumulated number of all notified AIDS cases (i.e., paediatrics, adolescents and adults) up to December 1997 was 204,858. Of the cases notified during 1997, nearly a third (31.6 percent) were diagnosed in Spain. Spain has consistently had the highest reported incidence of AIDS cases in Europe since 1990. Although the AIDS incidence rate among homo/bisexual men appears to be leveling off, it continues to increase among IDUs and heterosexual contact cases (see Table 1.1). However a recent age-period cohort analysis of AIDS in Europe suggested that if you look at the most recent cohorts of homo/bisexual men and IDU you find reduced HIV transmission, yet amongst young heterosexuals the epidemic is still expanding (Houweling, et al. 1998). There has also been a consistent increase in the number of AIDS cases amongst women (Table 1.2).

Up until December 1997, 7,778 cases of AIDS amongst children (i.e., younger than 13 years) had been notified in Europe; 37.6 percent had been transmitted via the maternal-infant route of which 36.9 percent of children were born to IDU mothers (European Centre for Epidemiological Monitoring of AIDS, 1997).

Of the 197,073 cumulative cases of adolescent/adult AIDS cases reported in Europe by December 1997, women account for 17.4 percent, increasing from 8.0 percent in 1985 and 13.0 percent in 1990 (European Centre for Epidemiological Monitoring of AIDS, 1997). Injecting drug use is the principal risk factor in female AIDS cases, with 50.0 percent of female cases notified by December 1997 occurring among IDUs, and 38.5 percent associated with heterosexual contact.

Table 1.1

Notified AIDS cases in Europe for adolescents and adults by year of diagnosis and transmission group reported by 31 December 1997^a, with adjustments for delayed notification

Heterosexual contact

DU

			•			
Year	N	percent	N	percent	N	percent
1991	6808	37.0	7480		2208	12.0
1992	7349	36.0	8246	40.8	2829	13.8
1993	7415	33.3	9092		3486	15.7
1994	7721	30.7	10738	42.2	4326	17.2
1995	7203	29.2	10414		4687	19.0
1996	5747	26.1	9462	43.0	4642	21.1
1997	3598	22.9	6497	41.4	3735	23.8

Homo/bisexual men

^aEuropean Centre for the Epidemiological Monitoring of AIDS, 1997

Table 1.2

Number of notified AIDS cases in Europe for adults and adolescents by year of diagnoses and sex reported by 31 December 1997^a, with adjustments for delayed notification.

Year	Year No. of cases		<u>Women (%)</u>
1991	18425	83.8	16.2
1992	20438	83.3	16.7
1993	22260	81.5	18.5
1994	25172	81.1	18.9
1995	24681	80.0	20.0
1996	21982	79.1	20.9
1997	15700	78.9	21.1

^aEuropean Centre for the Epidemiological Monitoring of AIDS, 1997

2.2. The HIV/AIDS epidemic in Spain

In Spain, a total of 48,989 accumulated AIDS cases have been notified up until December 1997, of which 19.0 percent were female adults/adolescents and 845 (or 1.7 percent) were paediatric. The most frequent category of transmission was IDU (64.3 percent), whilst 11.6 percent of cases were due to heterosexual contact and 1.6 percent due to maternal-infant transmission. In 1996 heterosexual transmission increased in importance to be the second most popular category (Centro Nacional de Epidemiologia Registro Nacional del SIDA, 1996). Amongst Spanish IDUs, the rising incidence in AIDS has been associated with specific birth cohorts, with rising incidence rates in the successive generations born during the 1950s, peaking in men born in 1962 and women born in 1964 (Castilla et al., 1997). Subsequent cohorts demonstrate a marked falloff in incidence for both sexes, possibly reflecting a tendency in new generations of drug users to avoid risky drug use practices. However, this decline appears to halt abruptly in men from the 1972 birth cohort onwards, which may reflect a return to risky behaviours among a new generation of drug users.

As in Europe, there has been a progressive increase in the number of AIDS cases amongst women which is illustrated by the decline in the male:female ratio of AIDS cases in Table 1.3. Amongst Spanish women, heterosexual contact accounts for 25.3 percent of transmissions and together with IDU accounts for 89 percent of all transmissions. Furthermore the percent of female cases reportedly resulting from heterosexual contact has increased steadily from 12.0 percent in 1986 to 34.6 percent in 1997. In other words in Spain and Europe, the current state of the AIDS epidemic has been characterised by the increasing proportion of women infected by HIV and AIDS.

Based on the increasing number of cases in female IDUs, it can be foreseen that this will be one of the major problem areas for controlling the spread of AIDS in Europe. Considering that heterosexual and perinatal transmission from female IDUs is also becoming a serious problem, it is evident that there is an urgent need for studies focusing on females in order to provide more appropriate tools to public health authorities and political leaders.

Table 1.3

•

Accumulated AIDS cases in Spain by men and women and male:female ratio,
without adjustment for delayed notification, 1981 to December 1997*

Year	Case	Men (%)	Women (%)	Male:female ratio
			•	
1981-84	70	92.9	7.1	13.1/1
1985	242	89.7	10.3	8.7/1
1986	731	84.0	16.0	5.3/1
1987	1796	83.3	16.7	5.0/1
1988	4024	82.8	17.2	4.8/1
1989	7117	83.1	16.9	4.9/1
1990	10936	82.8	17.2	4.8/1
1991	15379	82.5	17.5	4.7/1
1992	20303	82.2	17.8	4.6/1
1993	25644	81.9	18.1	4.5/1
1994	32753	81.6	18.4	4.4/1
1995	39496	81.3	18.7	4.3/1
1996	45522	81.1	18.9	4.3/1
1997	48790	81.0	19.0	4.3/1
Total	49989 ^b	81.0	19.0	4.3 /1

^aCentro Nacional de Epidemiologia Registro Nacional del SIDA, December 31, 1997.

^bThe total number of accumulated cases included 199 cases where year of diagnosis was not known.

2.3. The HIV/AIDS epidemic in IDUs

The prevalence of HIV infection amongst female IDUs is not well documented because little research has been carried out on prevalence and risk factors in female injecting drug users. As data from drug users are not collected or analysed by gender, women are under-represented in most studies. Risk factors for HIV infection differ substantially between men and women, and a rising prevalence of HIV among females is shown in most studies carried out in Europe and United States of America (USA), as can be seen in Table 1.1. It is surprising that few of the investigators discuss this data and that until recently little attention has been paid to this important problem.

Nevertheless, incidence studies also suggest that female IDUs have higher rates of HIV seroconversion than male IDUs (Solomon et al., 1993; Salmaso et al., 1991), whilst evidence available from the international literature also indicates that female IDUs have a higher rate of HIV (Chin, 1991; Hankins, 1989; PHLS, 1991; European Centre for the Epidemiological Monitoring of AIDS, 1992).

2.3.1. Addiction to IDU

As previously mentioned, up until December 1996, 39.3 percent of adolescent/adults notified AIDS cases in Europe were associated with IDU, thus making this the principal cause of transmission in the heterosexual population. In the heterosexual adolescent/adult population, 50.9 percent of accumulated AIDS cases in women (15,496) and 37.3 percent in men (55,325) were associated with IDU (European Centre for the Epidemiological Monitoring of AIDS, 1996). In other words, IDU is the principal risk factor for women with AIDS. The increasing number of AIDS cases in female IDUs is one of the major issues in Europe and, as noted elsewhere, drug addicts constitute the principal source of HIV transmission for heterosexuals (Des Jarlais, 1992).

2.3.2. The prevalence of HIV in IDU populations

It is not possible to provide an exact estimate of the prevalence of HIV infection within the IDU population. The majority of studies have demonstrated that, first, IDUs are not representative of the population *per se* and, secondly, there are geographic variations between and within different countries. It is also important to realise that the estimates of seroprevalence can depend on the type of investigation, the protocol followed in the study and where the sample was collected (i.e., AIDS clinic, drug-addiction treatment centres, on the street, etc.).

In Europe, IDUs account for the largest and an increasing proportion of AIDS cases. The reported prevalence of HIV among IDUs in several western European cities was found to be over 40 percent in the mid-1980s (Robertson et al., 1986; Rezza et al., 1994; Sopelana et al., 1991; Rebagliato et al., 1995). In a report on the HIV prevalence rates among IDUs in European countries from 1985 to 1995, Hamers et al. (1997) noted that decreases in prevalence rates have been observed in several cities, including Madrid (74 percent in 1986 to 59 percent in 1990). However, as these authors warn, declines in HIV prevalence should be interpreted with considerable caution, particularly in high prevalence settings, where this does not necessarily imply a reduction in new infections. Rather, declining prevalence rates may be due to differential loss (through cessation of injecting or death) whereby IDUs who were infected early in the epidemic have not been completely replaced by newly infected IDUs.

2.3.3. HIV prevalence rates in female versus male IDUs

In concordance with the increase in HIV and AIDS cases amongst women, the majority of investigations that have studied female IDUs have found higher levels of seroprevalence when compared to male IDUs, although some of these results have not been statistically significant. For example, in a study undertaken in Baltimore, United States of America (USA), between 1988 and 1989, IDUs were recruited into the study from drug treatment centres, health services and from the street. A quarter (24.1 percent) of the total study population (n=2921) was seropositive, with a seroprevalence rate of 25.9 percent amongst female IDUs versus 23.7 percent of male users (Nelson et al., 1991). Amongst active drug users (within the past year; n=2616), the HIV seropositive rate increased to 25.3 percent, with a non-significant gender difference (27.0 percent amongst men). It is, however, interesting to note that when the sample was stratified by sexual orientation, there was a statistically significant gender difference (Vlahov et al., 1990). In San Francisco (USA), however, a lower rate of seroprevalence (11.4 percent) was observed in a study of female IDUs (n=403) recruited in the community between 1991 and 1992 (Watters et al., 1994b).

Some European studies have also observed a higher rate of seroprevalence in female IDUs. A multicentre study in Italy in 1987, found a global seroprevalence rate of 33.2 percent amongst IDUs that were attending public centres for drug dependency. In this sample of 14,363 IDUs there was a statistically significant difference in the rates of HIV infection between female and male IDUs (40.3 percent versus 36.5 percent respectively) (Sasse et al., 1989). In a study in the following year, a North-South geographical variation was demonstrated with regards to the prevalence of HIV amongst IDUs (47.0 percent of respondents in the north Italy were HIV positive compared to 19.0 percent in the south). This gradient is probably due to the fact that the AIDS epidemic started in northern Italy. This study identified no gender differences: the rate of seroprevalence among 1,348 IDUs was 38.0 percent for women and 37.9 percent for men (Salmaso et al., 1991).

In London, Stimson et al. (1996) reported a reduction in HIV-1 prevalence rates amongst IDUs recruited from multiple sites. Between 1990 and 1993 the prevalence rate declined from 12.8 percent to 6.9 percent. They found no difference in prevalence rate by gender or in length of time of injecting drug use.

In Madrid, Spain, between 1985 and 1990, 60.0 percent of drug addicts admitted to a hospital detoxification unit were HIV-positive (Sopelana et al., 1991). Moreover the rate was 65.0 percent for the IDUs who accounted for 91.0 percent of the sample. However, a progressive, and statistically significant, decrease in the rate of seroprevalence was observed for the study population: 74.0 percent of drug addicts were HIV-positive in 1985 and 59.0 percent in 1990 (p < 0.05). Likewise, for IDUs, 74.0 percent were HIV-positive in 1985 compared to 59.0 percent in 1990 (p = 0.05). Finally, 67.0 percent of female addicts were HIV-positive compared to 65.0 percent of male addicts. However, the prevalence of HIV amongst IDUs recruited in Madrid may not be representative of the Spanish IDU population, in light of large regional variations that have been reported (Lardelli et al., 1993). This study also failed to demonstrate a statistically significant (i.e., p < 0.05) gender difference for HIV infection.

In Valencia, Spain, the HIV-1 seroprevalence rate observed in a sample of 2,687 IDUs recruited between 1987 and 1991 was 50 percent. No gender differences were reported with seroprevalence rates of 50 percent for men and 51 percent for women (Hernández-Aguado and

Bolumar, 1993). However, a study of 650 intravenous drug addicts admitted to a Hospital Detoxification Unit in Barcelona between 1987 and 1993 did find higher rates of HIV seroprevalence in females, especially when the time-period of addiction was shorter than 24 months (p < 0.05) (Egea et al., 1996).

In Austria, HIV seroprevalence was found to be 49.2 percent from a sample of users of an AIDS and drug dependency clinic (n=268) (Zangerle et al., 1992). There were no significant differences between female (56.5 percent) and male (45.5 percent) patients. There was, however, a tendency for the number of IDUs admitted to the drug dependency clinic to decline (72.2 percent in 1985 to 12.5 percent in 1990) with a progressive increase in the numbers using the AIDS clinic. Similarly, in Berlin, a progressive decrease in the global figures for seroprevalence between 1985 (25.0 percent) to 1992 (3.2 percent) has been reported for IDUs that were admitted for the first time to a clinic (Stark and Muller, 1993). The overall rate of seroprevalence between 1992 and 1993 was 19.1 percent amongst IDUs in Berlin, without any significant gender differences (23.1 percent of women versus 17.1 percent of men).

In Greece, between 1986 and 1989, data also suggests a decline (of around 10 percent) in the number of IDUs who presented themselves at an AIDS clinic who were seropositive. There was also a decline in the number of prisoners who reported being IDUs (of around 2 percent). For the IDUs who were not in prison, 3.8 percent of women and 2.2 percent of men were seropositive (Papaevangelou et al., 1991).

A number of other studies have also shown a higher rate of HIV seroprevalence in male IDUs than in female IDUs, although, as with a number of the previously cited studies, the differences were not statistically significant. For example, in a study of HIV seroprevalence amongst 473 IDUs attending a methadone maintenance programme in New York (USA), between 1985 and 1986, 41.2 percent of men were HIV-positive compared to 37.4 percent of women (Schoenbaum et al., 1989). Likewise, in 1990 in New Jersey (USA), global seroprevalence was reported to be 47.6 percent for drug addicts who had injected within the last six months and were not in treatment. Amongst men the rate was 48.8 percent and amongst women 43.2 percent (Freeman et al., 1994). It has also been observed that seroprevalence was lower (39.0 percent) amongst heterosexual IDUs of an Italian neighbourhood considered to be a low-risk area. In this study there were no statistically significant gender differences (41.0 percent of men were HIV-positive compared to 33.0 percent of women) (Serraino et al., 1991).

2.3.4. The size of the Spanish IDU population

It is estimated that there are 150,000-200,000 IDUs in Spain. Although it is difficult to know the exact rates, in the Community of Madrid the number of female drug users is estimated to be between 2000 and 3000.

A network of drug treatment programmes distributed throughout Spain offers services to nearly 50 percent of IDUs. In 1988, a surveillance system was set up by the Spanish government in order to gather information about drug users attending the general health care system and the public drug treatment centres. In 1995, among the 6,048 women who sought advice and treatment from the system, 2,736 were prior attendees (defined as individuals who had already been in treatment in previous years) and 3,059 were new entrants (defined as individuals seeking drug treatment for the first time). Specifically for the Madrid area, among the 717 women treated, 255 were prior attendees and 379 were new entrants (National Plan on Drugs, 1995).

3. Gender-Specific Risk Factors of HIV/AIDS amongst IDUs

It is likely that risk factors for HIV infection differ substantially between men and women. For example, there is evidence that in heterosexual populations, women have a greater susceptibility of contracting HIV infection via IDU and/or heterosexual contact. In addition to differential biological risk, women are also likely to be exposed to a greater risk of HIV infection from a range of sociocultural and economic factors. The remainder of this section reviews the literature on gender-specific risk factors associated with both IDU and the transmission of HIV/AIDS.

3.1. Sharing injecting equipment

A number of studies have demonstrated that the sharing of drug injecting equipment is a principal risk factor associated with the transmission of HIV in IDU populations (e.g., Marmor et al., 1987; Des Jarlais, 1992). That is, the re-using and sharing of injecting equipment, whether it is borrowed or lent, contaminated or inadequately sterilised, increases the likelihood of intravenous transmission between different IDUs (Freeman et al., 1994). In Berlin, for example, a study undertaken between 1984 and 1988, reported that 88.0 percent of 741 heterosexual IDUs interviewed had ever shared injecting equipment (Stark et al., 1990). Likewise, a study in Italy reported that 97.9 percent of drug addicts had intravenously injected drugs and that 70.8 percent of this sub-population reported to have at sometime shared injecting equipment (Salmaso et al., 1991). Moreover, Nicolosi et al. (1992b) demonstrated that the 69.6 percent of a sample of seronegative IDUs interviewed had shared needles and syringes. In Spain, the rates of sharing injection equipment appear to be lower. For example, in a nationwide sample of Spanish IDUs (n=1,074), 31.9 percent shared or had shared injecting equipment (Bravo et al., 1994). Furthermore, a large study of IDUs in Valencia (n=3,755) found that 18.0 percent reported recent sharing of injection equipment (in the last month) (Hernández-Aguado et al., 1994). In a multicentre Spanish study of prostitutes (n=1,556), 19.9 percent had injected drugs at some time and the majority of these had shared injecting equipment (Estébanez et al., 1992).

It has also been shown that the probability of infection is greater for those IDUs who frequently share injecting equipment when compared to those IDUs who never share (e.g. the odds ratio (OR) = 13.2 for Nicolosi et al. 1992b). Also in the study of IDUs in Baltimore, for those who had injected drugs in the previous year of the study, risk of HIV infection was associated with the sharing of needles and/or syringes (OR = 1.59; Vlahov et al., 1990), and especially with the use of injecting equipment attained in "shooting galleries" (OR = 2.21; Vlahov et al., 1990). Similarly, in Italy the principal risk factor of HIV for IDUs is the sharing of injecting equipment (OR = 9.5; Sasse et al., 1989). Serraino et al. (1991) make the same observation, that it is the sharing of injecting equipment that is the principal risk factor of HIV infection and an independent predictor of seropositivity (OR = 4.4; Serraino et al., 1991).

It has been suggested that for IDUs, women could have a greater chance of acquiring HIV than men, and that this may be associated with the practice of risk-taking behaviours. It seems that the proportion of female IDUs that frequently share equipment is greater than that of men. In an examination of sharing injecting equipment and sexual behaviour in a nationwide sample of Spanish IDUs (n=1074), gender was shown to be associated with needle sharing with females sharing significantly more frequently than males (42.7 percent versus 29.8 percent respectively) (Bravo et al., 1994). According to a study conducted in Madrid in 1990, in which 45.0 percent of

IDUs studied had used lent syringes in the previous six months, women shared injecting equipment more frequently than men (53 percent versus 42 percent respectively) (Estébanez et al., 1992). Sasse et al. (1989) in a study of 1323 IDUs have also shown that more women than men share equipment to prepare the drugs (58 percent of women versus 53 percent of men) and/or for injection (69.1 percent of women versus 61.6 percent of men). Furthermore, Freeman et al. (1994), found that 42.9 percent of female IDUs and 43.0 percent of males had, at least once, lent equipment and 18.9 percent of women and 25.8 of men had, at least once, borrowed equipment. However Hernández-Aguado et al. (1994) reported no gender differences in recent (within the last month) sharing of injection equipment (16 percent for men and 21 percent for women).

One of the major explanations for the tendency of female IDUs to share injecting equipment is that women often inject in groups and/or with their sexual partner. For example, it has been shown that female IDUs are more likely than men to be injected in the company of other people (Zunzunegui et al., 1993). In addition, for those IDUs that tend to inject with other people, women are more likely to share equipment than men, and are more likely to be injected by their sexual partner (Sotheran et al., 1992). In Madrid, it has also been observed that female IDUs who have a stable partner who is also an IDU, share their equipment more frequently than those whose partners were not IDUs (58.0 percent and 44.0 percent respectively) (Zunzunegui et al., 1993). Similar results have been reported in other studies examining risk behaviours of HIV amongst IDUs. For example, women are more likely than men to inject with their sexual partner or with friends, while men inject with a colleague or "running partner" or alone (Barnard, 1992). Furthermore, female IDUs share injecting equipment with their partner with a greater frequency than men and are more likely to be injected after their partner has injected (Freeman et al., 1994). According to the authors of this study, female IDUs would therefore have a greater chance of infection than male heterosexuals.

3.2. Prison

Although the use of drugs in jails and prisons is not generally permitted, there is evidence that drugs are available in the majority of prisons and it would seem that a large percentage of IDUs have reported to have continued their habit whilst incarcerated. Moreover, the ability to procure sterile injecting equipment in jails and prisons is reduced as the practice is prohibited. Consequently, the sharing and re-use of needles and syringes is a common practice between imprisoned IDUs. In a study carried out in Scottish prisons it was observed that 27.5 percent of prisoners had a record of IDU before arriving at the prison. In addition, 7.7 percent had injected at least once whilst incarcerated and of the 5.7 percent that had previously shared equipment 17.3 percent had continued with this practice in prison (Power et al., 1992). In a study of 145 female prisoners conducted at HMP (Her Majesty's Prison) Cornton Vale, Scotland, 46 percent of female inmates had injected drugs and 57 percent of these injector-inmates reported to have injected whilst in prison (Gore et al., 1997). In the United Kingdom, 27.4 percent of prisoners have reported to have injected drugs during their stay in prison (Turnbull et al., 1991). In Australia, over half of the needles and syringes found in three prisons in Adelaide seemed to have been utilised a number of times, suggesting that equipment sharing was prevalent (Seamark et al., 1994). In a multicentre study of seroprevalence conducted amongst 472 IDUs (of which 33.1 percent were women) in Berlin, 69 percent had been in prison (76.3 percent of the men and 53.8 percent of the women). In addition, almost half of the men (44.9 percent) and 21.2 percent of the women had injected drugs whilst in prison. This survey demonstrated a strong association between HTV infection and the use of shared needles for injecting during the stay in prison (OR = 3.9; Confidence interval (CI): 1.8 - 8.5) (Stark and Muller, 1993). Previous surveys in the same

city reported lower rates of injecting drug use in prison. For example, 63.2 percent of IDUs studied between 1984 and 1989 reported to have been in prison, of which 29.5 percent of men and 22.6 percent of women had injected drugs during their stay (Stark et al., 1990).

In a Canadian study conducted between 1988 and 1989, on seroprevalence and risk factors of HIV in women recently imprisoned, it was demonstrated that, when compared to an average seroprevalence of 6.9 percent in the sample population, 92.6 percent of seropositive women and 45.5 percent of seronegative women had a record of IDU prior to imprisonment (OR = 14.9; Hankins et al., 1994). Moreover, 76.2 percent of the seropositive women and 44.3 percent of the seronegative reported to have shared injecting equipment with a stranger before imprisonment (OR = 4.0), thus increasing the probability of infection by a factor of six when the person sharing the equipment was also seropositive.

3.3. Sexual behaviour

It has been suggested that the risk of acquiring HIV is the result of a complex interaction of drug use and sexual behaviour (Schoenbaum et al., 1989). Furthermore, as Freeman et al. (1994) point out, heterosexual female IDUs have an elevated risk of HIV infection from an infected partner, principally through sexual contact, but also as a result of shared injecting equipment as many of these women are injected by their partner after he has injected himself.

3.3.1. Heterosexual transmission

Heterosexual contact accounted for 14.9 percent of notified adolescent/adult AIDS cases in Europe up to 30 September 1997. However, this category of transmission differs between men and women. Heterosexual contact is the only category of transmission in which the proportion of cases of AIDS in women surpasses that of men (European Centre for the Epidemiological Monitoring of AIDS, 1996). More than a third (38.7 percent) of accumulated AIDS cases in adult and adolescent women, up until 30 September 1997, were attributed to heterosexual contact compared with 10.2 percent of men. A recent age-period cohort analysis of AIDS in Europe suggested that if you look at the most recent cohorts of homo-/bisexual men and IDU you find reduced HIV transmission, yet amongst young heterosexuals the epidemic is still expanding (Houweling, et al. 1998). Furthermore a national study of AIDS surveillance in the United States has also shown that the number of AIDS cases attributed to heterosexual contact is increasing at a faster rate than are the numbers of those acquired through other routes (Neal et al., 1997). These authors report that at this point in the U.S., persons with heterosexually acquired AIDS remain disproportionately women (65 percent) who report sexual contact with a male IDU partner (47 percent).

3.3.2. Heterosexual relations with an IDU partner

A number of studies have shown that heterosexual contact with an IDU partner is an important independent risk factor for HIV. Sasse et al. (1989) observed that sexual relations with a drug addict was associated with seropositivity with an estimated risk of 1.73. For the heterosexual sample (91.5 percent of the study population), HIV infection was most frequent among those that had had an IDU partner (OR = 1.34), with a further increase in risk for those who had had an increase in the number of partners. Similarly, a study by Schoenbaum et al. (1989) showed that

heterosexual contact with an IDU was an independent factor associated with HIV infection, and that this risk was higher for women. Evidence suggests that the importance of heterosexual behaviour as a risk factor is elevated by the number of partners and IDU partners and is more important for female IDUs than male IDUs. The probability of HIV infection for women increased by 24.0 percent with every additional IDU partner, demonstrating the increased HIV risk for women associated with heterosexual behaviour.

The possible impact of the sexual transmission of HIV by IDUs to the general population is reflected in a study undertaken in England and Scotland in 1988 to determine possible changes in sexual behaviour of IDUs that were attending a needle exchange service. It was reported that the majority of this population was sexually active and that 46 percent of users had had a non-IDU as a sexual partner. In addition 38 percent of the sample population had used condoms in sexual relations during the preceding three months. The authors estimated that, if it was assumed an average of two non-IDU partners for every sexually active IDU (69 percent of the 60,000 IDUs estimated in United Kingdom), then 82,800 non-IDUs were potentially at risk of acquiring HIV, in a three-month period, from an IDU partner (Donoghoe et al., 1989). In another study undertaken in England, for the 73 percent of the sample that declared heterosexual relations, over half (54.5 percent) of the non-commercial sexual partners (71.4 percent) than regular pairs (39.6 percent) (Rhodes et al., 1993). In Madrid, a higher rate (49.0 percent) of mixed IDU partners (i.e., one IDU and one non-IDU) was reported amongst stable partners (Zunzunegui et al., 1993).

It is thought that the probability of HIV transmission, via heterosexual contact from an IDU partner would be greater for women than for men, as various studies have demonstrated that heterosexual male IDUs are more likely to have non-IDU female partners. Murphy (1988) observed that among heterosexual IDUs who were attending a drug dependency clinic, 70.0 percent of male IDUs had a sexual partner who was not an IDU compared to 40.0 percent of female IDUs. In London, 78.0 percent of female drug users (heroin or cocaine) with a regular sexual partner stated that their partner had used heroin compared to 55.0 percent of male drug users (Powis et al., 1996). In New York (USA), 45.4 percent of male IDUs (almost all heterosexual) who were attending methadone maintenance programs had had a female IDU partner, compared to 57.0 percent of women who had had a male IDU partner (Schoenbaum et al., 1989). In Berlin, between 1984 and 1989, in a study of predominantly heterosexual IDUs, 85.2 percent of women and 75.5 percent of men reported to have had a sexual partner who was an IDU (Stark et al., 1990). In an American study, 42.1 percent of female drug addicts interviewed had had a male IDU sexual partner (Kline and Strickler, 1993). In a study of seroconversion carried out in northern Italy on a sample of 635 HIV seronegative IDUs, over half of the women studied had had a male IDU partner who was also seronegative, compared to 41.2 percent for men (Nicolosi et al., 1992b). Almost 69 percent of female IDUs questioned by Freeman et al. (1994) had a male IDU partner. According to the authors, the probability of having a heterosexual partner who was an IDU was two times greater for women than for men.

3.3.3. Heterosexual relations with a seropositive partner

Numerous studies have associated the transmission of HIV with a heterosexual relationship with a seropositive partner. For example, in Austria it was estimated that the risk of seroconversion for IDUs was greater if the stable partner was a seropositive IDU (Zangerle et al., 1992). Likewise, Stark and Muller (1993) observed that in a sample of 472 IDUs, HIV infection was associated with heterosexual relations with a seropositive partner (OR = 3.0; CI: 1.5 - 9.2). In a multicentre study reported by Salmaso et al. (1991), in which 97.9 percent of the population were IDUs, heterosexual activity with an HIV-positive partner or with a partner of unknown serostatus, was associated with HIV infection (OR = 5.6; CI: 1.6 - 8.9 and OR = 2.6; CI: 1.9 - 3.6 respectively). The authors also noted that women had a seropositive heterosexual partner more frequently than men (41.0 percent of women compared with 18.0 percent of men). The same relationship was found in another Italian study of IDUs in which having a sexual partner who was HIV-positive was independently associated with HIV infection (OR = 4.0; CI: 1.5 - 10.4) (Nicolosi et al., 1992b). Furthermore, more women (26.4 percent) than men (13.9 percent) had a seropositive partner. Thus according to these authors and as demonstrated in the previous section, women are more likely to have a seropositive partner than men, putting them at greater risk of sexual transmission of HIV than men.

3.3.4. Gender-specific susceptibility to HIV infection

Being exposed to a male seropositive partner is not the only gender-specific risk of HIV infection for women. For example, it has been suggested that women are two to three times more susceptible to HIV transmission from a man, than a man is from a women. Nicolosi et al. (1994a) carried out a case-control study to determine the relative efficiency of the male-female and female-male transmission for people only exposed to risk from having an HIV-positive partner. The risk factors for infection in women included: a high number of sexual relations (OR = 2.0), anal sexual intercourse with an AIDS infected partner and a history of STDs such as syphilis (OR = 5.2), and genital warts (OR = 6.2). The significant risk factors associated with seropositivity in male-female transmission were: vaginal intercourse more than twice per week (OR = 2.0), anal sexual intercourse (OR = 2.3), partners with AIDS (OR = 2.6), partner with fewer than 400 CD4 (OR = 2.2), history of STDs (OR = 2.9), vaginitis (OR = 4.5) and the use of the IUD (OR = 2.5). In the female-male transmission, significant associations with HIV infection were: vaginal intercourse more than twice per week (OR = 5.2) and an AIDS infected partner (OR = 7.8). This study lead the authors to conclude that relative efficiency of the male-female transmission is 2.3 times more efficient than that of female-male.

Further evidence for this hypothesis is available from a multicentre study, undertaken in Spain, on risk factors for the heterosexual transmission of HIV from men to women. The overall rate of male-female transmission was 16 percent, although no difference between male-female and female-male transmission was observed (although this may be due to the small number of partners studied for the female-male transmission). A significant association was observed for the following risk factors for the heterosexual transmission to a woman in a stable relationship: CD4 (OR = 4.1), history of STD in the woman (OR = 4.0), and anal sexual intercourse (OR = 3.1). The CD4 and the history of STDs are independent factors associated with the heterosexual transmission for man to woman. According to the authors, the risk of heterosexual transmission for a partner with a case index that is CD4 and in a woman that has a history of STDs and practiced unprotected anal sexual intercourse is almost 14 times greater than in a partner without these factors (Zunzunegui et al., 1992).

3.3.5. Use of condoms

The use of condoms amongst IDU heterosexuals is rare, especially with stable partners. Consequently, the use of this barrier method seems to be most frequent when a partner is not an IDU or when one member of the pair is known to be seropositive. The majority of drugdependent men and women (79.3 percent and 73.5 percent respectively) studied by Nicolosi et al. (1992b) stated that they never or almost never used condoms during heterosexual relations. In the study by Stark and Muller (1993) only 12.9 percent of heterosexual IDUs declared to have always used condoms during sexual relations with a regular or casual partner during the previous six months. Also, a multicentre WHO study observed that the use of condoms between IDUs was low, and that the majority of IDUs interviewed stated that they never used them with their regular partner, although the most frequent usage was with casual partners (WHO Collaborative Study Group, 1993). The same conclusion was arrived at by Rhodes et al. (1993) who showed that for seropositive IDUs, 73.2 percent never used condoms during heterosexual relations with a regular partner compared with 20.0 percent that stated they never used the same barrier method with a casual partner. Likewise, among sexually active heterosexual IDUs (n=315) interviewed in Madrid in 1990, only 22 percent were always or almost always using condoms during sexual relations, and more frequent use was reported with casual, as opposed to stable, partners (22.0 percent never used a condom with a casual partner and 49.0 percent with a regular partner). The most frequent use of condoms was reported for pairs in which one partner was a non-IDU (33.0 percent were always or almost always using condoms) (Zunzunegui et al., 1993).

In a study of injection equipment sharing amongst 3,755 IDUs in Valencia, Spain, the proportion of subjects who never used condoms for vaginal intercourse was 57 percent. These authors noted that sharing of injection equipment was associated with not using condoms for vaginal intercourse in both univariate and multivariate analysis (Hernández-Aguado et al., 1994).

In a cross-sectional study undertaken in North America amongst female drug addicts, 66.0 percent stated that they never had used a condom in spite of that fact that 46.2 percent had a sexual partner who was an IDU or an ex-IDU. Furthermore 47.5 percent of these cases did not know the serostatus of their partner (Kline and Strickler, 1993). In a study carried out in London, 59 percent of male heroin users and 65 percent of female heroin users stated that they never or almost never had used condoms during heterosexual relations in the past year. For cocaine users, condom use was marginally greater, although 47 percent of men and 46 percent of women stated they only rarely or never used a condom in the past year (Strang et al., 1994). A study by Freeman et al. (1994) also showed that the use of condoms was rare for heterosexual IDUs, especially among those with only one sexual partner. For example, 72.4 percent of men and 81.8 percent of women interviewed never used condoms with their partner. In the case of having multiple heterosexual partners, the female IDUs reported to use condoms more frequently than men, although only 17.6 percent stated that they always used them. According to Rhodes et al. (1994) the use of condoms is inversely related with the number of sexual relations. Paone et al. (1995) have also noted that female IDU participants in a syringe exchange program in New York City were more likely to "always" use condoms with casual partners than with primary partners. The authors also reported that 60 percent of prostitutes reported "always" using condoms.

Another aspect is that sexual behaviour appears to be safer with regular partners who know of their HIV seropositivity than with partners who are unaware of their serostatus. Sasse et al. (1989) observed in a sample of drug addicts attending a treatment centre that only 7.1 percent of the population and 7.6 percent of the heterosexual population, were always using condoms, and that their use was most frequent among those who were seropositive (c2 for trend = 35.07, p = 0.00001) and in heterosexual relations with a non-IDU partner (c2 for trend = 13.5, p = 0.001). Also in the study by Rhodes et al. (1993) IDUs that knew that they were seropositive were more likely to use condoms than those who did not (58.3 percent compared to 18.2 percent respectively). The risk of heterosexual transmission between IDUs and non-IDU partners is dependent, in part, on the knowledge of seropositivity and subsequent use of condoms.

In a sample of 317 IDUs collected in the street, 33 percent of heterosexual IDUs declared to have always used condoms with their partner in the previous month and a significantly greater number used them during sexual relations with a seropositive partner or a non-IDU partner. The use of condoms in relationships between seropositive IDU and non-IDU was 68 percent (Friedman et al., 1994). Another American study also carried out amongst female IDUs observed that the use of condoms was generally low: 34 percent of women stated that they always used condoms during sexual relations. For seropositive women, condom prevalence was 28 percent compared to 6 percent for seronegative women. For stable partners, 42 percent always used condoms, with 36 percent of seronegative women using condoms and 6 percent of seronegatives (Watters et al., 1994b).

Therefore the risk of sexual transmission of HIV is high for the partners of IDUs, especially for women and in particular for female IDUs. It has been suggested that for female IDUs the principal risk factor for HIV could be unprotected sexual activity (Watters et al., 1994b). Moreover, female IDUs have a greater risk of transmission through heterosexual relations with an IDU partner than from transmission via IDU itself (Schoenbaum et al., 1989). Rhodes et al. (1994) pointed out that the risk of heterosexual transmission for an IDU is progressive as injecting drug use would continue to be a particular risk. For example, 27.7 percent of IDUs with stable partners who were non-IDUs had shared needles at least sometime during the time of their study, and similarly 36.0 percent of IDUs with a casual non-IDU partner reported needle sharing. This would suggest a greater risk for women as male IDUs are most likely to have as a regular partner a non-IDU woman than vice versa.

3.4. STDs and prostitution

Sexually transmitted diseases, which are prevalent amongst those who engage in unprotected sexual activity, occur more frequently within the IDU population than the general population, especially amongst women. For example, 67.7 percent of IDUs studied in Berlin between 1984 and 1989, were positive for hepatitis B; a disease also strongly associated with HIV seropositivity (Stark et al., 1990). In a study of IDUs attending drug treatment in Dublin, gender differences were found with a greater proportion of women testing positive for the hepatitis C virus, the prevalence of which can be an indicator of HIV-related risk behaviour (Smyth et al., 1994). Furthermore, in a study of HIV seroprevalence carried out with IDUs in Baltimore, 60.0 percent of drug addicts had had some type of STD, with an increased frequency for female heterosexual IDUs (syphilis and genital herpes). According to the results of the statistical analysis, women with a record of syphilis would have a greater major risk of HIV infection than male heterosexuals with similar backgrounds (OR = 1.78; CI: 1.02 - 3.10) (Nelson et al., 1991).

The greater frequency of STDs amongst female IDUs is probably due to the more frequent exchange of sex for drugs and/or money when compared to men. Serraino et al. (1991) observed that for an IDU heterosexual population, women were more likely to be prostitutes than men. Likewise, according to an Italian study of seronegative IDUs, women would have exchanged sex for drugs or money more frequently than men (19.8 percent and 1.5 percent respectively) (Nicolosi et al., 1992b). In Berlin, one of the highest reported rates of prostitution among interviewed IDUs was recorded: 73.1 percent of women and 12.7 percent of men reported to have worked as a prostitute (Stark and Muller, 1993). Also, in London, female IDUs interviewed were more likely to be prostitutes than men (OR = 10.9; CI: 3.28 - 39.9) (Rhodes et al., 1994). Furthermore, three quarters of IDU women that participated in the study by Freeman et al. (1994) had exchanged sex for money in the preceding six months. According to the authors, the probability of being a prostitute for female IDUs was over five times greater than for male IDUs.

It is important to realise that, in some cases, the clients as well as the steady partners of prostitutes put these women at increased risk of HIV infection. In the study by Zunzunegui et al. (1992) 40 percent of seropositive men interviewed said that they had had commercial sexual relations with a prostitute. Although Nicolosi et al. (1992b) found that only 12.2 percent of HIV-positive males reported this type of sexual encounter (ranging from 22 percent in Milan to 2 percent in Turin). In Spain, it was shown that two thirds of IDU prostitutes interviewed had a stable partner who was a male IDU, compared with 8.6 percent of non-IDU prostitutes (Estébanez et al., 1992). According to Van Ameijden et al., (1994) 63 percent of prostitutes interviewed in Amsterdam had a stable partner, of which 28 percent were currently male IDUs. For Astemborski et al. (1994) female IDUs who had had over two sexual IDU partners were more likely to have been a prostitute with the greatest number (i.e., > 50) of customers.

Some of the studies undertaken amongst prostitutes in Europe and North America have shown that the relationship between prostitution and HIV infection is closely associated with the intravenous injection of drugs. In a study of HIV seroprevalence amongst prostitutes in nine European countries, 5.3 percent of the global population were seropositive, but HIV infection was more frequent amongst IDUs than non-IDUs (31.8 percent and 1.5 percent were seropositive respectively), and IDU was the principal risk factor for HIV (European Working Groups on HIV Infection in Female Prostitutes, 1993). According to the results of a Spanish study of female prostitutes, 60.2 percent of IDUs were positive for HIV-1 compared to 3.4 percent of non-IDUs. Consequently, IDU was the principal risk factor associated with HIV infection for prostitutes (OR = 47.6), along with unprotected sexual activity for non-IDUs (Estébanez et al., 1992). Amsterdam, the estimated seropositivity was 30 percent amongst prostitutes examined in the city, all of which had a record of IDU (Van Ameijden et al., 1994). In the study by Astemborski et al., (1994) in which the 39.2 percent of IDU women had exercised prostitution at least once in the previous 10 years, the overall seropositivity for HIV was 24.6 percent and the rate for IDU prostitutes who had had more than 50 customers was greater than those with fewer than 50 customers (47.6 percent versus 23.2 percent).

According to the results obtained from these diverse studies amongst female IDUs, prostitution and a history of STDs are independent risk factors with a strong association with HIV infection, providing evidence of the risk of sexual transmission of HIV in this population. Of the female prostitutes studied in Amsterdam between 1986 and 1992, the relative risk for STDs was 2.92, due to the high rate of STDs in seropositive women (92 percent) compared to seronegative women (57 percent) (Van Ameijden et al., 1994). In a study carried out in San Francisco amongst female IDUs it was shown that prostitution was independently associated with seropositivity (OR = 3.11) (Watters et al., 1994b). Also Astemborski et al. (1994), observed that for prostitutes, having more than 50 customers (OR = 3.0) and previous STDs (OR = 1.84) were independently associated with HIV infection.

On the other hand, in the study by Serraino et al. (1991), prostitution was not a predictive risk factor of HIV infection for the global population and it was only associated with infection when the analysis was conducted for women alone (OR = 3.1; CI: 0.9 - 10.1). Amongst imprisoned women in Canada with a history of IDU,' the strongest predictors of HIV seropositivity were sexual or needle contact with a seropositive person, self-reported genital herpes, and having had a regular IDU sexual partner, but it was not predicted by prostitution (Hankins et al., 1994). In the study by Watters et al. (1994b) on female IDUs, a record of syphilis was independently associated with HIV seropositivity (OR = 3.30; CI: 1.36 - 7.99).

The use of condoms by prostitutes varies according to the country studied. There is a lower use in the countries of southern Europe than in northern and central Europe (Estébanez et al., 1992). It has been suggested that IDU prostitutes do not seem to be a major cause of HIV transmission to the general population. Although clients of prostitutes can expect to be infected, it is the private sexual relations, where condom use is extremely low, that are associated with transmission (Van Ameijden et al., 1994). It seems to be the case that, although the use of condoms by prostitutes is generally low, frequency is greater during commercial sexual relations than during private sexual relations, despite the fact that these women have male IDUs as partners. Paone et al. (1995) found that close to 60 percent of women engaging in commercial sex work, in a study of IDU women participating in New York City syringe exchange programs, reported "always" using condoms. In a cross-sectional study carried out in 11 Spanish provinces, almost 50 percent of prostitutes included stated that they used condoms "always" or "almost always" during vaginal sexual intercourse with clients. IDU prostitutes were using condoms with customers less frequently than with non-IDU partners (p = 0.05). Without doubt, sexual relations with stable partners are also at risk as 96.3 percent declared that they never used condoms during oral sex and 90.2 percent never during anal sexual intercourse (Estébanez et al., 1992).

3.5. Early sexual experience

Early sexual experience, including childhood sexual abuse, has been linked to a number of factors which also have links to HIV risk. Studies of sequellae of childhood sexual abuse (victimization is reported more often by women than men) have identified associations with substance abuse (Briere & Runtz, 1988), prostitution (Zierler et al., 1991), low self-esteem (in Johnsen and Harlow, 1996), adult depression (in Johnsen and Harlow, 1996), risky sexual behaviour (Zierler et al., 1991; Johnsen & Harlow, 1996), younger age at first intercourse (Johnsen & Harlow, 1991), and early age of first pregnancy (Zierler et al., 1991). Furthermore, a study of early sexual experience (defined as becoming sexually active between the ages of 10 and 14) also noted links to risky sexual behaviour and STDs (Greenberg et al, 1992). A study of prostitutes versus "normal women" found that prostitutes had generally initiated sexual activity at a younger age and had a higher incidence of early involvement in abusive sexual encounters (incest and rape) (James and Meyerding, 1977). As Greenberg and colleagues suggest, early sexual experience (or age at first intercourse) should be used as a marker for identifying women at risk for STDs (and HIV) and a target for preventive interventions.

3.6. Factors determining reproductive outcome amongst female IDUs

As already mentioned above (Section 3.3.4), the efficiency of the sexual transmission of HIV will rely on the infectivity of the case index, the susceptibility of the partner and the type of sexual relation (Najera and Cols, 1990). However, the proceeding discussion is limited to the susceptibility of HIV transmission between heterosexual contact between men and women Women are more susceptible to infection due to a series of biological, anatomical and histological factors, that make them more vulnerable to heterosexual HIV transmission. For example, vaginal secretions are less infectious than semen due to the pH of vaginal acid which makes the vagina an unfavourable environment for HIV. On the other hand, the base pH of semen is not only more favourable for HIV, but the survival of the virus is prolonged in the interior of vagina following ejaculation (Voeller and Anderson, 1992). In addition, the deposited semen in the vagina suppress the proliferation of T cells, thereby permitting the survival of HIV in the female genital tract (Anderson and Hill, 1991). Furthermore, the time of exposure is prolonged, as the semen remains in the reproductive tract following sexual intercourse until it has been completely absorbed. The anatomical and histological differences between male and female genital organs also contribute to the increased susceptibility of women, as female organs present a greater surface area of contact than do the male organs and thus are more receptive to HIV. In men, the foreskin provides a protection from infection which, in the absence of ulcers or abrasions, would make the penis invulnerable to HIV (Polk, 1985).

Other additional factors that would influence the greater susceptibility of women would include: the presence of recurrent vaginitis (Nicolosi et al., 1994a) and other STDs (Zunzunegui et al., 1992) which, due to inflammation in the vagina and/or lesions, would facilitate infection (Holst et al., 1987); the use of the IUD (Nicolosi et al., 1994a; Zunzunegui et al., 1992) as this can produce chronic inflammation of the cervix and can cause micro-lesions after being inserted (Musicco et al., 1994); sexual relations during menstruation (Hellman et al., 1991; Green et al., 1990); anal sexual intercourse (Zunzunegui et al., 1992) due to the traumatic effect on the rectal mucus; and possibly the use of oral contraceptives. This last factor is debated as some authors conclude that oral contraception is associated with infection (Simonsen et al., 1990; Plummer et al., 1991), while others think that it provides a protective effect (Nicolosi et al., 1994a; Goedert et al., 1988; Mati et al., 1990).

3.6.1. Menstrual cycle

A number of studies have reported that female IDUs have irregular menstrual cycles. It is likely that high prevalence of irregular periods is caused by a number of interacting factors including the use of opiates (Bai et al., 1989) and other drugs, stress, weight gain (caused by obesity) and weight loss (caused by anorexia nervosa) (Symonds, 1987). It should, however, be noted that the cessation of menses, concomitant with the use of drugs (and opiates in particular), is due to amenorrhea and does not imply that a women is infertile (Ralph and Spigner, 1986). This has been illustrated by the reported increase in fertility among 201 opiate-dependent women on enrollment into a methadone maintenance programme in Liverpool, UK (Morrison et al., 1995).

Menstrual abnormalities have also been associated with HIV infection. In a study of 248 premenopausal HIV-infected women without AIDS and 82 HIV-negative women, the HIV-infected women were found to have an increase in amenorrhea (> 3 months) and in menstrual cycle intervals (> 6 weeks) and lower rates of premenstrual breast symptoms. Changes in menstrual function were also significantly associated with a past history of, but not current,

substance abuse, suggesting the possibility that socioeconomic factors rather than biologic effects of drugs may be responsible (Chirgwin et al., 1996).

3.6.2. Contraceptive efficacy

It has been suggested that IDUs have a lower uptake of contraceptives due, amongst other factors, to the (partly misguided) belief that opiates and other drugs cause infertility (Ralph and Spigner, 1986). The use of contraceptives, it is thought, is lower amongst female IDUs than for the population per se. For example, in a study of 115 heroin-addicted women from California, USA, 25.8 percent reported any type of contraception compared to 48.5 percent of a national sample (Ralph and Spigner, 1986). Likewise, in the study of 201 opiate users from Liverpool, UK, 21.4 percent of women reported condom use compared to 46.5 percent of young women attending British Family Planning Clinics (Morrison et al., 1995).

Efficacy, safety and acceptability are all factors which need to be considered when prescribing a contraceptive method. Further, these factors need to be matched to individuals and communities (Williamson and Joanis, 1994); that is whilst one method may be appropriate for the general population, it may be inappropriate for IDUs and other marginalised groups. Thus, in light of the high prevalence of hepatitis amongst IDUs, the contraceptive pill may be an inappropriate method. It has also been suggested that the use of hormonal contraception may be associated with a reduction in the use of barrier contraception such as the condom. Another complicating factor is the concurrent use of petroleum-based lubricants. A recent European study identified the use of petroleum-based lubricants to be a risk factor for HIV infection in non-IDUs (European Working Group on HIV Infection in Female Prostitutes, 1993). Another issue to consider is the "dual" use of barrier contraceptives. A number of studies have suggested a reduction in the rates of sexually transmitted diseases (STDs), including HIV, for women who use condoms with a spermicide (reviewed in Roddy, 1994). Whilst the evidence is not yet conclusive, it is important to stress that both methods must be used concurrently (Cain and Williamson, 1994). Finally, the IUD has been associated with HIV infection in a number of studies (reviewed in Carlin and Boag, 1995).

3.6.3. Pregnancy

A number of studies have concluded that female IDUs are as likely to have children as the general population (reviewed in Wells and Jackson, 1992), as illustrated by the 70.0 percent of ever-pregnant female IDUs reported from an American study (Brown, 1989). There are a number of reasons why female IDUs have a higher rate of pregnancy and, in particular, teenage pregnancy. For example, female IDUs often lead a somewhat "traditional" lifestyle with low levels of educational attainment and employment (Wells and Jackson, 1992). Self-esteem is probably low, and it may be perceived that childbirth may break this cycle. Another possibility is that female IDUs often experience irregular menstrual cycles resulting in infertile periods and many women then assume they are infertile and do not need to take precautions to avoid pregnancy.

A number of studies have suggested that pregnancy itself may be a risk factor for HIV infection. Although as yet there is no consensus, what cannot be disputed is the increase in the number of HIV-infected pregnant women. Indeed this is probably a reflection of the sociodemographic characteristics of HIV in women. In Europe, for example, 76 percent of AIDS-diagnosed females are in the 15-49 age group (European Centre for the Epidemiological Monitoring of AIDS, 1996). It is therefore important that policy makers and public health practitioners alike are aware of the effects of pregnancy on the natural history of HIV. It has been suggested that HIV pregnant women may be susceptible to atypical immunological changes and that this will facilitate the progression from HIV to AIDS. To date, studies examining immunological changes have produced contrasting results. More evidence, however, is available with regard to the effects of drug use. Whilst there seems to be no relationship between opiate use and disease progression, there is evidence of gender-specific difference in the progression rate, the interaction of pregnancy and/or IDU with HIV-infected women makes this likely. The confused state of knowledge with regard to the reproductive health of pregnant women demonstrates the need to examine the association between HIV/AIDS and pregnancy.

The effect of HIV seropositivity on pregnancy outcomes has also been examined. In general, most of these studies have looked at the effect on obstetric or neonatal complications (e.g. premature rupture of membranes, intrauterine growth retardation) and concluded that in industrialized settings the impact of asymptomatic HIV infection is negligible. A few studies have looked at the impact of seropositivity on reproductive decisions. One such study which compared the rates of reproductive events before and after HIV diagnosis in a cohort of women aged 20-34 years with HIV infection found that diagnosis of HIV infection in women produced a 44 percent decline in age-adjusted live-birth rates (Stephenson et al.,1996). The decline was attributed to increases in termination of pregnancy (50 percent), decreases in overall pregnancy rate and an increase in miscarriages. Thus the diagnosis of HIV infection may have a substantial impact in reducing live-birth rates which has significant implications for expanding HIV testing in women and highlights the need for better contraceptive support for HIV-infected women.

3.7. Sociocultural, educational, and economic factors

Sociocultural, educational and economic factors are likely to be negatively associated with IDU populations, thus creating difficulties in developing and implementing health education campaigns aimed at decreasing and modifying risk behaviours. In spite of the inherent risks of treating IDUs as a collective group, the sociodemographic characteristics most often observed in different studies have included: a low level of education (Vlahov et al., 1990) unemployment (Sasse et al., 1989; Salmaso et al., 1991); a low income (Vlahov et al., 1990); high levels of prostitution (Stark and Muller, 1993; Freeman et al., 1994; Stark et al., 1990; Van Ameijden et al., 1994; Astemborski et al., 1994); and, theft or drug dealing in order to finance a habit, thereby causing a high rate of previous imprisonment (Sasse et al., 1989; Sopelana et al., 1991; Serraino et al., 1991; Stark et al., 1990; Estébanez et al., 1992).

It has also been observed that the sociodemographic characteristics of IDUs differ according to whether the drug addict is receiving treatment or not. According to a study undertaken amongst an IDU population in Madrid, participants recruited in the street (as opposed to those recruited in treatment centres) were more likely to have less education, less employment, a record of previous imprisonment and were less likely to live with partners or family. The authors of this study suggested that IDUs recruited in the street were less socially integrated than those in treatment centres (Zunzunegui et al., 1990). This being the case, then social isolation will probably be associated with a greater vulnerability to HIV infection due to the lack of assimilation within preventive health programmes and continued risk-taking behaviour; all of which could lead in itself to a greater sense of social isolation.

Another indicator of low socioeconomic status and a high degree of social marginalisation is not having a fixed address. Although studies of homelessness in IDU populations are rare or don't exist, studies of homeless populations have identified elevated levels of HIV risk practices, including IDU, crack cocaine use, sexual relations with an IDU, exchange of sex for money or drugs, multiple sexual partners and inconsistent condom use (Fisher et al., 1995; Nyamathi et al., 1997).

Psychological factors associated with marginalisation, social isolation and low self-esteem, also play an important part in designing and implementing prevention strategies. A number of studies have detected high levels of psychological distress and depressive disorders among homeless women, drug-recovering women and drug-using prostitutes (Nyamathi et al., 1997; Fisher et al., 1995; El-Bassel et al., 1997; Alegría et al., 1994). Furthermore these women have been found to have partners who are also at high risk for emotional distress and risky behaviours (Nyamathi et al., 1997). Poor mental health combined with the negative impact of their closest sources of social support may undermine the motivation and ability to help themselves, whether in seeking and following through with drug treatment or in adopting safer sex behaviour.

Numerous studies of black or Hispanic populations, which have been historically marginalised, have shown significantly higher rates of HIV infection (Nelson et al., 1991; Vlahov et al., 1990; Schoenbaum et al., 1989; Lehman et al., 1994; Vlahov et al., 1991). Female IDUs may also be more likely to suffer greater social isolation than men. According to some studies, these women live alone more frequently than men, have more children under their care, and do not live with the family of origin (Freeman et al., 1994; Estébanez et al., 1992). This situation could drive these women to greater isolation and social marginalisation. Moreover, marginalisation could lead to prostitution, and thus to greater marginalisation and lower esteem. We have already shown that women are more likely to be prostitutes than men, while men are more likely to turn to robbery or to the sale of drugs (Stark et al., 1990). It has also been observed that amongst IDUs, more men than women have a principal source of income (Freeman et al., 1994), which would help integration into the wider society.

The importance of psychosocial factors has not been recognised with regard to gender-specific infection of HIV. For example, prevention campaigns have stated that the principle mode of HIV transmission is the use of shared injection equipment, and to a lesser degree the practice of unprotected sex. Consequently, drug addicts and the general heterosexual population, have avoided (or reduced) the sharing of needles and syringes whilst underestimating risk of unprotected sexual activity (Kline and Strickler, 1993). This could explain why IDUs have modified their risk behaviour with regard to sterilising or not sharing injecting equipment (Watters et al., 1994a; Donoghoe et al., 1989). Likewise, there could exist a certain tendency amongst women to under value the risk of sexual activity with an IDU partner, as opposed to sexual risk behaviour per se. We have already commented on the specific risk of sexual relations between a non-IDU female and an IDU partner. It may be the case that emotional factors including dependence and/or economics, reduce the capacity of a woman to introduce change into her sexual relations and that the social status of the partnership could influence the perception of the risk. In relations like these, women have to trust their partners with regard to current and/or past risk behaviour. In casual relations the reverse may be the case and the women would not necessarily trust the partner. It has been suggested that this would be an erroneous perception of risk, as it would impede these women from utilising condoms with their stable partner (Kline and Strickler, 1993).

The scarcity of dedicated studies into the social context of HIV risk factors makes it difficult to differentiate between global and gender-specific IDU behaviour. Women have a greater likelihood of sharing injecting equipment than men, as women are more likely to share in a group and use shared equipment from men. Although both have a greater probability of sharing equipment with their sexual partner than with other types of companions, women are more likely to inject with their partner (Freeman et al., 1994; Sotheran et al., 1992), which for women means that they are more likely to have a sexual relationship with an IDU partner (Murphy, 1988). Moreover, women are also likely to be injected afterward, thus further increasing the risk of transmission for these women (Freeman et al., 1994).

3.8 Access to and utilisation of treatment

There exists little documentation of the utilisation of drug dependency services by this population, although it seems that uptake is low. Of the 104 confirmed seropositive IDUs examined by Rhodes et al. (1993), 43.1 percent had never received help or treatment for drug dependency and 52.2 percent had not been formerly tested for HIV. It was also observed in London that 42.0 percent of heroin addicts and 75.0 percent of cocaine users had never received treatment for drug dependency (Strang et al., 1994). In a study carried out in Madrid amongst IDUs, only 42.0 percent of those interviewed in the street had received some type of help or treatment for drug dependency. Of these, 21.6 percent had received treatment in a municipal centre for drug dependence, 21.1 percent had received methadone treatment, 22.9 percent had been to a hospital detoxification unit, 22.9 percent had received ambulatory detoxification, and 30.0 percent had received care in the community (Zunzunegui et al., 1993). According to the results of a European study (conducted between 1989 and 1990 in 12 countries) on the knowledge, attitudes and beliefs of IDUs about measures taken for HIV prevention, the low utilisation of drug dependency services could be due to the low opinion of these services amongst IDUs. By comparison, the direct provision of sterile injection equipment was generally welcomed (Richardson et al., 1994).

Inequalities between men and women also occur with regard to attending services and to assimilating information. The inequality of access and/or utilisation of health services could explain the difference of survival according to a study following the progress of HIV-infected patients between 1990 and 1993 in North America. According to the results, women-irrespective of being IDUs- had lower survival rates than seropositive men, as women had a lower rate of utilisation, and access to, health services (Melnick et al., 1994). In another North American study examining the utilisation of health services by HIV infected people, it was observed that women and IDUs tended to use health services in an emergency. The authors suggest that IDUs, and women in particular, are slow to seek medical help until the problem becomes a medical emergency (Mor et al., 1992). Furthermore a multi-site U.S. study on access to care for drug abusers with HIV reported that multiple barriers in access to service exist for drug-using women (Weissman et al., 1995). The most important barriers were cost and waiting times. The authors also noted that significantly fewer women than men received pre- and post-test counseling and were advised to get medical services after their first positive HIV test.

According to the data recorded amongst drug addicts interviewed in Madrid in 1990, women had a lower rate of uptake into drug dependency services, although the differences were not statistically significant (Zunzunegui et al., 1993). About 50 percent of men and 40 percent of women in the study were recruited from drug dependency centres, suggesting a lower rate of access and/or utilisation for these women. Although data on the use of social services amongst these women was not recorded, it is probable that drug-dependent women have lower rates of access due, in part, to feelings of guilt and/or the fear that the state might consider removing their children.

Gender differences in the process and response to drug treatment have also been examined. As men have traditionally accounted for a higher proportion of those addicted to heroin and therefore a higher proportion of those entering treatment facilities, it has only recently been questioned whether treatment intervention methods and goals are too "male-oriented" and not appropriately geared for women. In general women are found to be more likely to drop out of treatment than men, though when they do carry through with treatment they have similar rates of success (reviewed in Anglin et al., 1987). Studies of the factors which contribute to positive treatment outcome for women, though few, have consistently indicated that a partner's involvement in entry into treatment and continued support in attending treatment are critical (reviewed in Anglin et al., 1987). In a study of behavioural differences between men and women during and after methadone maintenance treatment, it was noted that women may actually have higher motivation for treatment in that they took less time than men to seek entry into their first methadone maintenance treatment programme after addiction. They entered more often without direct legal pressure, fewer were discharged for negative reasons, and they complied to a greater extent with treatment regulations than men (Anglin et al., 1987). Yet they were no more likely than men to have positive treatment outcomes. Some of the factors observed to be associated with poor outcomes included: having a partner who is also a narcotics user; and a lack of social and vocational skills necessary to relate to and become assimilated into the non-drug-using world. Furthermore, studies of gender differences in treatment populations have indicated that women in treatment report more medical, psychiatric, family-social and employment problems than men (McLellan et al., 1992; Brown et al., 1993). Thus it has been suggested that drug treatment programmes need to modify their services to specifically address women's issues by taking into account the important role the partner plays in treatment, by addressing women's health and vocational needs, and by addressing child care needs.

Furthermore, a more recent study into the apparent gender paradox in drug treatment found that although women have more "risk factors" associated with drug-use relapse, they are no more likely to relapse than men, and found that women become more engaged in treatment specifically by participating more in group counseling (Fiorentine et al., 1997). They did not find support for the drug severity hypothesis (women are less likely to relapse because their drug-use histories are less severe) or the social support hypothesis (women are less likely to relapse because they have higher levels of social support, facilitating recovery). The authors attribute the differences in treatment engagement to gender norms concerning help-seeking, personal independence, strength, and control.

4. Conclusion

This chapter has sketched the outline of the epidemiological and sociological literature on IDU and particularly female IDUs, with reference to the HIV/AIDS epidemic. Despite its importance, very little research has been carried out in order to establish if female IDUs have additional risk factors and to increase our knowledge of this population so that more appropriate intervention programmes can be developed. In view of the problems outlined above, it is important to answer the following research questions:

- Does gender have an effect in the risk of HIV prevalence in IDUs?
- Are there any socioeconomic characteristics responsible for the low use of prevention measures?
- What support services are appropriate for female IDUs in order to limit the AIDS epidemic?

It is therefore the objective of this thesis to address these research questions. In the following chapter the formal objectives are presented. Chapter Three describes the methodology and data collection. Chapter Four presents the characteristics of the female IDUs interviewed for this study, including their sociodemographic characteristics, drug-taking risk behaviour, sexual behaviour, prostitution, reproductive health, and utilisation of services. Chapter Five examines the determinants of HIV seropositivity in this population. Chapters Six, Seven, Eight and Nine study the determinants of prostitution, imprisonment, syringe sharing, and condom use, respectively. Chapter Ten looks at the utilisation of social services and drug treatment amongst these women. Chapter Eleven provides a summary of all findings and presents concluding remarks on directions for future interventions and research with this population.

Chapter Two

*

Aim and Objectives of the Study

1. Introduction

Following the literature review in Chapter 1 (Section 4), three research questions were identified (Chapter 1, Section 5):

- Does gender have an effect in the risk of HIV prevalence in IDUs?
- Are there any socio-economic characteristics responsible for the low use of prevention measures?
- What support services are appropriate for female IDUs in order to limit the AIDS epidemic?

In order to address these questions it is appropriate to formulate the specific aim and objectives of the study (Section 2). In addition, this chapter introduces the methodologies used and will describe the background of the study, study design, sampling, field work, and data analysis. In so doing, the difficulties in recruiting a suitable representative sample of female injecting drug users (IDUs) from different research settings will be demonstrated.

2. Aim and Objectives of the Study

To study the social and behavioural determinants of HIV infection in female IDUs in Madrid in order to improve implementation of appropriate prevention programmes to control the AIDS epidemic in Spain.

2.1. Objectives

The general study framework was structured with the following objectives:

- 1. To describe the sociodemographic characteristics, drug-taking behaviour, sexual behaviour (including prostitution), reproductive health, and utilisation of services of female intravenous drug users in Spain.
- 2. To assess the direct and indirect determinants of HIV in the study population.
- 3. To identify determinants of potential risk factors for HIV.
- 4. To study attitudes among women towards safer sexual practices; and
- 5. To assess the effects of different kinds of drug treatment on risk behaviour among female IDUs.

2.2. Operational objectives

To achieve these objectives, the following operational objectives were structured.

2.2.1. Objective I

To describe the sociodemographic characteristics, drug-taking behaviour, sexual behaviour (including prostitution), reproductive health, and utilisation of services of female intravenous drug users in Madrid, Spain.

Specific objectives:

i) To provide baseline information about HIV risk factors in Spanish female IDUs so that changes in prevalence of HIV and risk behaviour can be measured over time.

- ii) To describe HIV and STD prevalence in the study population.
- iii) To describe the reproductive health of the study population.
- iv) To assess the extent of social service, drug treatment, and health service utilisation in the study population.

2.2.2 Objective II

To assess the direct and indirect determinants of HIV in the study population.

Specific Objectives:

- i) To evaluate the risk of heterosexual transmission of HIV infection amongst female IDUs.
- ii) To study the effect of injection practices on HIV prevalence in women IDUs.
- iii) To examine the role of social factors in transmission of HIV.

2.2.3 Objective III

To identify determinants of potential risk factors for HIV.

Specific objectives:

- i) To identify direct and indirect determinants of prostitution.
- ii) To identify factors associated with previous imprisonment.
- iii) To identify determinants of "needle sharing."
- iv) To identify determinants of "never-used condoms."

2.2.4 Objective IV

To study attitudes among women towards safer sexual practices.

Specific objectives:

- i) To study knowledge and attitudes towards measures to prevent HIV.
- ii) To study attitudes towards pregnancies.
- iii) To assess the extent to which women make decisions about sexual behaviour.

2.2.5. Objective V

To assess the effects of different kinds of drug treatment on risk behaviour in female IDUs. Specific objectives:

- i) To assess the effect of health care and drug services utilisation on HIV risk behaviour.
- ii) To identify determinants of having "never attended treatment."

Chapter Three

 $\mathbf{\dot{v}}$

Methodology and Data Collection

1. Background to the Study

The data used in the thesis was collected from a sample of 304 female IDUs recruited in Madrid at the time of interview. This sample is a subset of a five-centre study (Paris, Madrid, Rome, London and Berlin) funded by the European Union and coordinated by the author of this thesis (Estébanez et al., 1996).

2. Study Design

An observational cross-sectional design was chosen as the most appropriate way to fulfill the objectives of the study (i.e., to estimate the behaviour of female IDUs and to explore the associations between behaviour and a number of dependent variables, such as HIV positivity, prostitution, prison record, etc.). However, there are a number of limitations to these methods which need to be addressed in order to place the results in the appropriate context. (Hennekens and Buring, 1987).

First of all, a cross-sectional study is carried out at one point in time; therefore it only measures the prevalence and not the incidence of a disease. Thus the influence of certain risk factors which have been modified over time cannot be measured with this design. This limitation was lessened because we designed the questionnaire to assess past behaviours using a six-month recall period and dichotomous variables, such as "ever-shared injection equipment" versus "never", as advocated by McElrath (1994). Secondly, there is the issue of the consistency of behaviour over time. Whilst these two problems theoretically could be overcome with a prospective cohort study, the recruitment and retention of a sufficiently large sample of initially seronegative IDUs would be extremely difficult. Finally, a limitation which applies to all studies of IDUs is that of the representativeness of the sample. Due to the characteristics of the population studied, and their inaccessibility (due to the illicit and clandestine nature of drug injecting), random sampling methods cannot be used leaving a potential for selection bias. To minimize this bias we recruited subjects from a variety of settings (for more details see Section 6).

3. Study Sample

3.1. The social context of the study

Madrid, like many other Spanish and European cities, underwent an epidemic of heroin use in the early 1980s. The use of illicit drugs, particularly heroin and cocaine, remains high, with police and drug agencies reporting an increase in the availability of heroin and cocaine in the 1990s

(National Plan on Drugs, 1995). The economic recession in the late 1980s and the geographical position of Spain as a bridge for drug trafficking into Europe seems to have increased IDU in the areas surrounding Madrid including, for example, the Chabolismo slums, Gypsy populations and marginalised groups. The increased realisation of this problem by society due, in part, to the AIDS epidemic has resulted in government and non-government organisations (NGOs) developing an extensive network of drug services in the region.

3.2. Population

Based on estimates from the Plan Nacional de Drogas, in 1996 there were approximately 150,000-200,000 IDUs in Spain, with about 15,000-20,0000 in Madrid. Although exact figures are not available, it is estimated that in the Autonomous Community of Madrid the number of female drug users is between 2,000 and 3,000.

.

3.3. Profile of the study population in Spain

In the absence of a recent population census or a study of IDUs that could provide a sampling frame for Madrid, we used a surveillance system set up by the Spanish government in 1988 in order to gather information about drug users attending the general health care system and the public drug treatment centres.

3.4. Network in health and social services

In order to obtain the support of recruitment centres and health and social services, we studied the current system for the provision of national health services, with particular regard to drug treatment. This information was collected in 1995.

3.4.1. Health services

The current Spanish public health system (as of 1995) can be divided into the following administrative levels:

- Central level. The Ministry of Health & Consumer Affairs manage public health at the central level.
- Autonomous Community. Autonomous Communities have control over health promotion and prevention. The responsibility for health care is assumed by the autonomous communities rather than by central government.
- Health areas. Care is provided for 200,000 250,000 people. There is at least one health area per province. According to the Health Law, health areas should carry out their activities by working in teams in order to serve individuals, families and communities, through programmes for health promotion and prevention, treatment and rehabilitation.

The health areas are divided into basic health zones that constitute a national framework for primary health care. The Social Security system provides health care to 98 percent of the population of the Autonomous Community of Madrid. Health services act in the health areas of this Community at three levels:

- **Primary health care**. Health reforms in the 1980s led to the creation of Health Centres. Interdisciplinary teams work on the diagnosis, treatment and cure of diseases using specific protocols, and in community health through prevention and health promotion programmes.
- Specialised health care. Health care needing specialised professionals, technology and treatment can be carried out in outpatient centres with specialised surgeries, laboratories and diagnosis, supporting both the specialised level and primary health care.
- Hospitals. In some areas there are hospital day units for specific diseases such as AIDS and cancer.

General health care for drug users should be carried out by a doctor at the primary health care level. Nevertheless it is well known that drug addicts are usually reluctant to visit a doctor and prefer direct access to hospitals through casualty services.

3.4.2. Drug treatment services

The Regional Plan on Drugs is a General Programme, an assistance network, aiming to provide a response to the drug dependency problem on three different levels of intervention: prevention, assistance-rehabilitation and social reintegration. This network of drug treatment programmes distributed throughout Spain offers services to nearly 50 percent of IDUs. It is targeted at both female and male drug users that either want to stop taking drugs or don't want to or are not able to give them up. During 1995 only 20 percent of the total number of individuals assisted by this Plan were women. Services are all free of charge.

General intervention resources could be summarised as follows:

- Council centres and day centres. Local Centres that offer prevention, social rehabilitation and reintegration for drug users. Treatment is also offered.
- Detoxification units. Physical detoxification may be carried out as an outpatient in a Day Centre/Council Centre or as an inpatient in a hospital Detoxification Unit. In the Region of Madrid there are only 2 Units with a total capacity of 22 beds. Access is restricted to pregnant drug users, those with a cardiovascular pathology or another serious illness. A patient must be referred by a Local Centre and must be over 16 in order to gain access.
- Therapeutic communities. This service aims to offer individuals, through a wide counselling programme, the personal tools needed to complete their rehabilitation in an outpatient Local Centre. Access is only for those drug users with a minimum age of 18 referred by a Local Centre.
- Methadone programmes. This programme is extremely limited and is offered only to pregnant drug users, to drug users with a serious organic pathology, and to individuals highly adapted to drug consumption. Access to this programme is limited to drug users referred by a Local Centre and a minimum age of 18 is required.
- Isidro bus. A service designed to reach drug users on the street in Madrid. This service provides information on health and social resources, injecting equipment, health advice, etc.
- Sitade. A twenty-four-hour phone line that provides information on the programmes offered by The Regional Plan on Drugs.

3.4.3. Social services

The current social and community services available are:

- **IMI (Minimum Integration Income)**. This programme has been devised to fulfill the basic needs of those people who have social problems and have no income. The philosophy of the Programme is to give economic support to those prepared to commit themselves to beginning a specific Programme to deal with their problems.
- Women's Services. This is a social welfare service of the Ministry of Social Integration of the Region of Madrid. The main aim of the Service is the total integration of women who are at risk and in marginal positions. The following programmes form part of this service:
- **Programme for abused women and their children**. Legal advice about women's rights and duties when they have been victims of some kind of abuse. Admittance and shelter at a residence centre, when needed. Counselling and treatment.
- Programme for pregnant young women and for single mothers and their children.
- Programme for young women with personal and/or family problems. The first goal is to give women an integral training including: basic educational training, sexual and sanitary information, and work training. Help is also given to drug users.
- **Programme for prostitutes**. This programme is focused mainly on giving financial support for the maintenance of a Centre where street prostitutes can take a break from their work, get condoms and syringes, talk with a social worker, etc. However, access to this programme is difficult for IDU women.

The Council of Madrid also provides financial help with school dinners, domestic help, and books for children.

3.5. Criteria for Inclusion

The study population comprised women who had:

- (1) injected drugs for more than one month and in the six months prior to the date of interview;
- (2) decided to participate in the study after being informed of its objectives; and,
- (3) given their informal (oral) consent to be interviewed.

4. Sampling

One of the main difficulties in preparing this research is a common problem for research on IDUs; that is the lack of a sampling frame for the population under study. The precise number of IDUs in Madrid is unknown, hence it was necessary to employ non-random methods of sampling It has also been suggested that, given the general inaccessibility of IDUs, convenience samples should be accepted. Another problem arises from the degree to which users from one setting differ from users in other settings with respect to, for example, demographic characteristics, risk practices and the prevalence of HIV infection. For example, Powis et al. (1996) noted a higher proportion of women in their sample of drug users recruited from community settings when compared to the proportion normally found among treatment populations. Differences were also illustrated by Lampinen (1992) who identified a two-fold risk of HIV seropositivity among community-recruited IDUs compared with entrants to drug-treatment programmes. However Alcabes et al., (1992) has found that the correlates of HIV infection in IDUs are not substantially different when comparing subjects currently in treatment and those not in treatment. Nevertheless, since it is the aim of this study to provide descriptive as well as bivariate and multivariate findings, we decided to recruit in different settings, as has been recommended in the international literature (Stimson, 1992), while attempting to control for differences between settings statistically. In summary, after reviewing the Spanish literature, looking at previous studies commissioned by the European Union and talking with professionals in this area, we decided to recruit the sample in three different settings: treatment centres, methadone centres, and outreach programmes.

4.1. Treatment centres

In 1992, among the 6,048 women who sought advice about and/or treatment for drug use, 2,736 were prior attendees (defined as individuals who had already been in treatment in previous years), and 3,059 were new entrants (defined as individuals seeking drug treatment for the first time). For the Madrid area, among the 717 women treated, 255 were prior attendees and 379 were new entrants.

In the emergency services of Madrid in 1992, 1,718 women were treated for drug use problems, of whom 31 died.

4.2. Methadone centres

Methadone programmes at the time of the study (1995) were scarce and overly restricted due to controversies in health policy. At that time, there were 50 female drug users in methadone programmes and 1,565 in day-centres run by the "Community of Madrid".

4.3. Outreach programmes

Needle exchanges are administered by NGOs. In 1993, 437 female IDUs used the needle exchange buses which tour the area of Madrid. In addition, about 14 women IDUs started treatment in the project named "HOMBRE", and 102 in specialised detoxification centres.

4.4. Recruitment strategy

The current sample of 304 women IDUs who completed questionnaires were recruited as follows:

Drug dependency units

One hundred and seven (35.2 percent) were recruited from drug dependency units. This sample was taken from different services: "Proyecto Hombre", "CADs", and "ISIDRO BUS".

- "Proyecto Hombre". Fifty-nine women were approached, but only 26 fulfilled the criteria of injecting drugs in the last six months. Of the 26 women eligible, 1 refused to participate.
- Governmental Institutions: "CADs" and "ISIDRO BUS". Eighty-two eligible women were recruited by the staff at these centres.

Methadone services

Fifty (16.4 percent) women were recruited from different methadone units: governmental and non-governmental. Many of the women approached were currently engaged in a methadone programme and therefore not currently injecting (in the last six months), so a high proportion were not eligible (105 of 157 approached), and two of them refused to participate. Thirteen questionnaires were completed at the "Centro de Metadona Cruz Roja de Fucar", 32 at the "Centro Metadona Cruz Roja de Marroquina", and 5 at the "Servicios de Metadona de los CADS de San Blas y Tetuan".

Outreach Services

One hundred forty-seven (48.4 percent) were recruited from outreach services, including the "Médicos del Mundo" bus scheme for needle exchange. Of the 250 women approached, 95 were ineligible for having begun to inject too recently or failing to complete the questionnaire, and 8 refused to participate.

Thus, overall, 11 women out of 315 (3.5 percent) refused to participate in the study. We have no information about these women, but since the refusal rate was so low, we feel this will not bias our results. It should be noted that the refusal rate was higher among women recruited through outreach programmes versus regular treatment centres. We think this is due to the fact that those in treatment centres are more likely to have an established relationship with the staff and therefore feel less threatened by the interview. Furthermore, we believe we were able to attain such a low refusal rate because the interviewers hired for this study had a great deal of experience in working with drug users.

5. Data Collection

Two forms of data collection were used. A questionnaire was administered in a face-to-face interview to all eligible females, and a specimen of blood was collected to assess the HIV status of all who consented and had not been recently tested. Those who had been tested were asked to present their medical card which indicates their HIV status. These cards are issued by the medical services, and many drug users carry them to avoid problems with the police.

5.1. The questionnaire

The questionnaire used in this study for assessment of risk behaviour and use of services was based, with modifications, on: (1) that used in another study in Europe for males; and (2) national health surveys of women in Spain. The interview was checked in the pilot phase of the study for consistency and reliability in the five centres of the European study.

The questionnaire was geared towards IDUs by using a theoretical framework based on plausible biological risk characteristics about transmission of agents by parental and sexual routes, and by ethnographic observations.

In spite of the limitations of retrospective self-report on behaviour relating to sexual practices and drug use, this method has been evaluated in numerous studies. It is accepted that the use of retrospective self-report is a valid method for assessing HIV risk behaviour. We have chosen a recall period of six months based on the work of McElrath (1994) whose findings supported the use of retrospective self-reported data although there was moderate disagreement between the retrospective and the prospective data sets by the six-month follow-up (memory decay). These authors suggest certain strategies for maximizing the reliability of retrospective data, including the use of dichotomous measures of risk behaviour, careful construction of research instruments, a thorough training period for interviewers, and specific methods to enhance subject's recall.

Another concern in asking questions related to sexual behaviour is the reliability, depending on who asks the question and how the respondent reacts to such questions. Due to the very intimate nature of the questions asked, interviewers were trained to take care of these issues together with subcultural aspects, such as the slang used by these women (Pickering, 1988).

5.1.1. Subjects of the questions

After encountering problems with the pilot questionnaire about the length of the interview, wording of some questions and the assurance of confidentiality, the final questionnaire was structured as follows:

- Sociodemographic characteristics. Questions related to age, marital status, educational attainment, etc.
- **Drug use**. Our emphasis is on unprescribed drug use as reported by respondent, covering the following types of drugs: heroin (and other opiates), cocaine (and crack), amphetamines, barbiturates, benzodiazepines, ecstasy, hashish, alcohol and tobacco. There are also questions related to recency of consumption, injection drug use, frequency of injection and risk behaviours, including sharing of injection equipment, cleaning equipment, and injecting in prison.
- Sexual behaviour. Questions related to sexual relationships with regular partners, casual partners and clients. Number of HIV-positive partners and IDU partners. Frequency of oral sex, vaginal and anal intercourse. (Criteria for regular or casual partner and client were included in the questionnaire.)
- Use of condoms. Questions to achieve a better knowledge of the determinants of the use of condoms by this population, such as: reasons to start using condoms; sources of condoms; and frequency of condom use with regular and casual partners and clients during oral sex, vaginal and anal intercourse.
- **Reproductive health**. Questions including age at first period, frequency of periods, contraceptive methods used, pregnancy, number of miscarriages and abortions, number of children, smear test, STDs, HIV-positive children, and HIV tests.
- AIDS public education campaigns. Questions included knowledge of public health campaigns, etc.
- Women's experience of health, drug and social and community services at local levels. These questions were related to frequency of use and reasons for using each service, and to women's perception of the usefulness and the accessibility of each service. Measurement of service utilisation is scored on a series of items covering encounters with three different types of services - health, drug, and social. There were multiple service utilisations, so this will represent the total number of different types of service systems utilized during the previous six months.
- HIV test. At the end of the interview, study participants were asked if they had been tested for HIV. If they responded that they had been tested for HIV, they were asked to provide a written confirmation of their test results by their doctor. It should be noted that it is common in Madrid for the medical services to issue a medical card indicating HIV status and that it is common for drug users to carry these cards in order to avoid problems with the police. Those who had not been tested and who gave their informed consent were asked to provide a blood sample for HIV testing.

5.1.2. Formatting and coding of variables

Data were collected on more than 84 sociodemographic, behavioural, and health variables. All centres used the same protocol for sampling and administration and the same core questionnaire. Questionnaires were coded to ensure the anonymity and confidentiality of the study participants

In the pilot study some difficulties were met with in the content and format of some types of variables, and after discussion with the collaborators these variables were reclassified using continuous (such as the number of sexual relationships, number of injections) and categorical (such as dichotomies (yes/no)) variables. However, it should be noted that the level of detail required by continuous scales may exceed the capabilities of some participants and for this reason some responses are categorised using, for example: "always", "very often", "almost never," and "never".

5.1.3. Types of variables

- Qualitative and categorical variables
 - Demographic characteristics (i.e., sex, marital status, educational attainment, residence, source of income, etc.).
 - HIV prevalence.
 - Probable risk factors (i.e., prison, drug practices and variables concerning sexual behaviour).
 - The use of health services and social networks, and variables about knowledge, attitudes, access and acceptability of these services.

For the multivariate logistic regression analysis we used different scales, including dichotomous, nominal, and ordinal variables. Dummy variables/reference categories were assigned as the group with the least risk.

• Quantitative variables

- Continuous variables (i.e., age, age of first injection, age of first period, etc.).
- Discrete variables (i.e., parity, number of children, abortion, frequency of sexual relationships, frequency of injection, etc.).

Continuous variables were measured in the appropriate scale (i.e., years, etc.). Some of the continuous variables were recoded into categorical variables for the multivariate logistic regression analysis.

5.2. Testing for HIV status

HIV antibody testing was performed only after informed consent had been given. Refusal to be tested was not grounds for exclusion from the study, and was recorded. Samples of blood were collected and tested by ELISA (enzyme-linked immunosorbent assay for saliva) with positive tests confirmed by Western Blot.

5.3. Training of interviewers

The data collection period was preceded by a session of training for the interviewers. Around 10 potential interviewers participated in the training sessions. The training was conducted by the main investigator who presented to the interviewers the background and objectives of the study, and discussed each item and question on the questionnaire. The training session placed particular emphasis on completing questionnaires with each participant. Following training, the person responsible for the field work carried out the first questionnaire with each interviewer and conducted a special session on coding skills.

The interviewers were mostly workers in the treatment centres and people who have been working in research on drug users.

6. Field Work

6.1. General guidelines

Interviewers were selected from the staff of different centres. For the outreach sample, the process of "snow balling" the sample was used to increase the number of female IDUs interviewed. This involves making initial contacts with respondents and then using respondents to provide access through personal introductions and recommendations to friends and colleagues.

6.2. Strategies in the field

Originally, field work was planned to span over a six-month period, commencing in October 1994 and finishing in April 1995. However, unforeseen obstacles arose, and the data-collection phase was delayed and extended.

6.2.1. The first phase

The participating centres' staffs met on two occasions to prepare the questionnaire to be used in this study. This questionnaire was translated into Spanish and pre-tested in a pilot phase conducted between January and October 1994 when 25 women were interviewed. Results from this pilot study were used to refine the questionnaire and to adapt the wording of questions.

In this period, the major activities were to develop and adapt the final questionnaire for the main survey. Based on the final questionnaire, an interviewer's manual was prepared, and a technical training workshop was held for interviewers in order to achieve standardisation in the enrollment of participants and data collection.

6.2.2. The second phase

Between October 1995 and January 1996, the project was delayed in the recruiting of centres and interviewers due to political repercussions from changes in the Autonomous Government of Madrid. The main investigator and research assistant held numerous meetings with relevant individuals and organisations to explain the objectives of the study.

6.2.3. The third phase

Between October 1995 and March 1996, 309 face-to-face interviews were conducted by trained personnel, using the structured questionnaire which had previously been tested during the pilot phase of the study

7. Statistical Analysis

7.1. Treatment of missing values

First of all, participants with incomplete questionnaires were excluded from the study. Participants who failed to provide written confirmation of their HIV test were coded as "not known" in the HIV prevalence variable so that the rest of the data collected from these participants is included in the study. However, when HIV prevalence is the dependent variable in the multiple logistic regression analysis, the "not known" cases were excluded. This approach is warranted provided that the missing data occurred completely at random (Heitjan, 1997).

7.2. Descriptive analysis

In order to identify the risk factors for HIV infection and the predictive factors for other "at risk" situations (including risk behaviours), such as prison, sharing injecting equipment, prostitution, lack of treatment, we have done a univariate analysis considering each of these risk variables as the dependent variable.

Categorical variables

Frequency distributions and percentages were calculated using 2x2 tables. For categorical and discrete variables, levels of significance were tested using the chi-square test for proportions. Each variable was studied and their potential relationships as main effect and as confounders were analysed. This was done through univariate analysis using the Mantel-Hanzel statistic to obtain the odds ratio (OR) between the dependent and independent variables. The OR provides a measure of association but does not imply a causal relationship. However it is a very useful method to identify those risk factors which are associated with the variable of interest.

Quantitative variables

For continuous data, differences in means were tested using the approximated normal distribution, with the t-test and the z-score. The use of the normal distribution was justifiable due to the large sample size.

We transformed continuous variables into categorical intervals and calculated the frequency distributions.

7.3. Multivariate logistic regression analysis

Few attempts have been made to develop conceptual models of the complex relationship between injecting drug use and AIDS using multifactorial analysis. Frischer et al. (1993) used the Linear Structural Relations Program (LISREL) technique to develop a model to represent the grid of relationships among the observed data in a large-scale cross-national study of IDU in Scotland. The LISREL approach involves the construction of measurement and structural models. The measurement model consists of a number of underlying or latent variables that account for the common relationship among a number of observed variables. Each latent variable is defined by at least two valid indicators. In addition to the latent variables, exogenous variables can be included in the model, which are considered to be predictive of, but not predicted by the latent variables. These variables are generally sociodemographic in nature. The structural model involves path analysis, which has been defined as: "the process of estimating the coefficients of a set of linear structural equations (i.e., regressions) representing the predictive relationships hypothesized by the investigator" (Frischer et al., 1993). One of the advantages of the LISREL model is its ability to consider the model as a system of equations and estimate all the structural coefficients simultaneously.

Although path analysis is not used in the current study, the grid of relationships modeled by Frischer et al. (1993) did influence our conceptual framework and plan for logistic regression analysis. First of all, we chose the causal model rather than the predictive model because we are interested in studying behaviour, and this is the more appropriate model for explaining the behavioural process. The primary dependent variable we studied was HIV seropositivity. Then we examined various latent factors including: prostitution, imprisonment, unsafe drug use, unsafe sex, exposure to drug treatment and the exogenous factors associated with them, including sociodemographic and socioeconomic factors.

For each dependent variable, the relationship between the dependent variable and independent variables was evaluated through crude and adjusted odds ratios with their 95-percent confidence intervals. Those variables which had a statistically significant association with the dependent variable were included in the logistic regression models.

The strategy used for analysis in the multivariate logistic model followed a stepwise approach. A logical and hierarchical structure was used to simulate reality and determine the direct and indirect effects of the explanatory variables. In other words, the explanatory variables were grouped into explanatory groups or models and each group of variables was forced into the regression sequentially. The explanatory groups used to study each of the dependent variables

generally included age, socioeconomic status, marginalisation factors, risk behaviours, and coinfection with STDs or hepatitis B, in that order, although the models included varied according to the dependent variable being studied in each chapter.

The order in which each group was entered into the regression was established based on the biological and sociological hierarchy of their effect on the dependent variable with the more proximal determinants entered first and distal determinants entered later. These effects were estimated based on published literature as well as the author's knowledge of this topic. We also checked variables for interaction effects because in the causal model the effect can be changed by interaction between variables.

The idea behind building these different groups at different stages was to be able to see how each variable works under different "at-risk" situations and be able to consider the whole picture. For example, an explanatory variable such as "age" may have a direct effect on the dependent variable such as "HIV status" (via the decline in immune function with age) and it also may have an indirect effect on the dependent variable, HIV in this case (via exposure to risk factors, such as imprisonment). In other words, the hierarchical structure allows one to see how the relationship between an explanatory variable and the dependent variable changes when certain other explanatory variables are introduced. That is, it permits the identification of indirect, intervening, and direct effects, which is particularly useful in the study of HIV with such a variety of interrelated explanatory variables. Furthermore, in the study of HIV the identification of indirect effects can provide useful "risk behaviour markers" for targeting preventive interventions. For example, cocaine use may not be directly associated with HIV status but its direct relationship with lack of condom use, which is directly associated with HIV highlights cocaine use as a useful "risk behaviour marker".

Using the statistical software programme SPSS, each of the explanatory groups is entered into a stepwise logistic regression. For each step, when a group is entered, the most parsimonious model is chosen, through a process of backward elimination, and then the next group of variables is added. Therefore the final regression model is comprised of the best variables from each of the five explanatory models. The regression models are compared using the improvement in the -2 Log likelihood value (-2LL) measured with the Chi-Square. To determine if there are any significant interaction effects which should be included, interaction terms are defined and systematically analysed for all the variables included in the final model. The addition in goodness of fit gained by including interaction terms is then examined.

Thus, the final picture we obtain from the five different explanatory models is one which allows us to evaluate the effect of how these different variables have a direct effect on the dependent variable and how their effect is mediated through other variables in the causal pathway (see Figure 1).

8. Conclusion

In summary, we have chosen a cross-sectional study to explore the behaviour of female IDUs and determine the relationships between behaviour and a number of dependent variables. We have considered the limitations inherent in a cross-sectional design such as the time sequence of events and have adopted appropriate measures in the questionnaire. Because of the potential for selection bias given the population we studied, we implemented sampling techniques to correct for this, such as sampling from multiple sites. Finally, we have built a number of hierarchical models which at different stages explore how the different variables have a direct and indirect effect on the dependent variables and how their effect is mediated through other variables in the causal pathway.

Chapter Four

•••

Characteristics of Female IDUs

1. Introduction

Women make up an increasing proportion of people infected by the Human Immunodeficiency Virus (HIV). Originally, and in contrast with some African countries where the incidence of HIV and AIDS has been approximately equal in both men and women since the beginning of the epidemic, in North America and Europe women represented only a small fraction of the HIV-infected population. However, the female-to-male ratio has shown an ever increasing trend in recent years: of the 24,920 cases of AIDS reported in Europe by the end of 1996, women accounted for 21 percent of cases among adults and adolescents (European Centre for Epidemiological Monitoring of AIDS, 1996). This represents an increase from 8 percent in 1985 and from 13 percent at the end of 1990 (European Centre for Epidemiological Monitoring of AIDS, 1990).

The increase in female HIV/AIDS cases is due to a set of biological, social, cultural, economic and political factors that make women more vulnerable to HIV transmission than men. Heterosexual transmission, accounting for almost a third of female AIDS cases (European Centre for Epidemiological Monitoring of AIDS 1996), is favoured by a woman's anatomical features, and the relative efficiency rate of male-to-female transmission has been found to be over twice as high as female-to-male transmission (Nicolosi et al., 1994). Women are generally more susceptible to sexually transmitted diseases (STDs), a history of which has been identified as a major risk factor in heterosexual HIV infection (Holst et al., 1987; Musicco et al., 1994). In addition, women tend to have less access to information and health services, are economically dependent on men and have low levels of negotiating power in their sexual relationships with men.

However, the main identified risk factor for HIV transmission in women is injecting drug use, which in Europe accounts for more than half of reported female AIDS cases (European Centre for Epidemiological Monitoring of AIDS 1996). There are a number of gender-specific lifestyle differences which may be putting female injecting drug users (IDUs) at a greater risk of acquiring HIV than male IDUs. Prostitution as a means of financing a drug habit, for example, is more common in women than in men, and has been linked to HIV in both Europe and the United States (European Working Group on HIV Infection in Female Prostitutes, 1993; Watters et al., 1994b). Also, previous studies have shown that female IDUs are more likely to have seropositive male partners and to practice high-risk sexual behavior (Cohen et al., 1989; Klee et al., 1990; Estébanez et al., 1991). The sharing of needles, identified as a major risk factor for HIV transmission (Marmor et al., 1987; Des Jarlais, 1992) also appears to be more common in female than male IDUs (Salmaso et al., 1991; Sotheran et al., 1992).

Even though there is evidence of gender-specific high-risk behaviours, studies on HIV transmission in drug-dependent populations have failed to include gender-specific data. Thus, little research has focused on the role played by reproductive health in the complex relationships among women, drug use and HIV.

Evidence indicates that female IDUs do not constitute a homogeneous population and that different subpopulations may be identifiable on the basis of certain sociodemographic characteristics or behaviours involving the risk of transmitting HIV. The documentation of these characteristics and risk behaviours in female IDUs is important to predict the future course of the AIDS epidemic in the community and to define policies which can appropriately target subgroups at the highest risk. Before predictions can be made, it is important to examine the range of factors specific to women which may possibly influence HIV risk, such as reproductive health, as well as the types of sexual behaviour and the degrees to which injecting equipment is shared, and of course the rate of infection in this group.

Thus by knowing more about the social context within which these women live and their genderspecific needs, such as reproductive health, more effective policies and treatment programmes can be established.

This chapter will present the descriptive results from the interviews with 304 female IDUs recruited from a variety of settings in Madrid with two objectives in mind. Firstly, it is the author's intention that the baseline data presented herein will aid in the formation of policy for this population. Secondly, the data presented will allow the reader to have a firm understanding of the population at hand before examining the multitude of possible bivariate and multivariate relationships between the characteristics of female IDUs and HIV (Chapter Five, prostitution (Chapter Six), previous imprisonment (Chapter Seven), needle sharing (Chapter Eight), condom use (Chapter Nine), and exposure to treatment (Chapter Ten). The sociodemographic characteristics are described in Section 2; Drug-taking risk behaviour in Section 3; Sexual behaviour in Section 4; Prostitution in Section 5; Reproductive health in Section 6; and Service use in Section 7.

2. Sociodemographic Characteristics

Selected sociodemographic characteristics of the total sample population are presented in Table 4.1. The data presented in Table 4.1 identified a young population: more than 60 percent of interviewed female IDUs were less than 30 years old and the median age was 29 years. Almost 55 percent had left school when they were 15 years old, indicating that educational attainment was at a medium level. That is, the majority of the population completed secondary school education (i.e., up to the age of 16), but further education was rare.

In addition, other characteristics presented in Table 4.1 indicate that the population sample was marginalised and of a disadvantaged socioeconomic status. For example, almost one quarter (22.7 percent) are homeless, having no fixed address, although 22.0 percent report living with their parents. Furthermore, only 12.8 percent of women earned a regular salary whilst, conversely, 20.7 percent of women earned their main source of income through sex work and 24.3 percent through illegal activity. Moreover, 32.2 percent of female IDUs recruited into the study reported having been in prison since first injecting drugs.

When the sociodemographic data for Madrid were compared with the four other centres (Paris, Rome, London and Berlin) which participated in the study (Estébanez et al., 1996), no major differences were identified. For example, the median age for all five centres ranged from 29 years (in Spain) to 31 years (in France, Italy, and Germany). However, Spain recorded the lowest mean age of leaving education (the other four countries ranged from 17.08 in Italy to 20.01 in Germany) and the highest rate of homelessness (the other four countries ranged from 6.9 percent in Italy to 17.6 percent in France). Spain also recorded the highest rate of reporting illegal activity (theft, dealing, etc.) as their main source of income (the other countries ranged from 17.6 percent for England to 9.5 percent for France). In contrast, the proportion of Spanish IDUs reporting their main source of income as sex work was higher than the rates reported by Italy (8.3 percent) and England (10.9 percent), but lower than France (24.4 percent) and Germany (28.2 Surprisingly, Spain recorded the lowest rate of previous imprisonment (the other percent). countries ranged from 35.7 percent in England to 58.5 percent in Germany). Finally, a marked difference in living patterns was apparent between the northern European countries (England and Germany) and the southern European countries (Spain and Italy). For example, in Italy, 39.0 percent of female IDUs lived with their parents, compared to 2.7 percent in England.

When we look at data from a 1991 Spanish national survey, we find that amongst women aged 16 to 44 years, 6 percent were illiterate or had had no education, 29 percent completed primary school, 45 percent completed middle school, 9 percent completed professional training, and 11 percent completed university or higher (Instituto de la Mujer, 1991). If we look at unemployment, it appears that 23 percent of Spanish women who want to work are unemployed. This compares to 50 percent of the IDUs in the current study if we combine the following source of income categories: unemployed with regular income from state (6.3 percent), welfare from state (8.2 percent), illegal income (24.3 percent), and no source of income (4.9 percent). Another interesting point of comparison is the rate of living with parents. According to a study by Juan Antonio Fernández Cordón of the Institute of Economy and Geography of the Superior Council of Scientific Investigations (CSIC), the rate of living with one's parents for Spaniards between the ages of 25 and 30 varies between 74 percent for those who are unemployed and 58.6 percent for those who are employed (Díaz de Tuesta, 1997).

In other words, two inferences can be made from the data presented in Table 4.1. First, as would be expected from a study into the risk behaviour of female IDUs, the sociodemographic characteristics of the sample population are not typical of society in Madrid per se and represent a marginalised young community of women. In fact, this sample from Madrid appears to be more marginalised than the samples from other European capitals, as evidenced by their having less education, more homelessness, more illegal income, and more prison experience. Of particular concern is the relatively high rate of homelessness. This subgroup of female IDUs may be at particularly high risk for HIV given the risk behaviours frequently associated with homelessness (drug abuse, alcohol abuse, trading sex for money or drugs, sexual relations with drug-using partners, multiple sexual partners, unprotected sex) and the relationship of homelessness to other health risks, such as mental distress, depression, and trauma, including battery and rape (Fisher et al., 1985; Nyamathi et al., 1997).

Secondly, and of considerable interest, was the number of women (45.5 percent) who reported earning their main source of income through sex work or an illegal activity. It has often been assumed that male IDUs financed their drug habit through theft and drug dealing, whilst women have traditionally been assumed to do so through prostitution. In these data, there is evidence that women use both sources of income in roughly equal proportions. This finding coincides with the findings of a study of criminal activity among women crack addicts which indicated that the women are not only involved in prostitution but also in petty crimes and drug-selling activities and the levels of involvement in these activities are approaching those of their male counterparts (McCoy et al., 1995).

Age at the time of interview 15-19 20-24	7	
15-19	7	
	/	2.3
20-2-7	66	21.7
25-29	110	36.2
30-34	72	23.7
35-39	42	13.8
40-44	5	1.6
45+	2	0.7
TOTAL	304	- 100.0
Age left full-time education		
10-13	88	29.1
14-15	82	27.2
16-18	73	24.2
19+	56	18.5
still studying	3	1.0
TOTAL	302	100.0
Living with		
Parents	67	22.0
Relatives	25	8.2
Partner	76	25.0
Children	6	2.0
Partner and children	22	7.2
Friends	13	4.3
Alone	60	19.7
Other	35	11.5
TOTAL	304	100.0
Where they lived during the past 6 months		
No fixed address	69	22.7
Room rented on a daily basis	27	8.9
Rented flat	29	9.5
Your own (or partner's) flat	76	25.0
Someone else's flat	39	12.8
Shelter, welfare residence	3	1.0
Other	45	14.8
Your parents'/relatives' flat	16	5.3
TOTAL	304	100.0

Table 4.1 Sociodemographic characteristics of the studied population

.

	N	perce
ource of income during the last 6 months		
Employed with regular salary	39	12.8
Unemployed with regular income from state	19	6.3
Temporary work	21	6.9
Welfare from state (allowances)	21	8.2
Self-employed	11	3.6
Sex worker	63	20.7
Illegal income (theft, dealing, etc.)		24.3
Other	37	12.2
No source of income	·· 15	4.9
TOTAL	304	100.0
ho finances the major part of your expenses	1.00	c 0.0
Yourself	182	59.9
Yourself Your partner	45	14.8
Yourself Your partner Your family of origin	45 58	14.8 19.1
Yourself Your partner Your family of origin The social support system	45 58 14	14.8 19.1 4.6
Yourself Your partner Your family of origin	45 58	14.8 19.1
Yourself Your partner Your family of origin The social support system	45 58 14	14.8 19.1 4.6
Yourself Your partner Your family of origin The social support system Others TOTAL	45 58 14 5	14.8 19.1 4.6 1.6
Yourself Your partner Your family of origin The social support system Others TOTAL	45 58 14 5 304	14.8 19.1 4.6 1.6 100.0
Yourself Your partner Your family of origin The social support system Others TOTAL	45 58 14 5	14.8 19.1 4.6 1.6

Table 4.1 (continued) Sociodemographic characteristics of the studied population

3. Drug-Taking Risk Behaviour

Consistent with the marginalised socioeconomic characteristics of the study population (Section 2), the age at the time of first injection was quite young (Table 4.2). The median age of first injection was 19 years (range 11 to 35 years). More than half (53.4 percent) started injecting drugs before the age of 20. This, without a doubt, has an important impact on their psychological development and social behaviour as it is in this stage of life when the basis of their adult personality is established. Therefore, in these women future behavioural problems are found with greater frequency than in those who started injecting drugs later in life.

With regard to drug-taking behaviour there is some evidence that the practice of needle sharing had declined with time. For example, 71.6 percent of female IDUs reported having ever shared needles or syringes and 34.2 percent reported sharing in the past six months. However, the valid proportion for the past six months (i.e., the proportion of female IDUs who have shared over the past six months to those who have ever shared) was 47.9 percent. In fact, there is evidence that drug users in a variety of countries are modifying their risky practices. In a study conducted in the U.K., Hunter et al (1995) reported a decline in the proportion of IDUs that had shared injecting material during the six months prior to the interview over four consecutive years (from 1990 to 1993). Specifically they found a decline from 46.8 percent in 1990 to 39.3 percent in 1993. This reduction in recent needle sharing is comparable to rates found by Estébanez et al. (1996) in other E.E.C. countries: in the U.K. there was a reduction from 69.2 percent that had ever shared injection equipment to 38.5 percent who had shared in the previous six months; the reduction in Germany was from 56.4 percent to 26.9 percent; in Italy from 47.2 percent to 26.3 percent: and in France, where rates were quite a bit lower, from 34.4 percent to 10.0 percent. These data provide encouraging evidence that a decline is underway in the risky behavior of IDUs perhaps as a result of public health interventions advocating harm-reducing behaviour during the 1990s.

The proportion of female IDUs that shared needles in the past 6 months was 34.2 percent, and the frequency of at least once a month or less was 51.0 percent, although a significant number (30.6 percent) shared needles at least once a day (Table 4.2), thus reducing the real number at risk from transmission of HIV by this means to 9.9 percent (The 30.6 percent of the 34 percent). These figures imply that habits concerning the sharing of needles have changed and that HIV prevention campaigns aimed at the drug-using population have been successful.

Almost half of the women questioned (49 percent) injected drugs for the first time with their regular sexual partner. In a study of male and female drug users, Powis et al. (1996) also reported that half of female injectors (51 percent) reported having been given their first injection by a male sexual partner. (Male injectors more often received their first injection from a friend.) This demonstrates the great influence a stable male partner has over a woman, and must be taken into account when developing strategies against drug addiction. Maybe one of the most effective measures to prevent drug addiction in women would be to persuade the drug-using partner to discourage the women from taking drugs and to inform him of the enormous influence he can have in this regard and the consequent responsibility.

Unlike men, very few women (1 percent) participate in drug-taking behaviour alone (Barnard, 1992). This implies that women, more than men, need to feel supported by someone in making decisions of this kind, involving the transgression of a behavioural norm. This is another

indication of the great difference between men and women in the drug-injecting experience. Women are seen to be more affected by prejudice than men and thus need to have someone reaffirm their behavior and push them into action. The explanation for this can be found in the roots of a society which establishes different roles according to one's sex, producing gender differences.

Women most commonly shared needles with a regular sexual partner (47.5 percent), whilst 17.8 percent shared with a male friend and 10.9 percent with a female friend. The main source of injecting equipment for women was the pharmacy or chemist, although 56.6 percent of women reported the use of needle exchange schemes. This observation, however, probably reflects the sample characteristics of recruitment and thus any comparison with other (associated or independent) studies would be meaningless.

Another interesting observation is the types of drugs consumed in the past six months: 89.9 percent of female IDUs reported taking heroin; 63.8 percent cocaine; and 53.3 percent, a combination of both cocaine and heroin. Crack cocaine was used by 15.1 percent of women in the past six months. Pills, such as benzodiazepines and barbiturates, were consumed by 60.6 percent of this sample. The vast majority (97.7 percent) of female IDUs smoked tobacco.

In spite of beliefs to the contrary, alcohol consumption is high, with 70 percent of the sample reporting alcohol consumption in the last six months. These data are similar to those obtained by Estébanez (1996) in the U.K. However, in the other countries consumption was significantly less: 35.5 percent in Italy, 49.8 percent in France, and 44.4 percent in Germany.

The most commonly injected drugs in Spain were heroin (78.9 percent), cocaine (52.6 percent) and heroin and cocaine together (49.7 percent). When compared to the other four centres, rates of heroin injection were higher for Italy (99.5 percent), rates of cocaine injection were lower for Italy (10.6 percent), England (25.8 percent) and France (16.3 percent), and rates of injecting both heroin and cocaine were lower for all four countries (ranging from 9.0 percent for England to 39.4 percent for France) (Estébanez, 1996).

Neither does our data coincide completely with that of other Spanish studies such as Bravo et al. (1996), carried out among drug users of both sexes, in which 38.3 percent had injected cocaine and 71.2 percent had injected heroin and cocaine.

Chitwood and Comerford (1990) found that injectors of cocaine used in combination with opiates were more likely to have multiple sex partners and high-risk injecting behaviour than cocaine-only or opiate-only injectors. Kowalewski (1990) reported that the disinhibiting effects of cocaine may predispose users to HIV infection through high-risk sexual behaviour. These findings are supported by the work of Joe & Simpson (1995) who reported that HIV risks increased as a direct function of cocaine usage level, though only for males. Among females, low-to-intermediate levels of cocaine use (less-than-weekly) were associated with high-risk behaviours. We were not able to establish statistically significant associations with any of these relationships.

Only 15.5 percent used crack in a non-injected form, a finding which coincides with data from the Spanish study by Bravo (1996) in which 13.4 percent used it in the month prior to the study. However, there is quite a bit of variation in crack use reported in other countries studied by

Estébanez (1996), with 11.8 percent in France, 0.9 percent in Italy, 0.4 percent in Germany, and 48.4 percent in England. In a study carried out in the U.S.A. by Watters (1994b), about half (54.3 percent) of the women IDUs reported smoking crack in the 30 days before the interview.

Crack is thought to be an important risk factor for HIV infection and other STDs because of the associated behaviours of the users, including a tendency toward unprotected sexual intercourse and multiple sexual partners (Wilson, 1997; Watters, 1994b; Sterk, 1988; Goldsmith, 1988). Furthermore, studies of criminal activity among women crack addicts indicate that they are not only involved in prostitution but also in petty crimes and drug-selling activities, and the levels of involvement in these activities are approaching those of their male counterparts (McCoy et al., 1995).

Indeed it should be stressed that drugs apart from heroin can be injected and thus need examining. For example, in a sample study based in North West England, Klee (1993) concludes amongst other things, that the injecting of amphetamines was associated with a greater frequency of intercourse which in itself may be associated with a greater risk of HIV infection.

It is interesting to note that 32.7 percent of those who were in prison injected drugs during that time. Drug use in prison has been widely reported. For example, a recent study of female prisoners in Scotland's Corton Vale prison, found that 46 percent of inmates reported IDU and 57 percent of these injector-inmates reported to have injected whilst in prison (Gore et al., 1997). There is also evidence from a study of 559 Scottish inmates (480 males and 79 females) in which 7.7 percent reported IDU on at least one occasion during a period of imprisonment (Power et al., 1992).

Variable	N	percent
Age at first injection		
13 or -	11	3.6
14-16	70	23.1
17-19	81	26.7
20-22	67	22.1
23-25	44	14.5
26-29	24	7.9
30 or +	6	2.0
TOTAL	303	100.0
injected drugs for the first time with		
A regular sexual partner	148	49.0
A female friend	58	19.2
A male friend	35	11.6
Alone	3	1.0
Multiple users	45	14.9
Others	10	3.3
A relative	3	1.0
TOTAL	302	100.0

Table 4.2Drug-use habits

Drugs "used" within the last 6 months (n=304; multiple responses possible)

Heroin	271	89.9	
Cocaine	194	63.8	
Heroin and cocaine simultaneously	162	53.3	
Methadone	72	23.7	
Other opiates	45	14.8	
Amphetamines	21	6.9	
Benzodiazepine	113	37.2	
Barbiturates	71	23.4	
Crack	46	15.1	
Ecstasy	19	6.3	
Alcohol	213	70.1	
Tobacco	297	97.7	
Hashish	126	41.4	

Table 4.2 (continued) Drug- use habits

Variable	 N	percent
•		

.

Drugs "injected" within the last 6 months (n=304; Heroin	240	
Cocaine	160	52.6
Heroin and cocaine simultaneously	151	49.7
Methadone	9	3.0
Other opiates	6	2.0
Amphetamines	3	1.0
Benzodiazepine	11	3.6
Barbiturates	3	1.0
Crack	·· 1	0.3
Ecstasy	1	0.3
ver shared used needles or syringes		
Yes	217	71.6
No	86	28.4
TOTAL	303	100.0
nared needles within the last 6 months		
Yes	104	34.2
No	199	65.8
TOTAL	303	100.0
equency of needle sharing in the last 6 months		
Once a month or less	50	48.1
Once a week	18	17.3
Once a day	11	10.6
More than once a day	19	18.3
No response	6	5.7
TOTAL	104	100.0
naring behaviour in the last 6 months		
Only borrowing	9	8.6
Only lending	3	2.9
Both, but more borrowing	16	15.4
Both, but more lending	25	24.0
Both equally	48	46.2
No response	3	2.9
TOTAL	104	100.0
5.0		

Table 4.2 (continued) Drug-use habits

Variable	<u>N</u>	percent
Get used needles from		
A regular sexual partner	48	47.5
A female friend	11	10.9
A male friend	18	17.8
Other	24	23.8
TOTAL	101	100.0
Give your used needles to		
A regular sexual partner	46	48.4
A female friend	- 10	10.5
A male friend	15	15.8
Other	24	25.3
TOTAL	95	100.0
Method used to clean used needles		
Cold water	47	46.1
Hot water	10	9.8
Boiling	0	0.0
Bleach	18	17.6
Alcohol	4	3.9
More than one method	10	9.8
Other	2	2.0
I don't clean them	11	10.8
TOTAL	104	100.0
Why you refused the offer of a used needle		
Because of HIV/AIDS	119	40.3
Because of hepatitis	1	0.3
Appearance of needle/syringe	12	4.1
Because of person offering	29	9.8
Because I never borrow	66	22.4
For other reasons	68	23.1
TOTAL	295	100.0
Injected drugs in prison		
Yes	32	32.7
No	66	67.3
TOTAL	98	100.0

.

4. Sexual Behaviour

The sexual behaviour of female IDUs is examined with respect to regular partners, casual partners and clients. The characteristics and sexual behaviour of prostitutes (i.e., with clients) are examined in Section 5. Selected characteristics describing the sexual behaviour with regular partners are presented in Table 4.3.

Contrary to the popular belief that IDUs are less sexually active than the general population, there are many recent studies that show that the majority of drug injectors are having sex: 82 percent in England (Donoghoe, 1991), 88 percent in North West England (Klee, 1990), 77 percent in Amsterdam (Van den Hoek et al., 1990), 93 percent in South Florida (Chitwood, 1990), and 89 percent of a sample of female injectors in New Jersey (Freeman, 1994). Similarly, our study concluded that out of those interviewed, most were sexually active.

Regular partner: The vast majority (i.e., 96.4 percent) of female IDUs reported ever having a regular sexual partner, and 61.4 percent reported having a regular partner at the time of the study. Furthermore, 87.7 percent of women reported ever having a regular partner who was an IDU. This high percentage of IDU partners is confirmed in other studies (Cohen et al., 1989; Mulleady et al., 1990; Dolan et al., 1991; Donoghoe, 1989) and is explained by the fact that there are many more male IDUs than female IDUs. Moreover, in the current study 49.5 percent of women reported ever having a regular partner who was HIV positive.

Vaginal intercourse was the most common (i.e., 88.8 percent) sexual activity. Oral sex was also common with 70.8 percent of women reporting it in the past 6 months. Anal intercourse was less common with 28.1 percent of women reporting "always" or "sometimes" having it in the past six months.

If these results are compared with sexual practices in the general population in Spain, according to the study by Usieto (1992), we find that they are comparable only as far as vaginal intercourse, in which 9 percent reported never or rarely having it. Comparing rates of anal and oral sex, there were big differences from the general population. Oral sex was practiced less in the general population (42 percent of women and 45 percent of men) than in our study. Usieto also reported that only 9 percent of men had ever had heterosexual anal intercourse in their lives, and no data was available on the anal sex practices of women. However, another study conducted in Spain amongst IDUs by Bravo (1994) found that 18.3 percent of women had had anal sex.

Similar figures were obtained by Strang (1994) in a study in London in which 20 percent of female heroin users had experienced anal intercourse within the last year and 10 percent within the last month. In another study on women attending a STD clinic in London by Sherr (1991), 10 percent reported anal intercourse with a frequency of 1-30 occasions over the previous month. Spira (1993) in his study about trends in sexual behaviour found that the occurrence of anal intercourse varies greatly from one country to another, and notes an important increase during the last 20 years. Spira found that 26.2 percent of French women and 21.7 percent of Danish women reported ever having anal intercourse. Clearly, preventive programmes need to address the risks of anal sex amongst heterosexuals.

Condoms were "always" used by 22.3 percent of women in our study who reported having vaginal intercourse in the past six months. These data coincide with those found by Van den Hoek (1992) who found that 25 percent of IDUs with steady partners reported using condoms for vaginal intercourse.

In anal intercourse, only 22 percent reported "always" doing so with condoms. Similarly, a study among California heterosexuals found that only 13.5 percent of women reported "always" using condoms for anal intercourse and 74 percent reported "never" using condoms (Erickson et al, 1995).

Finally, very few women used condoms during oral sex. For example, only 2.4 percent of women who had had oral sex in the past six months, "always" used a condom. Although generally considered a "lower-risk sexual behaviour", the transmission of HIV by unprotected receptive oral-genital contact (with and without ejaculation) is being described with increased frequency, particularly as the frequency of higher-risk sexual behaviour declines (Berry and Shea, 1997). As these authors suggest, recommendations for safer sex practices need to include discussion of the risks of receptive oral-genital contact.

There is a significant association between never using condoms with the regular partner and other risk factors, such as having shared needles in the last six months. In fact, the risk of having shared injecting equipment is 3.5 times higher for those women who never used condoms in vaginal intercourse with their regular partner, as can be seen in Table 4.5

Casual partner: As illustrated in Table 4.4, 79.5 percent of women reported having casual sexual partners. In the study by Van de Hoek in Holland, only 23 percent of the women IDU's had had casual partners in the previous six months. The proportion of women who reported having a casual partner who was also an IDU or was HIV-positive was 66.4 percent and 30.1 percent, respectively. Condom use was higher with casual partners. For example, 51.0 percent of women who reported having vaginal intercourse with a casual partner in the past six months, "always" used a condom. Likewise, 40.0 percent of women, "always" used condoms during anal intercourse with a casual partner, and 12.0 percent during oral sex.

With respect to casual partners, a significant association was also found between never using condoms and having shared needles in the last six months. In fact, the risk of sharing needles is 7.3 times higher for women who never use condoms in vaginal relationships with a casual partner (Table 4.6).

Condom use requires a degree of confidence within the relationship to be able to talk about sex. However, this is not always the case, as sex is often a spontaneous event. The results of the survey into sexual activity imply that female IDUs are at an increased risk from heterosexual transmission of HIV, putting a greater emphasis on the man as he is expected to take the lead in sexual relations and should therefore be responsible for the supply of condoms.

Variable	N	percent
Ever had regular partner		
Ever had regular partner Yes	293	06 4
No	293 11	96.4 3.6
	11	5.0
TOTAL	304	100.0
Ever had regular IDU partner		
Yes	257	87.7
No	36	12.3
TOTAL	293	100.0
Ever had regular HIV+ partner		
Yes	138	49.5
No	141	50.5
TOTAL	279	100.0
Changed behaviour with regular HIV+ partner		
Yes	64	46.7
No	73	53.3
TOTAL	137	100.0
How behaviour changed with regular HIV+ partner		
Stopped having sexual contacts	19	30.2
Stopped vaginal sex	1	1.6
Stopped anal sex	0	0.0
Stopped oral sex	0	0.0
Stopped sex during your periods	0	0.0
Increased condom use	43	68.3
TOTAL	63	100.0
Current regular partner		
Yes	180	61.4
No	113	38.6
TOTAL	293	100.0

Table 4.3Sexual relationship with regular partner

Variable	<u>N</u>	percen
Frequency of sexual relationships with current partner		
Never	43	23.9
Once a month	55	30.6
Once a week	61	33.9
Once a day	21	11.7
TOTAL	180	100.0
Kind of sex in the last 6 months		
-Vaginal intercourse		
Always/sometimes	158	88.8
Never	20	11.2
TOTAL	178	100.0
-Anal intercourse		
Always/sometimes	50	28.1
Never	128	71.9
TOTAL	178	100.0
-Oral sex		
Always/sometimes	126	70.8
Never	52	29.2
TOTAL	178	100.0
Condom use in the last 6 months		
-Vaginal intercourse		
Always	35	22.3
Never/sometimes	122	77.7
TOTAL	157	100.0
-Anal intercourse		
Always	11	22.0
Never/sometimes	39	78.0
	50	100.0
TOTAL	50	100.0
-Oral sex	•	A 1
Always	3	2.4
Never/sometimes	123	97.6
TOTAL	126	100.0
61		

Table 4.3 (continued) Sexual relationship with regular partner

Variable	N	percent
Ever had casual partners		
Yes	241	79.5
No	62	20.5
TOTAL	303	100.0
Ever had casual IDU partners		
Yes	156	66.4
No	79	33.6
TOTAL	235	100.0
Ever had casual HIV+ partners		
Yes	53	30.1
No	123	69.9
TOTAL	176	100.0
Kind of sex in the last 6 months with casual partners		
-Vaginal intercourse	00	04.1
Always/sometimes	98	96.1
Never	4	3.9
	100	100.0
TOTAL	102	100.0
TOTAL -Anal intercourse	102	100.0
	102	100.0
-Anal intercourse		
-Anal intercourse Always/sometimes	15	14.7
-Anal intercourse Always/sometimes Never	15 87	14.7 85.3
-Anal intercourse Always/sometimes Never TOTAL -Oral sex	15 87	14.7 85.3
-Anal intercourse Always/sometimes Never TOTAL	15 87 102	14.7 85.3 100.0

.

Table 4.4 Sexual relationship with casual partner

Variable	N	percent
Condom use in the last 6 months with casual partners -Vaginal intercourse		
Always	50	51.0
Never/sometimes	48	49.0
TOTAL	98	100.0
-Anal intercourse		
Always	6	40.0
Never/sometimes	9	60.0
TOTAL	15	100.0
-Oral sex		
Always	9	12.0
Never/sometimes	66	88.0
TOTAL	75	100.0

Table 4.4 (continued) Sexual relationship with casual partner

Table 4.5 Condom use with regular partner and sharing injecting equipment

•

Condom use	Yes	No	Total	OR (95 % CI)
Vaginal intercourse				
Always	5	30	35 (22.3 %)	l(reference)
Never/sometimes	45	77	122 (77.7 %)	3.5(1.2-11.1)
			157 (100 %)	
Anal intercourse				
Always	1	10	11 (22.0%)	l(reference)
Never/sometimes	15	24	39 (78.0 %)	6.2(0.7-143.8)
			50 (100 %)	
Oral intercourse				
Always	-	3	3 (2.4 %)	l(reference
Never/sometimes	47	76	123 (97.6 %)	undefined
			126 (100 %)	

Equipment shared

.

64

Table 4.6 Condom use with casual partner and sharing injecting equipment

.

Condom use	Yes	No	Total	OR(95 % CI)
Vaginal intercourse				
Always	7	43	50(51.0 %)	1(Reference)
Never/sometimes	26	22	48(49.0 %)	7.3(2.5-21.9)
			98 (100 %)	
Anal intercourse				
Always		6	6(40.0 %)	1(reference)
Never/sometimes	2	7	9(60.0 %)	undefined
			15 (100 %)	
Oral intercourse				
Always	3	6	9(12.0 %)	1(reference)
Never/sometimes	23	43	66(88.0 %)	1.1(0.2-6.0)
			75 (100 %)	

EQUIPMENT SHARED

.

5. Prostitution

Selected characteristics, describing the sexual behaviour with clients can be found in Table 4.7, where it is shown that 46.1 percent of women, reported ever having a client. The median age at first client was 22 years. The median number of clients per month was 9, range: minimum 0 and maximum 240. It should be noted, however, that 64 of the women (or 40.4 percent) could be considered ex-prostitutes (i.e., they had no clients in the past six months).

Condom use was higher with clients in comparison to regular and casual partners (Tables 4.3 and 4.4). For example, 66.7 percent of women who "always" or "sometimes" had vaginal intercourse with a client, "always" used a condom. Similarly, 40.9 percent of women did so during anal intercourse; 28.6 percent during oral sex; and 5.6 percent whilst masturbating a client. It should, however, be noted that these figures may be subject to small number problems.

The reported rate of prostitution in Spain (46.1 percent) was similar to that reported for all five centres in the European study (45.7 percent; range: 33.5 percent in Italy to 62.0 percent in Germany; Estébanez et al. 1996). These rates are higher than those observed by Donoghoe et al. (1991) who reported that between 7 and 10 percent of drug injectors had accepted drugs for sex. Likewise, in a study of drug injectors in South Florida, 21 percent reported trading sex for money and 21 percent sex for drugs (McCoy and Khoury, 1990). The relationship between drug use and prostitution is also highlighted by data looking at IDU amongst sample populations of prostitutes. For example, in a study of prostitutes in Frankfurt, 74 percent of women were IDUs (Doerr et al 1990); in Madrid, 45 percent of female prostitutes were IDUs (Colomo et al., 1990); and in Glasgow, 59 percent of female prostitutes were drug users (McKeganey et al., 1990).

In the current study, the majority of women (63 percent) reported that they initially exchanged sex for drugs or money, in order to pay for drugs. This relation has been documented frequently, but there has been no conclusive evidence as to whether drug use was the cause or effect of prostitution (Serraino et al., 1991; Nicolosi et al., 1992). These data, however, provide some evidence that drug taking is the cause of prostitution, as opposed to its effect. In this study, 81 percent of prostitutes reported that drugs came first and afterwards came prostitution. To explore this relationship in further detail the determinants of prostitution are examined through bivariate and multivariate analysis in Chapter 6. However, it is worth stressing that the link between drug use and sexual behaviour is complicated by a multitude of other factors including, for example, the disinhibiting effects of drugs, direct pharmacological effects on sexual behaviour and the "milieu in which sexual opportunities are interwoven with drug and alcohol use, such as bars and clubs" (Donoghoe, 1992).

With respect to sexual relations with clients, a significant association was shown between never using condoms and sharing needles, with the risk of sharing needles 4.8 times higher in women who never use condoms (Table 4.8).

$17 \text{ or } -$ 20 14.6 $18 \text{ to } 22$ 52 38.0 $23 \text{ to } 27$ 43 31.4 $28 \text{ or } +$ 22 16.1 $\overline{\text{TOTAL}}$ $\overline{137}$ $\overline{100.0}$ Which came first $\overline{137}$ $\overline{100.0}$ Which came first 111 81.0 Prostitution 17 12.4 Both 9 6.6 $\overline{\text{TOTAL}}$ $\overline{131}$ $\overline{100.0}$ Why did you first do this 1 $\overline{131}$ $\overline{100.0}$ Why did you first do this 88 62.9 47 33.6 I needed money other than for drugs 47 33.6 42.9 0.7 $\overline{\text{TOTAL}}$ 140 100.0 100.0	Variable	<u>N</u>	percen
Yes 140 46.1 No 164 53.9 \overline{TOTAL} $\overline{304}$ $\overline{100.0}$ Age at first sex with client 17 or - 20 14.6 18 to 22 52 38.0 23 to 27 43 31.4 28 or + 22 16.1 100.0 100.0 Which came first Drugs 111 81.0 100.0 Which came first 137 100.0 100.0 Which came first 137 100.0 0 Which came first 111 81.0 17 12.4 Both 9 6.6 100.0 0 0 Why did you first do this 131 100.0 0 0 Why did you first do this 1 131 100.0 0 Why did you first do this 1 140 100.0 0 Verage number of clients per month during the last 6 months Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 9			
No 164 53.9 $\overline{\text{TOTAL}}$ $\overline{304}$ $\overline{100.0}$ Age at first sex with client 17 or - 20 14.6 18 to 22 52 38.0 23 to 27 43 31.4 28 or + 22 16.1 $\overline{\text{TOTAL}}$ $\overline{137}$ $\overline{100.0}$ Which came first 137 $\overline{100.0}$ Which came first 131 $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ Why did you first do this 8 62.9 I needed money ordrugs 8 8 I needed somewhere to stay 4 2.9 Other 1 0.7 $\overline{\text{TOTAL}$ 140 $\overline{100.0}$ Average number of clients per month during the last 6 months 55 Ex-prostitute		1.40	
$\overline{\text{TOTAL}}$ $\overline{304}$ $\overline{100.0}$ Age at first sex with client 17 or - 20 14.6 18 to 22 52 38.0 23 to 27 43 31.4 28 or + 22 16.1 $\overline{\text{TOTAL}}$ $\overline{137}$ $\overline{100.0}$ Which came first $\overline{1377}$ $\overline{100.0}$ Which came first $\overline{1377}$ $\overline{100.0}$ Which came first $\overline{177}$ 12.4 Both 9 6.6 $\overline{\text{TOTAL}}$ $\overline{131}$ $\overline{100.0}$ Why did you first do this 1 1 I needed money for drugs 88 62.9 I needed money other than for drugs 47 33.6 I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 $\overline{\text{TOTAL}}$ 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14			
Age at first sex with client 17 or - 20 14.6 18 to 22 52 38.0 23 to 27 43 31.4 28 or + 22 16.1 TOTAL 137 100.0 Which came first 137 100.0 Which came first 111 81.0 Prostitution 17 12.4 Both 9 6.6 TOTAL 131 100.0 Why did you first do this 131 100.0 Why did you first do this 88 62.9 I needed money for drugs 88 62.9 I needed money other than for drugs 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 <	NO	164	53.9
17 or - 20 14.6 18 to 22 52 38.0 23 to 27 43 31.4 28 or + 22 16.1 TOTAL 137 100.0 Which came first 137 100.0 Which came first 111 81.0 Prostitution 17 12.4 Both 9 6.6 TOTAL 131 100.0 Why did you first do this 1 131 I needed money for drugs 88 62.9 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 103 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	TOTAL	304	100.0
18 to 22 52 38.0 23 to 27 43 31.4 28 or + 22 16.1 \overline{TOTAL} $\overline{137}$ $\overline{100.0}$ Which came first Drugs 111 81.0 Prostitution 17 12.4 Both 9 6.6 \overline{TOTAL} 131 100.0 Why did you first do this 131 100.0 Why did you first do this 131 100.0 Why did you first do this 88 62.9 I needed money for drugs 88 62.9 I needed somewhere to stay 4 2.9 Other 1 0.7 \overline{TOTAL} 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7	Age at first sex with client		
23 to 27 43 31.4 28 or + 22 16.1 \overline{TOTAL} $\overline{137}$ $\overline{100.0}$ Which came first $\overline{137}$ $\overline{100.0}$ Which came first $\overline{111}$ 81.0 Prostitution 17 12.4 Both 9 6.6 \overline{TOTAL} $\overline{131}$ $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ I needed money for drugs 88 62.9 I needed somewhere to stay 4 2.9 Other 1 0.7 \overline{TOTAL} 140 100.0 Average number of clients per month during the last 6 months $Ex-prostitute$ $Ex-prostitute$ 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13	17 or -	20	14.6
$28 \text{ or } +$ 22 16.1 TOTAL $\overline{137}$ $\overline{100.0}$ Which came first $\overline{137}$ $\overline{100.0}$ Which came first $\overline{177}$ 12.4 Both 9 6.6 \overline{TOTAL} $\overline{131}$ $\overline{100.0}$ Why did you first do this 17 12.4 Both 9 6.6 \overline{TOTAL} $\overline{131}$ $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ I needed money for drugs 88 62.9 I needed somewhere to stay 4 2.9 Other 1 0.7 \overline{TOTAL} $\overline{140}$ $\overline{100.0}$ Average number of clients per month during the last 6 months Ex-prostitute Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	18 to 22	52	38.0
$28 \text{ or } +$ 22 16.1 $\overline{\text{TOTAL}}$ $\overline{137}$ $\overline{100.0}$ Which came first 111 81.0 Prostitution 17 12.4 Both 9 6.6 $\overline{\text{TOTAL}}$ $\overline{131}$ $\overline{100.0}$ Why did you first do this 17 12.4 Both 9 6.6 $\overline{\text{TOTAL}}$ $\overline{131}$ $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ I needed money for drugs 88 62.9 I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 $\overline{\text{TOTAL}}$ 140 100.0 Average number of clients per month during the last 6 months $8xerage number of clients per month during the last 6 months Ex-prostitute 55 40.4 10.3 61 190 13 9.6 91 120 13 9.6 $	23 to 27	43	31.4
Which came first Drugs 111 $\$1.0$ Prostitution 17 12.4 Both 9 6.6 $\overline{\text{TOTAL}}$ 131 $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ Why did you first do this 131 $\overline{100.0}$ Why did you first do this 111 $\$88$ 62.9 I needed money for drugs $\$8$ 62.9 I needed somewhere to stay 4 2.9 Other 1 0.7 $\overline{\text{TOTAL}}$ 140 $\overline{100.0}$ Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	28 or +	22	
Drugs 111 81.0 Prostitution 17 12.4 Both 9 6.6 TOTAL 131 100.0 Why did you first do this 131 100.0 Why did you first do this 131 100.0 Why did you first do this 88 62.9 I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	TOTAL	137	100.0
Prostitution 17 12.4 Both 9 6.6 TOTAL 131 100.0 Why did you first do this 131 100.0 Why did you first do this 88 62.9 I needed money for drugs 47 33.6 I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	Which came first		
Both 9 6.6 TOTAL 131 100.0 Why did you first do this 1 111 I needed money for drugs 88 62.9 I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	Drugs	111	81.0
TOTAL 131 100.0 Why did you first do this 1 100.0 Why did you first do this 88 62.9 I needed money for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	Prostitution	17	12.4
Why did you first do this 88 62.9 I needed money for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 40.4 I to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	Both	9	6.6
I needed money for drugs 88 62.9 I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	TOTAL	131	100.0
I needed money for drugs 88 62.9 I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	Why did you first do this		
I needed money other than for drugs 47 33.6 I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 40.4 I to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1		88	62.9
I needed somewhere to stay 4 2.9 Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 40.4 I to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1			
Other 1 0.7 TOTAL 140 100.0 Average number of clients per month during the last 6 months 55 40.4 Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1			
Average number of clients per month during the last 6 months Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1			
Ex-prostitute 55 40.4 1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	TOTAL	140	100.0
1 to 30 20 14.7 31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	Average number of clients per month during the last 6 months		
31 to 60 14 10.3 61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1	•		40.4
61 to 90 13 9.6 91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1		20	14.7
91 to 120 13 9.6 121 to 150 14 10.3 151 or + 7 5.1			
121 to 150 14 10.3 151 or + 7 5.1		13	
151 or + 7 5.1		13	
	121 to 150	14	10.3
TOTAL 136 100.0	151 or +	7	5.1
	TOTAL	136	100.0

Table 4.7 Sexual relationship with clients

-Vaginal intercourse 52 66.7 Always 26 33.3 TOTAL 78 100.0 -Anal intercourse 9 40.9 Always 9 40.9 Never 13 59.1 TOTAL 22 100.0 -Oral sex 22 28.6 Never 55 71.4	Variable	N	percent
-Vaginal intercourse 78 91.8 Always/sometimes 7 8.2 \overline{TOTAL} 85 100.0 -Anal intercourse Always/sometimes 22 25.9 Never 63 74.1 \overline{TOTAL} 85 100.0 -Oral sex 7 8.2 Always/sometimes 7 8.5 Never 63 74.1 \overline{TOTAL} 85 100.0 -Oral sex 7 90.6 Never 8 9.4 \overline{TOTAL} 85 100.0 -Masturbation 72 85.7 Never 12 14.3 \overline{TOTAL} 84 100.0 Condom use in the last 6 months with clients - -Vaginal intercourse 26 33.3 \overline{TOTAL} 78 100.0 -Anal intercourse 4 22 100.0 -Anal intercourse 26 33.3 1 \overline{TOTAL} 78 100.0 - -Anal intercourse 22 100.0 <td>Kind of sex in the last 6 months with clients</td> <td></td> <td></td>	Kind of sex in the last 6 months with clients		
Always/sometimes 78 91 8 Never 7 8.2 $\overline{\text{TOTAL}}$ 85 100.0 -Anal intercourse Always/sometimes 22 25.9 Never 63 74.1 $\overline{\text{TOTAL}}$ 85 100.0 -Oral sex 100.0 -Oral sex 7 90.6 Never 8 9.4 $\overline{\text{TOTAL}}$ 85 100.0 -Oral sex 7 90.6 Never 8 9.4 $\overline{\text{TOTAL}}$ 85 100.0 -Masturbation Always/sometimes 72 85.7 Never 12 14.3 12 14.3 $\overline{\text{TOTAL}}$ 84 100.0 10.0 Condom use in the last 6 months with clients - 26 33.3 $\overline{\text{TOTAL}}$ 78 100.0 - -Anal intercourse 4lways 9 40.9 Never 13 59.1 13 59.1 $\overline{\text{TOTAL}}$ 22 100.0 - -Oral sex <td></td> <td></td> <td></td>			
Never 7 8.2 $\overline{\text{TOTAL}}$ $\overline{\text{85}}$ $\overline{100.0}$ -Anal intercourse Always/sometimes 22 25.9 Never 63 74.1 $\overline{\text{TOTAL}}$ $\overline{\text{85}}$ $\overline{100.0}$ -Oral sex Always/sometimes 77 90.6 $\overline{\text{Never}}$ 8 9.4 $\overline{\text{TOTAL}}$ $\overline{\text{85}}$ $\overline{100.0}$ $\overline{\text{Never}}$ 8 9.4 $\overline{\text{TOTAL}}$ $\overline{\text{85}}$ $\overline{100.0}$ $\overline{\text{Never}}$ $\overline{12}$ $\overline{14.3}$ $\overline{\text{TOTAL}}$ $\overline{\text{84}}$ $\overline{100.0}$ $\overline{\text{Never}}$ $\overline{12}$ $\overline{14.3}$ $\overline{\text{TOTAL}}$ $\overline{\text{84}$ $\overline{100.0}$ $\overline{\text{Condom use in the last 6 months with clients}}$ $\overline{\text{Never/sometimes}}$ $\overline{22}$ $\overline{66.7}$ $\overline{\text{Never/sometimes}}$ $\overline{26}$ $\overline{33.3}$ $\overline{\text{TOTAL}}$ $\overline{78}$ $\overline{100.0}$ -Anal intercourse $\overline{\text{Always}}$ 9 40.9 $\overline{\text{Never}$ $\overline{13}$ $\overline{59.1}$ $\overline{\text{TOTAL}}$ $\overline{22}$ $\overline{100.0}$ $\overline{22}$ $\overline{100.0}$	-	78	91.8
-Anal intercourse 22 25.9 Never 63 74.1 TOTAL 85 100.0 -Oral sex $Always/sometimes$ 77 90.6 Never 8 9.4 TOTAL 85 100.0 -Oral sex 77 90.6 Never 8 9.4 TOTAL 85 100.0 -Masturbation $Always/sometimes$ 72 85.7 Never 12 14.3 100.0 -Masturbation $Always/sometimes$ 72 85.7 Never 12 14.3 100.0 -Masturbation 72 85.7 Never 12 14.3 TOTAL 84 100.0 -Oral intercourse 26 33.3 TOTAL 78 100.0 -Anal intercourse $4lways$ 9 40.9 Never 13 59.1 71.4 TOTAL 22 100.0 71.4 <td></td> <td></td> <td></td>			
Always/sometimes 22 25.9 Never 63 74.1 $\overline{\text{TOTAL}}$ 85 100.0 -Oral sex Always/sometimes 77 90.6 Never 8 9.4 $\overline{\text{TOTAL}}$ 85 100.0 -Masturbation 85 100.0 -Masturbation 85 100.0 -Masturbation 72 85.7 Never 12 14.3 $\overline{\text{TOTAL}}$ 84 100.0 Condom use in the last 6 months with clients - -Vaginal intercourse 26 33.3 $\overline{\text{TOTAL}}$ 78 100.0 -Anal intercourse 4 40.9 Never 13 59.1 $\overline{\text{TOTAL}}$ 72 28.6 Never 13 59.1 $\overline{\text{TOTAL}}$ 22 28.6 Never 55 71.4 $\overline{\text{OVAL}}$ 22 28.6 Never 55 71.4	TOTAL	85	100.0
Always/sometimes 22 25.9 Never 63 74.1 $\overline{\text{TOTAL}}$ 85 100.0 -Oral sex Always/sometimes 77 90.6 Never 8 9.4 $\overline{\text{TOTAL}}$ 85 100.0 -Masturbation 85 100.0 -Masturbation 85 100.0 -Masturbation 72 85.7 Never 12 14.3 $\overline{\text{TOTAL}}$ 84 100.0 Condom use in the last 6 months with clients - -Vaginal intercourse 26 33.3 $\overline{\text{TOTAL}}$ 78 100.0 -Anal intercourse 4 40.9 Never 13 59.1 $\overline{\text{TOTAL}}$ 72 28.6 Never 13 59.1 $\overline{\text{TOTAL}}$ 22 28.6 Never 55 71.4 $\overline{\text{OVAL}}$ 22 28.6 Never 55 71.4	-Anal intercourse		
Never 63 74.1 TOTAL 85 100.0 -Oral sex Always/sometimes 77 90.6 Never 8 9.4 TOTAL 85 100.0 -Masturbation 85 100.0 -Masturbation 85 100.0 -Masturbation 84 100.0 -Vaginal intercourse 26 33.3 TOTAL 78 100.0 -Anal intercourse 3 9 40.9 Never 13 59.1 13 59.1 TOTAL 22 100.0 - -Oral sex Always 22 28.6		22	25.9
-Oral sex Always/sometimes Never TOTAL TOTAL -Masturbation Always/sometimes Never 12 14.3 TOTAL -Vaginal intercourse Always Soundom use in the last 6 months with clients -Vaginal intercourse Always Soundom use in the last 6 months with clients -Vaginal intercourse Always Soundom use in the last 6 months with clients -Vaginal intercourse Always Soundom use Always Soundom use Soundom use Always Soundom use Soundom use Always Soundom use Always Soundom use Soundom use Soundom use Soundom use Soundom use Soundom use Always Soundom use Soundom use Soundom use Always Soundom use Soundom use Soundom use Always Soundom use Soundom			
Always/sometimes 77 90.6 Never 8 9.4 $\overline{\text{TOTAL}}$ 85 100.0 -Masturbation 72 85.7 Always/sometimes 72 85.7 Never 12 14.3 $\overline{\text{TOTAL}}$ 84 100.0 Condom use in the last 6 months with clients -Vaginal intercourse Always 52 66.7 Never/sometimes 26 33.3 $\overline{\text{TOTAL}}$ 78 100.0 -Anal intercourse 9 40.9 Always 9 40.9 Never 13 59.1 $\overline{\text{TOTAL}}$ 22 100.0 -Oral sex 22 28.6 Never 55 71.4	TOTAL	85	100.0
Always/sometimes 77 90.6 Never 8 9.4 $\overline{\text{TOTAL}}$ 85 100.0 -Masturbation 72 85.7 Always/sometimes 72 85.7 Never 12 14.3 $\overline{\text{TOTAL}}$ 84 100.0 Condom use in the last 6 months with clients -Vaginal intercourse Always 52 66.7 Never/sometimes 26 33.3 $\overline{\text{TOTAL}}$ 78 100.0 -Anal intercourse 9 40.9 Always 9 40.9 Never 13 59.1 $\overline{\text{TOTAL}}$ 22 100.0 -Oral sex 22 28.6 Never 55 71.4	-Oral sex		
Never 8 9.4 TOTAL 85 100.0 -Masturbation Always/sometimes 72 85.7 Never 12 14.3 TOTAL 84 100.0 Condom use in the last 6 months with clients -Vaginal intercourse 84 100.0 Always 52 66.7 83.3 66.7 TOTAL 78 100.0 78 100.0 -Anal intercourse 26 33.3 78 100.0 -Anal intercourse Always 9 40.9 9 Never 13 59.1 70.0 70.0 -Oral sex Always 22 28.6 71.4		77	90.6
-Masturbation Always/sometimes Never 12 14.3 TOTAL 70TAL 70TAL 70TAL 70TAL 78 100.0 78 100.0 78 100.0 -Anal intercourse Always 78 100.0 -Anal intercourse Always 9 40.9 Never 13 59.1 TOTAL 70TAL 22 100.0 -Oral sex Always 22 28.6 Never 55 71.4	-		
Always/sometimes 72 85.7 Never 12 14.3 TOTAL 84 100.0 Sondom use in the last 6 months with clients -Vaginal intercourse Always 52 66.7 Never/sometimes 26 33.3 TOTAL 78 100.0 -Anal intercourse 31.00.0 - Always 9 40.9 Never 13 59.1 TOTAL 22 100.0 -Oral sex 22 28.6 Never 55 71.4	TOTAL	85	100.0
Never 12 14.3 TOTAL 84 100.0 Condom use in the last 6 months with clients . -Vaginal intercourse Always 52 66.7 Always 26 33.3	-Masturbation		
Never 12 14.3 TOTAL 84 100.0 Condom use in the last 6 months with clients . -Vaginal intercourse Always 52 66.7 Always 26 33.3	Always/sometimes	72	85.7
Condom use in the last 6 months with clients-Vaginal intercourse5266.7Always2633.3TOTAL78100.0-Anal intercourse78100.0-Anal intercourse940.9Never1359.1TOTAL22100.0-Oral sex2228.6Never5571.4		12	
-Vaginal intercourse 52 66.7 Always 26 33.3 TOTAL 78 100.0 -Anal intercourse 9 40.9 Always 9 40.9 Never 13 59.1 TOTAL 22 100.0 -Oral sex 22 28.6 Never 55 71.4	TOTAL	84	100.0
Always 52 66.7 Never/sometimes 26 33.3 TOTAL 78 100.0 -Anal intercourse 9 40.9 Always 9 40.9 Never 13 59.1 TOTAL 22 100.0 -Oral sex 22 28.6 Never 55 71.4	Condom use in the last 6 months with clients		
Never/sometimes 26 33.3 $\overline{\text{TOTAL}}$ $\overline{78}$ $\overline{100.0}$ -Anal intercourse Always 9 40.9 Never 13 59.1 $\overline{\text{TOTAL}}$ $\overline{22}$ $\overline{100.0}$ -Oral sex Always 22 28.6 Never 55 71.4			
TOTAL 78 100.0 -Anal intercourse 9 40.9 Always 9 40.9 Never 13 59.1 TOTAL 22 100.0 -Oral sex 22 28.6 Never 55 71.4	•		-
-Anal intercourse 9 40.9 Always 9 40.9 Never 13 59.1 TOTAL 22 100.0 -Oral sex 22 28.6 Never 55 71.4	Never/sometimes	26	ز.زز
Always 9 40.9 Never 13 59.1 TOTAL 22 100.0 -Oral sex 22 28.6 Never 55 71.4	TOTAL	78	100.0
Never 13 59.1 TOTAL 22 100.0 -Oral sex Always 22 28.6 Never 55 71.4	-Anal intercourse		
TOTAL 22 100.0 -Oral sex	Always	9	
-Oral sex Always 22 28.6 Never 55 71.4	Never	13	59.1
Always 22 28.6 Never 55 71.4	TOTAL	22	100.0
Never 55 71.4	-Oral sex		
Never 55 71.4		22	28.6
TOTAL 77 100.0			71.4
	TOTAL	77	100.0

.

Table 4.7 (continued) Sexual relationship with clients

•

Variable	<u>N</u>	percent
Condom use in the last 6 months with clients (cont.)		
Masturbation (n=72)		
Always	4	5.6
Never	68	94.4
TOTAL	72	100.0
Usually ask the client to use a condom (n=109)		
Yes	88	80.7
No	21	19.3
TOTAL	109	100.0
Behaviour if the client refuses to use a condom (n=87)		
Lose the client	46	52.9
Have sex and worry later	17	19.5
Have sex and do not worry	15	17.2
Other	9	10.3
TOTAL	87	100.0

Table 4.7 (continued) Sexual relationship with clients

• •

Table 4.8 Condom use with clients and sharing injecting equipment

.

		- 1-	P	
Frequency condom use	Yes	No	Total	OR(95 %CI)
Vaginal intercourse				
Always	13	39	52(66.7 %)	1(Reference)
Never/sometimes	16	10	26(33.3 %)	4.8(1.6-15.0)
			78 (100 %)	
Anal intercourse				
Always	3	6	9(40.9 %)	l(reference)
Never/sometimes	6	7	13(59.1 %)	1.7(0.2-14.4)
			22 (100 %)	
Oral intercourse				
Always	7	15	22(28.6 %)	l(reference)
Never/sometimes	22	33	55(71.4 %)	1.4(0.5-4.6)
			77 (100 %)	

Equipment shared

6. Reproductive Health

The reproductive and sexual health of the sample population is summarised in Table 4.9. The majority of women (67.4 percent) reported having irregular periods, although 46.3 percent of women had not consulted a health professional about this. Most of the women (89.2 percent), suggested that their irregular periods were due to drug taking (for which there is a biological explanation due to hormonal changes). This could confirm the link between heroin and the female reproductive system and the need to investigate this area which has been largely unexplored until now. However, the gynaecological effects of HIV may also be an explanation, and further study is needed in order to explore the possible cause-effect relationship.

As noted above, a women's fertility is partly determined by her ability to conceive. A number of studies have reported that female IDUs have irregular menstrual cycles, an observation confirmed by the current data (Table 4.9). It is likely that high prevalence of irregular periods is caused by a number of interacting factors including the use of opiates (Bai et al., 1989) and other drugs, stress, weight gain (caused by obesity) and weight loss (caused by anorexia nervosa) (Symonds, 1987). The association between IDU and irregular periods was known by the majority of women who reported their disorder to be "because of drugs". It should, however, be noted that the cessation of menses, concomitant with the use of drugs (and opiates in particular), is due to amenorrhea and does not imply that a women is infertile (Ralph and Spigner, 1986). This has been illustrated by the reported increase in fertility of 201 opiate-dependent women on enrollment into a methadone maintenance programme in Liverpool, UK (Morrison et al., 1995).

Menstrual abnormalities have also been associated with HIV infection. In a study of 248 premenopausal HIV-infected women without AIDS and 82 HIV-negative women, the HIV-infected women were found to have an increase in amenorrhea (> 3 months) and in menstrual cycle intervals (> 6 weeks) and lower rates of premenstrual breast symptoms. Changes in menstrual function were also significantly associated with a past history of, but not current, substance abuse, suggesting the possibility that socioeconomic factors rather than biologic effects of drugs may be responsible (Chirgwin et al., 1996).

Three-quarters (75.7 percent) of women had ever been pregnant, although at the time of the interview only 4.6 percent of women were pregnant; 21.7 percent of women reported ever having a miscarriage. One-third experienced their first pregnancy before the age of 18. As a point of comparison, Spanish demographic indicators from 1995 estimate that the fecundity rate is 1.13 and that the median age of first birth is 28.6 (Demografia y Salud, 1996). In other words the study population appears to be experiencing pregnancy at a younger age than the general population.

Moreover, a number of studies have concluded that female IDUs are as likely to have children as the general population (reviewed in Wells and Jackson, 1992). The rate of reporting "ever being pregnant" (75.7 percent) is nearly identical to the 70.0 percent of "ever-pregnant" female IDUs reported from an American study (Brown, 1989). In a comparison of European countries, teenage pregnancy rates in Spain (31.3 percent of women were first pregnant between 15-17 years) were similar to Germany (30.2 percent) and England (29.9 percent), though lower than France (40.9 percent) and higher than Italy (16.1 percent) (Estébanez, 1996).

There are a number of reasons why female IDUs have a higher rate of pregnancy and, in particular, teenage pregnancy. For example, female IDUs often lead a somewhat "traditional" lifestyle with low levels of educational attainment and employment (Wells and Jackson, 1992). Self-esteem is probably low, and it may be perceived that childbirth may break this cycle. Although the current data do not provide evidence to examine this hypothesis, the inclusion of a question on desired family size in future studies may be of use.

Rates of miscarriage, however, appear to be similar to those reported in other countries. For example, in a household survey of 15 to 59 year olds in the UK conducted between May 1990 to November 1991, 20.7 percent of all women reported a lifetime experience of a miscarriage (Wellings et al., 1994). In Italy, for 1993, the ratio of the number of miscarriages to live-births was 26.2 percent (Statistiche della Sanita, 1995).

There was a higher rate of voluntary abortions amongst this sample of female IDUs than the population *per se.* For example, 39.1 percent of women reported ever having an abortion (Table 4.9). Reasons given for voluntary pregnancy termination included : "Because of drugs" (18.6 percent); "Because I do not want a child/more children" (14.4 percent); "It was too much responsibility" (25.4 percent); and, "Because I am HIV+" (22.0 percent). The highest reported lifetime experience of abortion was 15.4 percent amongst women aged 35 to 44 (Johnson et al., 1994). In Italy, the rate for voluntary abortions was 10.4 percent (Statistiche della Sanita, 1996). This rate should be considered in light of the findings of a recent study which found that rates of pregnancy termination doubled after diagnosis with HIV (Stephenson et al., 1996). It should be noted that there is a considerable deficit in the scientific literature with regard to female IDUs and pregnancy termination and that the data from the current study are almost unique for Europe.

Close to 45 percent of those interviewed regularly have a smear test. Over 16 percent have had an abnormal result from a smear test at some time in their lives.

There was a high prevalence (28.2 percent, i.e., 100 percent minus 71.8 percent; Table 4.9) of STD amongst female IDUs. The most common self-reported STD in the last year was candidiasis (15.8 percent), followed by genital warts (10.3 percent) and herpes (8.4 percent). Finally, hepatitis B and C were the most commonly reported infectious diseases (25.4 percent and 31.3 percent respectively), although the reliability of these data along with the STD information has to be queried in light of its self-reported nature.

Compared to other studies, there are variations between the prevalence and incidence of STDs probably reflecting social, geographical and climatic factors. For example, in a European seroprevalence study of HIV among patients with STDs (for both men and women) the most frequently reported diagnosis was genital warts. The second most frequent diagnosis was male urethritis followed by chlamydia infections (European Working Group on HIV Infection in Prostitutes, 1993). In a sample of 220 heterosexual female IDUs from Sydney, Australia, 1.4 percent reported ever having syphilis in their lifetime (Ross et al., 1991). More importantly, the viral STD may be significantly associated with HIV, with injecting drug use facilitating this process (Musicco et al., 1994).

The role of STDs in transmitting HIV cannot be underestimated. For example, there is significant evidence that STDs are the major risk co-factor associated with the HIV epidemic (Beck et al., 1996). It is likely that conditions that increase the number of lymphocytes in the

female genital tract, such as STDs that elicit an inflammatory response, may potentiate the risk of HIV transmission by increasing the pool of infected cells in a seropositive person or the pool of target cells in a seronegative person (Kreiss et al., 1988). In addition, the high-risk sexual behaviour of IDUs may further exaggerate this problem, although to date the literature on the relationship between drug or alcohol use during sexual activity and the likelihood of participating in high-risk sexual activity is inconsistent (Stall and Leigh, 1994). Nevertheless, in examining STDs in a sample of 1245 IDUs (for both sexes) in Sydney, Australia, Ross et al., (1991) conclude that the lifetime prevalence of STDs in IDUs is high and that previously recognised gradients of STDs across sexual orientation, with homosexual men having generally higher reported STD rates than heterosexual men, with bisexual men at intermediate risk, is also For women the pattern was generally consistent with the finding that fewer apparent. homosexual women would have STDs than heterosexual women. Finally, it is worth emphasising the relationship between human papillomavirus (HPV) infection of the female genital tract, cervical cytology, and human immunodeficiency virus (HIV): Increased rates of positive smear tests among a growing population of young HIV-infected and immunocompromised women may herald an increase in cervical carcinomas in this population (Feingold et al., 1990). Indeed a high rate of invasive cervical cancer among female IDUs with AIDS has been reported in Italy (Serraino et al., 1996).

As illustrated in Tables 4.9 and 4.10, the type of contraception used by IDUs differs from the total female population. In the previous five years, a large number had relied on withdrawal as a means of contraception (40.4 percent). For female IDUs, condoms were the most popular form of contraception; 84.9 percent reported their use. The pill, which is an inappropriate method for IDUs due to their hepatic pathology, was used by 42.3 percent of women. This compares to a survey of a representative sample of Spanish women aged 15 to 49 from 1997 in which the most utilized contraceptive methods were: condoms (21 percent); the pill (14.26 percent); IDU (5.68 percent); tubal ligation (5.20 percent); withdrawal (1.49 percent); rhythm (0.87 percent); diaphram (0.29 percent); spermicide (0.10 percent); and others (0.19 percent). Over fifty percent of the population reported using no methods of contraception (Schering España, 1997).

It has been suggested that IDUs have a lower uptake of contraceptives due, amongst other factors, to the (partly misguided) belief that opiates and other drugs cause infertility (Ralph and Spigner, 1986). In the five years preceding the current survey only 4.3 percent of women reported using no form of contraception (Table 4.9). This finding is in contrast to other studies which have noted that the use of contraceptives tends to be lower amongst female IDUs than the population *per se.* For example, in a study of 115 heroin-addicted women from California, USA, 25.8 percent reported any type of contraception compared to 48.5 percent of a national sample (Ralph and Spigner, 1986). Likewise, in the study of 201 opiate users from Liverpool, UK, 21.4 percent of women reported condom use compared to 46.5 percent of young women attending British Family Planning Clinics (Morrison et al., 1995).

Efficacy, safety, and acceptability are all factors which need to be considered when prescribing a contraceptive method. Further, these factors need to be matched to individuals and communities (Williamson and Joanis, 1994); that is, whilst one method may be appropriate for the general population, it may also be inappropriate for IDUs and other marginalised groups. Thus, in light of the high prevalence of hepatitis amongst IDUs (Table 4.9), the contraceptive pill may be an inappropriate method. Despite this, and as noted above, 29.9 percent of women surveyed reported using the contraceptive pill (Table 4.9). It has also been suggested that the use of

hormonal contraception may be associated with a reduction in the use of barrier contraception such as the condom. Whilst there is little evidence of this in the current data (81.3 percent of women report condom use; Table 4.6), another complicating factor is the concurrent use of petroleum-based lubricants. A recent European study identified the use of petroleum-based lubricants to be a risk factor for HIV infection in non-IDUs (European Working Group on HIV Infection in Fernale Prostitutes, 1993). Another issue to consider is the "dual" use of barrier contraceptives. A number of studies have suggested a reduction in the rates of sexually transmitted diseases (STDs), including HIV, for women who use condoms with a spermicide (reviewed in Roddy, 1994). Whilst the evidence is not yet conclusive, it is important to stress that both methods must be used concurrently (Cain and Williamson, 1994).

Because of drugs 165 89.2 Is my constitution 6 3.2 Other 14 7.6 TOTAL 185 100.0 Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No 74 24.3 TOTAL 304 100.0	Variable	N	percen
Yes 99 32.6 No 205 67.4 TOTAL 304 100.0 They have consulted a health professional about this 109 53.7 No 94 46.3 TOTAL 203 100.0 Why they think the periods are not regular 203 100.0 Why they think the periods are not regular 6 3.2 Other 14 7.6 TOTAL 185 100.0 Fype of contraception used in last 5 years (percentage using each contraception) Condoms Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No 74 24.3 TOTAL 304 100.0 Ster been pregnant 230 75.7 No 74 24.3 TOTAL 304 100.0			
No 205 67.4 TOTAL 304 100.0 They have consulted a health professional about this 109 53.7 No 94 46.3 TOTAL 203 100.0 Why they think the periods are not regular 203 100.0 Why they think the periods are not regular 89.2 15 89.2 Is my constitution 6 3.2 Other 14 7.6 TOTAL 185 100.0 Viderawal 123 40.5 Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No 74 24.3 TOTAL 304 100.0 Ster been pregnant 304 100.0 Are 304 100.0			
TOTAL 304 100.0 They have consulted a health professional about this 94 46.3 Yes 109 53.7 No 94 46.3 TOTAL 203 100.0 Why they think the periods are not regular Because of drugs 165 89.2 Is my constitution 6 3.2 Other 14 7.6 TOTAL 185 100.0 Fype of contraception used in last 5 years (percentage using each contraception) Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 100 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant Yes 230 75.7 No 74 24.3 TOTAL 304 100.0 304 100.0 Aver been pregnant now Yes 14 4.6 No 290 95.4	· · · · ·		32.6
They have consulted a health professional about thisYes10953.7No9446.3TOTAL203100.0Why they think the periods are not regular203100.0Because of drugs16589.2Is my constitution63.2Other147.6TOTAL185100.0Fype of contraception used in last 5 years (percentage using each contraception)207Condoms24781.3Withdrawal12340.5Contraceptive pill9129.9Spermicides4514.8IUD216.9Cap/diaphragm103.3Depro-provera injection62.0No7424.3TOTAL304100.0Are pregnant304100.0Yes144.6No29095.4	No	205	67.4
Yes 109 53.7 No 94 46.3 TOTAL 203 100.0 Why they think the periods are not regular 203 100.0 Why they think the periods are not regular 6 3.2 Because of drugs 165 89.2 Is my constitution 6 3.2 Other 14 7.6 TOTAL 185 100.0 Fype of contraception used in last 5 years (percentage using each contraception) Condoms Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No 74 24.3 TOTAL 304 100.0 Are pregnant now Yes 14 4.6 No 290 95.4 95.4	TOTAL	304	100.0
No 94 46.3 TOTAL 203 100.0 Why they think the periods are not regular 203 100.0 Because of drugs 165 89.2 Is my constitution 6 3.2 Other 14 7.6 TOTAL 185 100.0 Fype of contraception used in last 5 years (percentage using each contraception) 100.0 Condoms 247 81.3 Withdrawal 123 40.5 Contraception used in last 5 years (percentage using each contraception) 0.0 Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Cver been pregnant 74 24.3 TOTAL 304 100.0 Are preg	They have consulted a health professional about	this	
TOTAL $\overline{203}$ $\overline{100.0}$ Why they think the periods are not regular $\overline{203}$ $\overline{100.0}$ Because of drugs165 89.2 Is my constitution6 3.2 Other147.6TOTAL $\overline{185}$ $\overline{100.0}$ Cype of contraception used in last 5 years (percentage using each contraception)Condoms 247 81.3 Withdrawal123 40.5 Contraceptive pill9129.9Spermicides4514.8IUD216.9Cap/diaphragm103.3Depro-provera injection62.0No7424.3TOTAL $\overline{304}$ $\overline{100.0}$	Yes	109	53.7
Why they think the periods are not regularBecause of drugs16589.2Is my constitution63.2Other147.6TOTAL185100.0Condoms24781.3Withdrawal12340.5Contraception used in last 5 years (percentage using each contraception)Condoms247Condoms247Spermicides45IUD21Cap/diaphragm10Joo3.3Depro-provera injection6Viver been pregnant13Yes230TOTAL304TOTAL304Yes144.6No29095.4	No	94	46.3
Because of drugs 165 89.2 Is my constitution 6 3.2 Other 14 7.6 TOTAL 185 100.0 Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No 74 24.3 TOTAL 304 100.0	TOTAL	203	100.0
Is my constitution 6 3.2 Other 14 7.6 TOTAL 185 100.0 Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant Yes 230 75.7 No 74 24.3 TOTAL 304 100.0 Are pregnant now Yes 14 4.6 No 290 95.4	Why they think the periods are not regular		
Other 14 7.6 TOTAL 185 100.0 Type of contraception used in last 5 years (percentage using each contraception) 247 81.3 Condoms 247 81.3 123 40.5 Condoms 247 81.3 123 40.5 Condoms 247 81.3 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 3 4.3 Ever been pregnant Yes 230 75.7 No 74 24.3 70.0 74 24.3 TOTAL 304 100.0 30.4 100.0		165	89.2
TOTAL185100.0Type of contraception used in last 5 years (percentage using each contraception) Condoms24781.3Withdrawal12340.5Contraceptive pill9129.9Spermicides4514.8IUD216.9Cap/diaphragm103.3Depro-provera injection62.0No contraception used134.3Ever been pregnant23075.7No7424.3TOTAL304100.0Are pregnant now Yes144.6No29095.4	Is my constitution	6	3.2
Type of contraception used in last 5 years (percentage using each contraception)Condoms24781.3Withdrawal12340.5Contraceptive pill9129.9Spermicides4514.8IUD216.9Cap/diaphragm103.3Depro-provera injection62.0No contraception used134.3Ever been pregnantYes23075.7No7424.3TOTAL304100.0Are pregnant now144.6No29095.4	Other	14	7.6
Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant 13 4.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4	TOTAL	185	100.0
Condoms 247 81.3 Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant 13 4.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4	Type of contraception used in last 5 years (percer	ntage using each contra	acention)
Withdrawal 123 40.5 Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant 13 4.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4			
Contraceptive pill 91 29.9 Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant 230 75.7 No 74 24.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4	Withdrawal		
Spermicides 45 14.8 IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant 230 75.7 No 74 24.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4			
IUD 21 6.9 Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant 230 75.7 No 74 24.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4			
Cap/diaphragm 10 3.3 Depro-provera injection 6 2.0 No contraception used 13 4.3 Ever been pregnant 230 75.7 No 74 24.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4	•		
Depro-provera injection62.0No contraception used134.3Ever been pregnant 230 75.7No7424.3TOTAL $\overline{304}$ $\overline{100.0}$ Are pregnant now 14 4.6No 290 95.4			
No contraception used 13 4.3 Ever been pregnant 230 75.7 No 74 24.3 TOTAL $\overline{304}$ $\overline{100.0}$ Are pregnant now 14 4.6 No 290 95.4			
Yes 230 75.7 No 74 24.3 TOTAL $\overline{304}$ $\overline{100.0}$ Are pregnant now 14 4.6 No 290 95.4		-	
No 74 24.3 TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4	Ever been pregnant		
TOTAL 304 100.0 Are pregnant now 14 4.6 No 290 95.4	Yes	230	75.7
Yes 14 4.6 No 290 95.4	No	74	24.3
Yes 14 4.6 No 290 95.4	TOTAL	304	100.0
No 290 95.4	Are pregnant now		
TOTAL 304 100.0	No	290	95.4
	TOTAL	304	100.0

Table 4.9 Selected variables describing reproductive and sexual health

,

Variable		N	percent
Ever had a miscarriage			
Yes		66	21.7
No		238	78.3
TOTAL		304	100.0
Ever had a voluntary abortion			
Yes		119	39.1
No		185	60.9
TOTAL		304	100.0
Have a regular smear test			
Yes		134	44.4
No		168	55.6
TOTAL		302	100.0
Ever had an abnormal smear test			
Yes		49	16.2
No		253	83.8
TOTAL		302	100.0
Self-reported STDs in last year (percentage for each ST	ΓD) (N=27	/3)	
Candidiasis	<i>, , ,</i>	4 3	15.8
Genital warts		28	10.3
Herpes		23	8.4
Syphilis	13	4.8	
Chlamydia		7	2.6
Gonorrhea		5	1.8
Salpingitis/PID		0	0.0
Other STD		4	1.5
No STD		196	71.8
Self-reported infectious diseases in last year (percentag	e for each	disease)	(N=284)
Hepatitis A		8	2.8
Hepatitis B		72	25.4
Hepatitis C		89	31.3
Hepatitis D		2	0.7

Table 4.9 (continued) Selected variables describing reproductive and sexual health

.

7. Service Use

Before reviewing the results of the survey with regard to services used (Tables 4.10, 4.11 and 4.12), it is important to be aware of the limitations of a comparative analysis such as this. As the data were collected from a convenient sample of female IDUs, information on awareness and attitudes may simply reflect differences in the sample population studied. For example, if some female IDUs are interviewed whilst attending a drug dependency clinic, it would not be surprising that awareness of this service was high. This being said, it is important to review this data because drug-dependent women are at an increased risk for a variety of medical and social problems and because this burden is shared by a wide variety of public agencies.

The majority of female IDUs reported using health services in the last six months. This is contrary to the perception that, although women seek medical care more readily than men, female IDUs are confronted with unique barriers making them less likely to use such facilities (Weisner and Schmidt, 1992). For those women who did not use health services in the past six months, 67.2 percent stated that they "Did not need them". Over half (i.e., 78.1 percent) of the population reported using GP or primary health services, followed by casualty services (41.4 percent), hospitals (32.1 percent) and AIDS screening (25.7 percent). The remainder of health services listed in Table 4.10 were used, on average, by less than ten percent of female IDUs.

The number of female IDUs reporting that the health services they used were "very helpful" was 26.5 percent. However, only 57.7 percent of the sample considered the health services used to be "accessible". The major inference to be made from these findings is that female IDUs find atypical health services to be more accessible than standard services. Thus it would be appropriate for female IDUs to be a target group for specific medical care (Dawe, Gerada and Strang, 1992), and for treatment to be available from a wide variety of public health services (Weisner and Schmidt, 1992). It should also be noted, that in a comparison of IDU and homosexual men in the USA, no difference between recorded satisfaction was found, thus indicating that gender-specific differences may be limited (Johnston, 1994).

As with the analysis of health services, the type of drug service provided will partly depend on how the sample population was recruited. However, as primary care for IDUs with HIV/AIDS may be the responsibility of medical personnel and hospital staff, drug-abuse treatment programmes will involve a multi-disciplinary team of individuals and agencies (Mondanaro, 1987).

Finally, the provision of social services to both IDUs and HIV-positive women is imperative. These women are subject to an interplay of psycho-social factors which require the attention of many different agencies (Harrison, 1989). As part of a multi-disciplinary approach, health and drug services should be supplemented by complementary social services (Glass, 1993). Despite these requirements, the current study identifies alarmingly low utilisation of social services by female IDUs. Only 25.7 percent of women had used these services in the past 6 months (Table 4.12). Moreover, the quality and accessibility of social services was, generally speaking, lower than the health and drug services. It seems likely that some of the non-utilisation is caused by administrative problems and lack of accessibility.

Drug-using women have multiple health-care, drug-treatment and social-service needs. A multidisciplinary, multi-agency approach is needed to meet these requirements. However, policy makers must be aware that many women may believe that these services do not meet their needs and it is therefore important that gender-specific policy be implemented. Indeed, the preceding review of the provision of health, drug, and social services demonstrates that female IDUs utilise a wide range of different facilities. Because drug-dependent women are at an increased risk from a variety of medical and social problems (Mondanaro, 1987), it is appropriate that policy should focus on a range of settings including, for example, medical centres, mobile drug units, the criminal justice system, welfare agencies, etc., (Weisner and Schmidt, 1993). In implementing such a policy it is important the providers are aware that treatment is available from a number of sources and that low utilisation may not necessarily impact on the effectiveness of a particular service (Weisner and Schmidt, 1992). It is encouraging to note the relatively high utilisation of services by female IDUs in Europe (Estébanez, 1996). Indeed this contradicts perceived opinion which may have been formulated from studies based in the USA. The fact that European countries predominantly operate public (as opposed to private) health services may qualify a number of findings made based on the American system.

Policy discussion should consider the nature and quality of health, drug and social services utilised by female IDUs, and programmes must assist drug-dependent women to escape from a passive role (Mondanaro, 1987). Again, these data seem quite encouraging in this regard in light of the low number of women who reported that they "did not care" when asked why they did not utilise a specific service. Furthermore, treatment goals should continue to be based on a policy of harmreduction with the specific aim of retaining women in any treatment/maintenance programme.

Less encouraging is the heterogeneity of services, emphasising the need for a European programme of standardisation. This is especially important for social policy which, according to these data, has failed in the delivery of services to female IDUs and HIV women in Europe (Estébanez, 1996). Likewise, any European (or Spanish) policy should be specifically targeted at women as these data suggest that policy to date is non-gender-specific (hence the low provision of family planning services).

Variable	N	percen
Ever used health services in last 6 months		
Yes	237	78 .0
No	67	22.0
TOTAL	304	100.0
Why never used health services		
I did not need them	45	67.1
I did not care	·· 9	13.4
Too far away	4	6.0
Administrative problems	4	6.0
Personal or social problems	3	4.5
No medical insurance	2	3.0
TOTAL	67	100.0
Health services used (percentage using each servic	ce) (n=237)	
GP or primary health care	185	78.1
Out-patient clinic	47	19.8
Casualty hospital	98	41.4
Hospital	76	32.1
Family planning	9	3.8
STD clinic	7	3.0
Psychiatric settings	6	2.5
Private sector	30	12.7
AIDS screening	61	25.7
Quality of health services used (n=237)		
Useless	23	9.7
Not very helpful	36	15.2
Helpful	116	49.0
Very helpful	62	26.1
Accessibility of health services (n=237)		
Not accessible	25	10.5
		<i>ca</i> 0
Accessible	137	57.8

Table 4.10 Type of health services used by pooled female IDUs

.

.

Variable	<u>N</u>	percent
Even used drug sorriges during last (marthe		
Ever used drug services during last 6 months Yes	240	00.0
No	249 54	82.2
140	54	17.8
TOTAL	303	100.0
Why never used drug services		
I did not need them	24	44.4
I did not care	13	24.1
No confidentiality	9	16.7
Too far away	4	7.4
Personal or social problems	3	5.5
Administrative problems	1	1.9
TOTAL	54	100.0
Drug services used (percentage using each services	ce) (n=249)	
Mobile drug-dependency unit	178	71.5
Drug dependency units	94	37.8
Out-patient rehabilitation centre	56	22.5
Methadone distribution centre	54	21.7
Therapeutic community	31.	12.4
Residential rehabilitation centre	15	6.0
NGO, self-help groups	9	3.6
Juality of drug services used (n=249)		
Useless	13	5.2
Not very helpful	25	10.0
Helpful	125	50.1
Very helpful	86	34.6
ccessibility of drug services (n=249)		
ccessibility of drug services (n=249) Not accessible	29	11.6
	29 120	11.6 48.2

Table 4.11 Type of drug services used by pooled female IDUs

Variable	N	percent
Ever used social services during last 6 months		
Yes	78	25,7
No	225	74.3
TOTAL	303	100.0
Why never used social services		
I did not need them	89	40.1
I did not care	43	19.4
Administrative problems	. 38	17.1
Service refused	19	8.6
Personal or social problems	11	5.0
No confidentiality	10	4.5
Long waiting list	6	2.7
Too far away	3	1.3
No social benefit	3	1.3
TOTAL	222	100.0
Social services used (percentage using each service) ((n=78)	
Children support services	36	46.2
Services providing financial help	19	24.4
Housing services	18	23.1
Support specific for drug users	15	19.2
Legal assistance	15	19.2
Quality of social services used (n=78)		
Useless	3	3.9
Not very helpful	10	12.8
Helpful	37	47.4
Very helpful	28	35.9
ccessibility of social services (n=78)		
Accessibility of social services (n=78) Not accessible	15	10 2
Accessibility of social services (n=78) Not accessible Accessible	15 49	19.2 62.8

.

Table 4.12 Type of social services used by pooled female IDUs

8. Conclusion

The foremost conclusion to be drawn from this descriptive analysis of the characteristics of female IDUs is the low socioeconomic characteristics. Moreover, it is also worth noting that the population is pretty homogenous; that is (on average), the sample of female IDUs examined represents a very marginalised population located in Madrid. This has major implications for the forthcoming analysis of HIV prevalence (Chapter 5), drug use and prostitution (Chapter 6), and drug use and prisons (Chapter 7). It is therefore essential that the reader is consistently aware of the exceptional nature of the sample population. That is, in analysing the results of this project, the applicability of research and policy conclusions has to be put in context.

Chapter Five

Determinants of HIV amongst Female IDUs

1. Introduction

HIV prevalence in Spain is one of the highest in Europe, as noted in Chapter 1. A study of drug users receiving ambulatory treatment in Spain (the majority of which were men), reported significant variation in HIV seropositivity by the 17 autonomous regions (Lardelli et al., 1993). For example, the lowest rate was for Navarre (14.3 percent) and the highest rate in La Rioja (66.7 percent). The rate in the community of Madrid was 46.3 percent. The current study found an HIV prevalence of 52.6 percent amongst the 304 female IDUs (see Table 5.1). After adjusting for the 15.5 percent of women with unknown test results, the headline prevalence rate of HIV amongst female IDUs was 62.3 percent (for n = 257). Although this rate appears to be surprisingly high, one needs to reconsider the population from whence it came. This is an extremely marginalised group of women with a multitude of possible risk factors, living in the largest metropolitan area of a country with one of the highest rates of HIV in Europe. In fact, this study has perhaps succeeded in capturing a sample of a population which most likely represents one of the highest risk subgroups in Europe.

The purpose of this chapter is to explore the multitude of possible risk factors associated with HIV in this population of female IDUs and to develop a model to explain and/or examine direct and indirect determinants of HIV.

2. Methods

The methods used for this study are described in Chapter 3. The sample size for this chapter is 257, as 47 cases were removed because the HIV status was unknown. In a majority of cases (97.3 percent) HIV status was determined from a medical report.

First, a bivariate analysis will be conducted to identify significant relationships between HIV and a variety of independent variables. As concluded in Chapter 4, the sociodemographic characteristics of female IDU describe a marginalised population with a low socioeconomic status. Whilst none of the significant risk factors associated with HIV infection are known to have a direct causal effect, the status of the population acts as a generic proximate determinant of HIV infection. Through the analysis of specific risk-taking and risk-avoiding behaviours it may be possible to identify specific proximate determinants of HIV infection. In doing so, it is important to be aware of the limitations of bivariate analysis where, for example, specific risk factors may be either confounding and/or interacting with other risk factors.

The only variation from the methods described in Chapter 3 occurs in the logistic regression model in which the following five groups of explanatory variables have been identified:

Group 1: Age

Biological age is the first explanatory variable entered into the model because of its known relationship with HIV, as well as its influence on other independent variables.

Group 2: Socioeconomic status

Three variables were selected to define the general socioeconomic status of female IDUs: age left full-time education; source of income during past six months (illegal and prostitution *versus* legal); and, place of residence in past six months (no fixed abode *versus* abode). Each of these factors has been identified in prior research as having an influence on HIV status (see literature review in Chapter 1).

Group 3: Marginalisation

In addition to socioeconomic status, three variables were chosen to characterise the extreme marginalised nature of the sample population. These dichotomous variables were: previous imprisonment; having a regular partner who is an IDU; and prostitution. A number of earlier studies have identified these variables as potential determinants of HTV status (see Chapter 1).

Group 4: Sexual and drug-taking risk behaviour

Specific known risk behaviours of HIV infection were included in the fourth group. These included: the average number of clients a female IDU has had per month for the past six months (coded into four categories: never, i.e., not a prostitute; no clients, i.e., an ex-prostitute; 1 to 60 clients; and, 60+ clients); age at first injection; condom use; cocaine injection; heroin+cocaine injection; and a regular HIV-positive sexual partner. All of these risk behaviours were selected based on evidence from previous research of their potential or known influence on HIV status.

Group 5: Co-infection

The final group contained two dichotomous variables which are known to facilitate the transmission of HIV: hepatitis B and other sexually transmitted diseases (STDs). There is a variety of evidence to suggest that previous STDs are independently associated with HIV infection (see Chapter 1).

3. Results

First the bivariate relationships with HIV and selected independent variables are examined (see Table 5.2). Then findings from the regression analyses will be presented.

3.1. Sociodemographic characteristics and HIV prevalence

Several of the sociodemographic variables had significant relationships with HIV prevalence. First, there was a significant positive association with age at time of interview and HIV prevalence. For example, HIV prevalence was 49.1 percent for women aged 24 or less compared to 68.1 percent for women aged 35 or more. The linear trend test was significant. This association is most likely a function of exposure, as opposed to a direct determinant of HIV.

Second, those women who left full-time education before they were 13 years old (only 29.1 percent; Table 4.1; Chapter 4) were twice as likely to be HIV-positive than those who left school at age 16 or older.

Imprisonment since first injection was also significantly associated with HIV prevalence, with those who had been in prison since their first injection nearly three times more likely to be HIV-positive.

Finally, not having a fixed address also nearly tripled the risk for HIV infection.

3.2 Drug-using habits and HIV prevalence

Age at first injection was significantly negatively associated with HIV prevalence. Those who started injecting by age 16 were three times more likely to be HIV-positive than those who started at age 20 or later.

Needle and/or syringe sharing was also significantly related to HIV prevalence. More surprising was a significant negative association found between needle sharing in the last six months and HIV prevalence. In other words, it would seem that those women who were aware that they were HIV-positive have ceased sharing needles.

Finally, it is worth mentioning that of the 89 women who reported having been in prison, 34.8 percent reported injecting whilst in prison and 87.1 percent of these were HIV-positive. Although the relationship did not attain statistical significance (Table 5.2), it is worth noting that the very high rate of HIV amongst female IDUs who injected drugs in prison probably reflects a cumulative aggregation of risk factors. The association between female IDU and previous imprisonment will be explored further in Chapter 6.

3.3. Sexual habits and HIV prevalence

Significant relationships were found with HIV prevalence for ever having had an HIV sexual partner, a casual sexual partner, a casual IDU sexual partner, a casual HIV sexual partner, and a client.

The frequency of condom use for vaginal intercourse with a regular sexual partner was negatively associated with the dependent variable, whereas no significant relationships were found for condom use with a casual sexual partner or casual client. The latter finding suggests that those who are HIV-positive are more inclined to always use condoms with their regular partners.

3.4 Reproductive health and HIV prevalence

The information collected on the relationship between reproductive and sexual health and HIV infection amongst female IDUs is, as far as we know, almost unique. Moreover the sample population is consistent with the atypical characteristics of a marginalised population. The majority of women reporting not having regular periods are HIV-positive. Another interesting result, which would also require further research to identify a causal effect, was the women who reported an abnormal smear test. These women also reported a higher prevalence of HIV which may have arisen due to the higher risk of HIV infection for women with cancerous cells.

A number of studies have suggested that pregnancy may be a risk factor for HIV infection. Although as yet there is no consensus, the bivariate analysis suggests that "ever-pregnant" women are 2.2 times more likely to be infected by HIV.

Another important observation with regard to contraceptive use, was the high (although nonsignificant) rate of HIV prevalence amongst women who use spermicides. This confirms a previous observation which speculated that the use of petroleum-based products has a detrimental effect on effectiveness of condoms as a barrier contraceptive and thus is associated with an increased risk in HIV infection (European Working Group on HIV Infection in Female Prostitutes, 1993).

Finally, there was a high prevalence of STDs amongst female IDUs, which was significantly associated with HIV. For example, HIV prevalence amongst women who had no STD disease in the last year was 50.0 percent versus 81.9 percent who had a STD. Approximately one-quarter of the women interviewed reported having had hepatitis B, and this was also significantly related to being HIV-positive.

3.5. Results of hierarchical logistic regression analysis

The results of the first stage of the analysis can be found in Table 5.3. A number of observations can be made from these results. First, as illustrated by the significant decline in the -2 log likelihood (-2LL) value, each model is an improvement on the preceding model with the sole exception of Stage 1 (which measured the improvement - i.e., 7.89 - from the constant value to the addition of the age variable). The second observation regards the variables that are eliminated from the model. These are: source of income; regular partner who is an IDU; number of clients per month; condom use; heroin+cocaine injection; prostitution; age at first injection; and, hepatitis B infection. It is important to realise that although these variables are not included in the model (Group 5), a regular partner who is HIV-positive is by far the most important variable (i.e., OR = 12.2). Other significant factors revealed in the final model include: age 35 and over; no fixed address; ever shared needles; and co-infection with an STD.

It should be noted that although a direct association with HIV is often detected as a variable and is introduced in the model, thereafter this association seems to become indirect. For example, in Groups 1, 2 and 3 the age group 30-34 is a significant predictor of HIV serostatus (OR = 2.7, 3.7 and 3.1 respectively). However, for Groups 4 and 5, when other variables are introduced, the size and significance of the association declines.

4. Discussion

This chapter analyses the variety of factors including age, socioeconomics, marginalisation, risk behaviours, and co-infections that may be determinants of the HIV serostatus in an urban population of female IDUs. Before making any conclusions from these data, it is worth reiterating the limitations of this study. As noted in Chapter 3, a cross-sectional study is carried out at just one point in time and therefore it only measures prevalence as opposed to the incidence of a disease. Furthermore, as the period of time which has elapsed since infection is unknown, the data may be biased with regard to factors associated with disease progression. Another limitation to the adopted study design, was the necessity to use a convenience sample as opposed

to an equal probability scheme. Given these limitations it should also be noted that the multivariate logistics regression model does not examine risk factors (i.e., determinants) of HIV, but looks at statistical association in the data-set. For example, variables concerning risk behaviour during the whole injecting career (such as imprisonment) are included with variables concerning risk behaviour in the previous six months (such as shared needles/syringes in last six months).

Keeping in mind the caveats of this analysis described above, and the fact that HIV prevalence in Spain is one of the highest in Europe, a number of preliminary inferences can be made from these results.

First, it appears the sexual behaviour of these women is one of the strongest determinants of HIV infection. As unequivocally demonstrated in the regression analysis, having a partner who is HIV-positive increases the likelihood of HIV infection. Although the study design precludes us from making assumptions about directionality of this relationship, we believe it is more likely that the women are being infected by their male partners than the reverse based on scientific evidence suggesting that the relative efficiency of male-female transmission is two to three times more efficient than that of female-male transmission (Nicolosi et al., 1994a). Given that nearly 50 percent of this population of female IDUs reported having a regular partner who is HIV-positive, the importance of this factor cannot be underestimated. Clearly, one of the important conclusions to be made from this study is that the sexual partners of female IDUs are as important a component in explaining the HIV epidemic in this population as other risk factors, including high-risk drug-taking behaviour. Accordingly, and as discussed in further detail in Chapter 8, policy should emphasise this and provide information to female IDUs on reducing sexual risk behaviour. For example, the relatively low use of condoms with regular partners reported in Table 4.3 (Chapter 4), needs urgent attention.

Another sexual behaviour which warrants attention is prostitution. Although not an important factor in the regression analysis, sexual behaviour with a client did have a significant relationship with HIV prevalence, which has been demonstrated in previous studies in other western countries (Serraino et al., 1991; Zunzunegui et al., 1992). Another interesting, although non-significant, observation was the normal distribution with the average number of clients per month (not shown in Table 5.2). This has also been previously shown for prostitute populations in Amsterdam (Fennema et al., 1995) and Spain (Estébanez et al., 1996) where it has been suggested that after a certain point HIV prevalence decreases as women become more professional and reduce their risk- taking behaviour. It should also be noted that as mentioned above, HIV infection is more likely to be transmitted from men to women and that this pathway could be a contributory factor in HIV infection amongst female prostitutes (Nicolosi et al., 1994a).

The sharing of injecting equipment was also a significant determinant in the regression analysis. This finding was not too surprising, as a number of studies have demonstrated that the sharing of drug-injecting equipment is a principal risk factor associated with the transmission of HIV in IDU populations (e.g., Marmor et al., 1987; Des Jarlais, 1992; Nicolosi et al., 1992; Vlahov et al., 1990; Sasse et al., 1989; and Serraino et al., 1991). The fact that there is also a negative association between sharing injection equipment in the last six months and HIV prevalence is good news. Apparently those women who were aware that they were HIV-positive have ceased sharing needles. This is an observation that has been noted elsewhere (Delgado-Rodríguez et al. 1994), and emphasises the need for policy makers to provide easily accessible voluntary testing facilities to marginalised groups. A more in-depth look at the sharing of injection equipment in this population is undertaken in Chapter 8.

Age appears to have a direct as well as indirect influence on HIV status. One interpretation of this direct relationship found between age and HIV prevalence is that there is an exposure effect: the older an IDU gets the more likely she will become infected with the HIV virus. However,, it is also important to consider a recent study by Castilla et al. (1997), who found a relationship between specific birth cohorts and AIDS incidence amongst Spanish IDUs. The cohorts with the highest incidence of AIDS were born between the late-1950s to the late 1960s. The authors suggest that this peak in incidence possibly reflects the spread of injection drug use in Spain (particularly heroin) since the late 1970s. This peak roughly corresponds to the users in our study between the ages of 27 and 37, offering another possible explanation of the higher prevalence of HIV in the older age groups. However,, this finding should not direct interest away from the younger age groups which are experiencing higher rates of HIV incidence, particularly among heterosexuals (Houweling, 1998). The indirect influence of age on HIV prevalence will be studied further in subsequent chapters in which we will look at its relationship to prostitution, imprisonment, and treatment attendance.

The direct relationship between no fixed address and HIV prevalence is also an interesting result. Very few studies have looked at the relationship between homelessness and HIV prevalence in IDUs. Given that earlier studies have noted more HIV risk practices among the homeless (Fisher et al., 1995; Nyamathi et al., 1997), we anticipated an indirect relationship between homelessness and HIV prevalence with risk behaviours and prostitution acting as intervening variables. We were surprised to find in our model that the addition of "risk behaviour" actually improved the relationship between no fixed address and HIV prevalence, and the addition of "STD co-infection" only weakened the relationship slightly. Although more research needs to be done to replicate this finding in other IDU populations and to establish the directionality of this relationship, public health officials ought to consider that this may be an important subgroup of IDUs to target with HIV prevention interventions.

The association found between "ever being pregnant" and HIV (OR=2.2;CI=1.2 to 4.0; Table 5.2) is an important finding of this study. Although we did not find evidence of a direct relationship with HIV infection in this study, we do feel it is important to consider as an indirect determinant. The relationship between HIV infection and pregnancy has been the subject of various studies. It is possible that pregnancy may accelerate progression to AIDS through temporary immunosuppression during which women are at greater risk of contracting infections (Schoenbaum et al., 1988). A number of studies have concluded that female IDUs are as likely to have children as the general population (reviewed in Wells and Jackson, 1992). In the current sample, 75.7 percent of women reported "ever being pregnant" (Table 4.6). This is nearly identical to the 70.0 percent of "ever-pregnant" female IDUs reported from an American study (Brown, 1989). There are a number of reasons why female IDUs have a higher rate of pregnancy and, in particular, teenage pregnancy. For example, female IDUs often lead a somewhat "traditional" lifestyle with low levels of educational attainment and employment (Wells and Jackson, 1992). Self-esteem is probably low, and it may be perceived that childbirth may break this cycle. Although the current data do not provide evidence to examine this hypothesis, the inclusion of a question on desired family size in future studies may be of use.

A number of studies have also suggested that pregnancy itself may be a risk factor for HIV infection. Although as yet there is no consensus, the current data suggests that "ever-pregnant" women are 2.2 times more likely to be infected by HIV. What cannot not be disputed is the increase in the number of HIV-infected pregnant women. Indeed this is probably a reflection of the sociodemographic characteristics of HIV in women. In Europe, for example, 76 percent of HIV-infected females are in the 15-49-year age group. It is therefore important that policy makers and public health practitioners alike are aware of the effects of pregnancy on the natural history of

HIV. It has been suggested that HIV pregnant women may be susceptible to atypical immunological changes and that this would facilitate the progression from HIV to AIDS. To date, studies examining immunological changes have produced contrasting results. More evidence, However,, is available with regard to the effects of drug use. Whilst there seems to be no relationship between opiate use and disease progression, there is evidence that continued IDU may accelerate progression from HIV to AIDS. Although there is no evidence of gender-specific differences in the progression rate, the interaction of pregnancy and/or IDU with HIV-infected women makes this likely. The confused state of knowledge with regard to the reproductive health of pregnant women demonstrates the need for cohort studies to examine the association between HIV/AIDS and pregnancy.

Another important finding is the relationship between co-infection with an STD and HIV prevalence. As noted throughout the previous chapter, the relationship between STD and HIV is well reported. The fact that STD co-infection is a significant risk factor in the final logistic regression model (i.e., Model 5 in Table 5.3) of the multivariate analysis re-enforces this finding. As also noted previously, this is most likely due to the ease with which HIV can be transmitted following ulcerations, weakening of the tissues, and other physical changes associated with STDs (Holst et al., 1987). As discussed in Chapter 4, it has also been suggested that IDU facilitates this process (Musicco et al., 1994). However,, there is much discussion as to whether or not the relationship between HIV and STDs is a direct or indirect relationship (Beck et al., 1996). Direct transmission could be facilitated by breaks in the anatomical barriers due to ulcerating diseases, like genital ulcers and warts. Alternatively, since infected monocytes and macrophages constitute a main reservoir of HIV in infected individuals, they could act as transport vehicles for the virus within the infected host and between individuals. Thus effective and inexpensive containment strategies would include antimicrobial treatment for STDs and prevention of infection (via condom use) in non-infected individuals (Beck et al., 1996; Kreiss et al., 1988).

Previous imprisonment and injection whilst in prison are factors that need further attention. The results of the current study revealed that a third of the women who had been to prison injected drugs whilst in prison, and that this was associated with HIV infections (although the result was non-significant). A previous study into the prevalence of HIV amongst IDU prostitutes identified imprisonment as an important determinant (Estébanez et al., 1995). In that paper, it was questioned if previous imprisonment was acting as a generic marker for a marginalised population or if it was a causal risk factor of HIV infection. Likewise, a multifactorial analysis of 612 IDUs in Berlin, revealed that the most important risk factor for HIV infection was needle sharing in prison (Müller et al., 1995). Thus, regardless of whether there is a direct relationship between imprisonment and HIV prevalence, or imprisonment is merely acting as an indicator of high-risk behaviour whilst outside of prison, the fact that we have identified an association between HIV infection and imprisonment (and that there is also an association between HIV infection and injecting in prison) presents a major challenge to public health practitioners. Drugs are apparently readily available in prison whilst the availability of sterile syringes and needles and/or disinfectant is limited. Therefore it would seem that drug behaviour in prison is a factor which requires further research, and is examined further in Chapter 7.

Finally, this model needs to be validated on another independent group of women. Conceptual models of disease determination, such as the model presented here, are going to be of much use to policy makers.

5. Conclusions

- C Sexual behavior plays an important role in determining HIV serostatus in IDU women.
- C Those who continue to share injection equipment are at a high risk for contracting HIV, and those aware of their HIV-positive status appear to be discontinuing this practice.
 - C Co-infection with an STD is an important marker for those at risk for HIV.
 - C The roles of imprisonment and pregnancy in the transmission of HIV need to be further studied with prospective cohorts.
 - C It is important to note that the aim of this study was not to prove causal relationships with HIV, but to identify factors which can provide guidelines for the formulation of policy.

Variable	<u>N</u>	percen
Even been tested for UIN/ (N=20.1)		
Ever been tested for HIV (N=304)	202	
Yes	292	96.1
No	12	3.9
No response	0	0.0
Outcome of past HIV test (N=292)		
HIV +	166	54.6
HIV -	124	40.8
Not applicable/No response	. 12	4.6
Sample taken for current HIV test (N=304)		
Yes	7	2.3
No	297	97.7
Written confirmation of HIV test (N=304)		
Yes	257	84.5
No	47	15.5
Current HIV prevalence (N=304)		
HIV +	160	52.6
HIV-	97	31.9
Not known	47	15.5
Headline HIV prevalence (N=257) ^a		
HIV+	160	62.3
HIV-	97	37.7

Table 5.1 Selected variables describing HIV prevalence

a Headline rate excludes not-known current HIV prevalence results.

•

<u>Variable</u>	Total	HIV	······································	OR(CI.95 %	<u>)</u> p
		+ (%)	- (%)		
Sociodemographics					
Age					
24 or less	57	49.1	50,9	l(reference)	*a
25-29	89	61.8	38.2	1.7(0.8-3.5)	
30-34	69	72.5	27.5	2.7(1.2-6.1)	
35 or more	47	68,1	31.9	2.2(0.9-5.4)	
Age left education					
13 or less	74	73.0	27.0	2.2(1.1-4.3)	∗a
14-15	74	63.5	36.5	1.4(0.7-2.7)	
16 or more	112	55.4	44.6	l(reference)	
Been in prison since first	injection				
Yes	89	77.5	22.5	2.8(1.5-5.0)	***
No	173	55.5	44.5		
Abode					
No fixed address	57	78.9	21.1	2.7(1.3-5.3)	**
Fixed address	205	58.5	41.5		
Income					
Illegal+prostitution	122	67.2	32.8	1.4(0.8-2.4)	n.s.
Legal	140	59.3	40.7		
Drug-using habits					
Age began injecting drugs	;				
16 or less	74	78.4	21.6	3.1(1.5-6.4) *	** ^a
17 - 19	71	62.0	38.0	1.4(0.7-2.7)	
20 or more	117	53.8	46.2	l(reference)	
Ever shared needles/syring	ges				
Yes	203	70.9	29.1	4.3(2.3-8.0)	***
No	58	36.2	63.8		
Ever shared needles/syring	ges in last 6 mo	nths			
Yes	93	61.3	38.7	0.4(0.2-0.8)	**
No	111	78.4	21.6		
njected in Prison					
Yes	31	87.1	12.9	2.6(0.8-8.5)	n.s.
No	58	72.4	27.6		

Table 5.2 HIV prevalence^b by selected variables

Table 5.2 (continued)	HIV	prevalence by	y selected variables
-----------------------	-----	---------------	----------------------

Variable	Total	HIV		OR(CL95 %	<u>6) p</u>
		+ (%)	- (%)		
Sexual habits					
Ever had a regular sexual	l partner				
Yes	254	62.6	37.4	0.6(0.1-2.8)	n.s.
No	8	75.0	25.0	0.0(0.1-2.0)	11.5.
Ever had a regular sexual	IDU partner				
Yes	226	64.6	35.4	1.6(0.8-3.3)	n.s.
No	36	52.8	47.2	(
Ever had a regular sexual	HIV partner				
Yes	133	88.0	12.0	14.4(7.5-27.6) ***
No	116	33.6	66.4		,
Frequency of condom use	e with regular so	exual partner		·	
-Vaginal intercourse					
Never/sometimes	99	47.5	52.5	0.3(0.1-0.7)	**
Always	33	75.8	24.2		
Ever had a casual sexual j					
Yes	216	68 .1	31.9	3.3(1.7-6.4)	***
No	46	39.1	60.9		
Ever had a casual sexual l	•				
Yes	145	75.2	24.8	3.6(2.1-6.1)	***
No	111	45.9	54.1		
Ever had a casual sexual H	IV partner				
Yes	52	86.5	13.5	6.5(2.8-15.4)	***
No	149	49.7	50.3		
Frequency of condom use	with casual sex	cual partner			
Vaginal intercourse					
Never/sometimes	45	71.1	28.9	0.9(0.3-2.3)	n.s.
Always	38	73.7	26.3		
Ever had a client					
Yes	128	68.8	31.3	1.6(1.0-2.7)	*
No	134	57.5	42.5		
Frequency of condom use Vaginal intercourse	with casual clie	ent			
Never/sometimes	24	62.5	37.5	0.8(0.3-2.3)	n.s.
Always	48	66.7	33.3	0.0(0.3-2.3)	11.3.
·	0	···· /	22.2		

.

Variable	Total	HIV		OR(CI.95 %	<u>)</u> p
		+ (%)	- (%)		
Reproductive habits and	l infections				
Regular periods					
No	187	• 71.1	28.9	3.3(1.8-6.0)	**
Yes	75	42.7	57.3		
Age at first pregnancy					
< 18	62	71.0	29.0	1.3(0.7-2.5)	n.s
18 or more	136	65.4	34.6		
Contraception currently					
Yes	134	64.2	35.8	1.1(0.7-1.9)	n.s
No	124	61.3	38.7		
IUD used in last 5 years					
Yes	19	36.8	63.2	0.3(0.1-0.8)	*
No	243	65.0	35.0		
Condom used in last 5 yea	rs				
Yes	223	65.0	35.0	1.8(0.9-3.5)	n.s.
No	39	51.3	48.7		
Spermicide used in last 5 y	ears				
Yes	41	73.2	26.8	1.7(0.8-3.6)	n.s.
No	221	61.1	38.9		
Contraceptive pill used in I	ast 5 years				
Yes	76	38.2	61.8	0.2(0.1-0.4)	***
No	186	73.1	26.9		
Ever been pregnant					
Yes	202	67.3	32.7	2.2(1.2-4.0)	**
No	60	48.3	51.7		
Abnormal smear test					
Yes	44	75.0	25.0	2.6(1.2-5.6)	*
No	142	53.5	46.5		
STD in last 5 years					
Yes	72	81.9	18.1	4.5(2.2-9.5)	***
No	162	50.0	50.0		
Hepatitis A in last 5 years					
Yes	8	100.0	0.0		*
	238	60.5	39.5		

Table 5. 2 (continued) HIV prevalence by selected variables

•

Variable	Total	HIV		OR(CL95 %)	р
		+(%)	- (%)		Ľ
Hepatitis B in last 5 years					
Yes	67	76.1	23.9	2.5(1.3-4.7) *	: *
No	179	56.4	43.6		
Hepatitis C in last 5 years					
Yes	84	59.5	40.5	0.9(0.5-1.5) n	.S.
No	162	63.0	37.0	(10 10)	
Hepatitis D in last 5 years					
Yes	2	100.0	0.0	— n	.S.
No	244	61.5	-38.5		
Tuberculosis in last 5 years					
Yes	45	86.7	13.3	5.1(2.1-12.5) **	**
No	201	56.2	43.8	·····,	
Some infection last year					
Yes	139	68.3	31.7	1.9(1.1-3.3) *	
No	107	53.3	46.7	· · · ·	
Type of Treatment					
Methadone maintenance	64	79.7	20.3	4.7(2.0-10.9)***	
Other treatment	128	64.1	35.9	2.1(1.1-4.0) *	
No treatment	70	45.7	54.3	1.0(reference)	

Table 5.2 (continued) HIV prevalence by selected variables

a: p value of linear trend test.

b: HIV prevalence is the headline rate presented in Table 5.1 and excludes not-known cases. Figures presented as percentages.

¢ use, nervine+cocarne injection and age girst injection; stage 5: Hepatitis B and genital warts.

Cilents per month. Condom use. Heroine+cocaine injection and ape direct injection. Stane 5. Henatitic R and denital warts
c) For models.B.C.D and E the following variables were excluded from the equation. Stage 2:Income: Stage 3: Regular sexual partner IDU and prostitution: Stage 4: Number of
b) * p<0.05; ** p<0.01; *** p<0.001.
eliminated, whilst the new socioeconomic variables were selected trhough process of backward elimination.
a) the most parsimonious logistic regression were derived for each stage of the analysis using SPSS. For example, for Group 2 the Age variable was fixed, it could not be

		HIERARCHICAL LOGISTIC REGRESSION OF HIV ON RISK	SSION OF HIV ON RISK FACTORS		
	HODEL A AGE N=262	HODEL B Socioecon Status N=262	HODEL C MARGINALISATION N-262	HODEL D RISK BEHAVIOUR N=262	HODEL E CO-INFECTION N=262
	OR (I.C. 95 X)	OR (I.C. 95\$)	OR (I.C. 95%)	OR (I.C. 95%)	OR (1.C.95%)
AGE Age group					
25-29 30-34 35+	1.7 (0.9-3.3) 2.7 (1.3-5.7)** 2.2 (1.0-4.9)	2,1 (1,0-4,2)* 3,7 (1,7-8,1)** 3,4 (1,4-8,3)**	1.8 (0.9-3.7) 3.1 (1.4-6.9)** 2.9 (1.2-7.2)*	1.4 (0.6-3.5) 2.0 (0.7-5.2) 3.0 (1.0-8.8)*	1.5 (0.6-3.8) 2.1 (0.8-5.8) 3.4 (1.1-10.4)*
SOCIDECONOMIC STATUS Age leaving education					
13 or less 14 - 15			2.8 (1.2-6.4)* 2.2 (1.0-4.9)	1.6 (0.6-4.4) 1.8 (0.7-4.8)	1.7 (0.6-4.8) 2.0 80.7-4.8)
16 - 18 19 + Abode		1,8 (0,8-3,8)**	1.5 (0,7-3,3)	1.4 (0.5-3.7)	1.4 (0.5-3.9)
No fixed adress Fixed adress		2.4 (1.2-5.0)*	2.3 (1,1-4,7)*	3.4 (1.4-8.1)**	2.9 (1.2-1.7)
MARGINALISATION Previous imprissonment					
Yes No			2.1 (1.1-3.9)*	2.3 (1.1-4.9)*	2.0 (0.9-4.4)
RISK BEHAVIOUR					
Yes				2.6 (1.2-5.7)*	2.6 (1.2-5.8)*
No Inject cocaine					
Yes				0.6 (0.3-1.1)	0.5 (0.3-1.0)
Regular sex partnerHIV+					, , , , ,
No				13.1 (6.4-26.9)***	12.2 (5.9-25.3)***
CO-INFECTION					
yes					28(12-65)*
No					
-211	337.46	319,646	314,005	231.541	225, 209
Improvement Chi-Square	7,896*	17.814**	7.896* 17.814** 5.641*	82.464***	9.369**

Chapter Six

**

HIV, Drugs and Prostitution: An Examination of the Determinants of Prostitution amongst Female IDUs

1. Introduction

The close links between IDU and prostitution have been known for some time. For example, Goldstein (1979) suggested that between 30 to 70 percent of female IDUs were also prostitutes, and that 40 to 85 percent of prostitutes were also drug users. More recently, studies have reported broadly similar findings (see Gossop et al., 1994). However, differences in rates have been reported by other studies, and these probably reflect real differences in sociocultural context as well as differences in sampling. Nevertheless, as noted by Gossop et al. (1994), there is consistent evidence of an association between drug use and prostitution. Thus, with the emergence of the AIDS pandemic, the nature of this relationship has assumed increased importance.

At the beginning of the AIDS pandemic, it was thought that female sex workers, or prostitutes, might constitute a high-risk group for the transmission of the human immunodeficiency virus (HIV). Based on prevalence studies that showed high rates of infection in African prostitutes, it was feared that such women might also be spreading HIV heterosexually in Europe and the United States. This belief has resulted in even greater stigmatisation of sex workers, and in some places has led to restrictive legislation, such as mandatory HIV testing and the quarantine of infected persons (Committee on AIDS Research and the Behavioural, Social and Statistical Sciences, 1990).

Prostitutes, however, are a heterogeneous group in regard to their working environment,

socioeconomic situation, health status, and their knowledge and practice of protective measures. Seroprevalence rates in this group vary a great deal from country to country, within the same country, and from one type of prostitute to another. More than sexual promiscuity itself, HIV infection has been seen to be associated with other risk factors, such as the use of intravenous drugs, having a steady partner who is an intravenous drug user, and the presence of sexually transmitted diseases (Hankins et. al., 1994).

The justification for taking an in-depth look at prostitution arises from its importance in this community. It will be recalled from Chapter 4 that 46.1 percent of female IDUs reported ever having a client. It is therefore of interest to examine the characteristics of female IDU prostitutes and to see how, if at all, policy should differ for this subgroup of women. The objectives of this chapter are to:

1) Look at the relationships between IDU, HIV risks, and prostitution;

2) Examine the determinants of prostitution amongst our study population of female IDUs; and

3) Propose some directions for future research as well as interventions that could be undertaken to reduce risk behaviours and the practice of prostitution.

2. Methods

The methods used for this study, including the bivariate and multivariate analyses, are described in Chapter 3. The only variation from those methods occurs in the logistic regression model in which the following four groups of explanatory variables were identified:

Model 1: Age

Biological age is the first explanatory variable entered into the model because of its anticipated relationship with prostitution among IDUs, as well as its influence on other independent variables.

Model 2: Socioeconomic status

Three variables were selected to define the general socioeconomic status of female IDUs: age left full-time education (coded into four categorical age groups: 13 or less; 14-15; 16-18; and 19+); place of residence in past six months (no fixed abode versus abode); and living with children (coded into three categories: has no children; has children and lives with them; and has children, but doesn't live with them). These variables were selected based on evidence in the literature that they are important socioeconomic markers for this population.

Model 3: Marginalisation

In addition to socioeconomic status, five variables were chosen to characterise the extreme marginalised nature of the sample population. These variables were: one categorical variable - age at first pregnancy (coded into two groups: younger than 18 and 18 or more), and four dichotomous variables (yes versus no): previous imprisonment; having a current regular partner; having a regular sexual partner who was an IDU; and having a casual sexual partner who was an IDU. All of these variables have been documented in the literature as having a relationship with prostitution. Early sexual experience, for example, has been described as a determinant of prostitution (James and Meyerding, 1971) and a marker for risky sexual behaviour in women (Greenberg et al., 1992). Prison experience has been noted as an important determinant of HIV risk amongst prostitutes in Spain (Estébanez et al., 1992). Finally, it has been found that prostitutes engage in more risky sexual behaviour (e.g. less frequent condom use) with non-paying partners (Ward et al., 1993) and that having an IDU partner is a predictor of HIV amongst non-IDU prostitutes (Estébanez et al., 1992).

Model 4: Risk behaviours and HIV seroprevalence

Specific known risk behaviours and HIV seroprevalence were entered into this model to assess whether the relationships with prostitution are direct or mediated through the socioeconomic and marginalisation variables.

The risk behaviours included: ever having regular HIV-positive partner; cocaine injecting; heroin+cocaine injecting; age at first injection; ever sharing needles; age at first injection and using alcohol. HIV seroprevalence was coded into three groups: positive, negative, and unknown.

3. Results of Descriptive Analysis

The principal data from the descriptive analysis are presented in Table 6.1. Amongst the IDUs of our study, 46.1 percent (140 women) reported having ever practiced prostitution, of whom 39.3 percent (55) were considered ex-prostitutes, having not had a client in the six months prior to the interview; another 15.0 percent were characterised as having a low level of promiscuity with fewer than 50 clients per month, while 42.9 percent were high promiscuity, having 50 or more clients per month.

When the date of initiating prostitution was compared to the date of initiating drug use, we found that drug use was initiated first in 79.3 percent of the cases, prostitution came before drug use in 12.1 percent, and 6.4 percent reported initiating both simultaneously. No data was available for the final 2.1 percent.

Almost two thirds of our sample request that their clients use condoms, while 15.0 percent reported not making this request, and 22.1 percent preferred not to respond to this question.

4. Results of Bivariate Analysis

The bivariate relationships with prostitution and selected independent variables are examined in Table 6.2. Following are the principal findings from this table.

4.1. Sociodemographic characteristics and prostitution

No association was found between prostitution and the age of the women. Leaving school at a later age appears to have a certain protective relationship with prostitution, with more than half of the female IDUs who left school before age 16 practicing prostitution, compared to 37.9 percent of those who left school after age 16.

4.2. Marginalisation variables

Not having a fixed address or living in a rented room paid for on a daily basis was found to have the strongest relationship with prostitution, carrying over five times the risk. Previous imprisonment was also significantly related, nearly doubling the risk of practicing prostitution. However, caution is warranted in the interpretation of these findings given that directionality of the relationship cannot be determined in cross-sectional designs. Therefore we can't really say that not having a fixed abode or previous imprisonment are determinants of prostitution. In fact in both cases, the relationship could even be reciprocal. For example, the circumstances surrounding the world of prostitution frequently result in prison and vice versa. Likewise, the circumstances surrounding not having a fixed abode may lead one into prostitution, while it can also be argued that the practice of prostitution implies this type of housing. Meanwhile, in other cases, both circumstances may have a common determinant. The only real conclusion we can establish is that prostitution is more common when a stable domicile doesn't exist and when one has been to prison. We found an association between prostitution and living with one's children. Those who have children but don't live with them have the highest rate of prostitution (71.3 percent), compared to those who live with their children (31.1 percent), and those who reported having no children (40 percent). This result seems logical, due to the incompatibility between the lifestyle and schedules assumed by a prostitute and the responsibility of caring for a child.

4.3. Drug-using habits

Amongst injected drugs, we only found an association between prostitution and the injection of heroin+cocaine, with 51.7 percent of the IDUs who injected this combination reporting to have practiced prostitution, compared to 40.5 percent who had not injected this combination. The use of barbiturates is more than twice as likely among those who practice prostitution. In all probability, the odd hours implicit in prostitution may cause these women to begin using barbiturates to induce sleep which later becomes an abuse problem, reinforcing their drug addiction.

We did not find an association between age of beginning to inject drugs and prostitution. But there were some positive associations between other risk behaviours and prostitution. Those who had ever shared injection material and those who had shared injection material in the last six months were two times more likely to practice prostitution. Furthermore, those who had injected while in prison were nearly three times more likely to practice prostitution.

4.4. Sexual habits

As far as having a relationship with a stable partner, even if the partner is an IDU or HIVpositive, there were no significant differences between those who practiced prostitution and those who didn't.

Those who reported having ever had a casual partner were more than four times more likely to practice prostitution than those who didn't. Similarly, it is more common for those who have had a casual HIV-positive partner (OR=2.5;CI=1.4-4.7) or IDU partner (OR=3.2;CI=2.0-5.1) to work as prostitutes.

4.5 Reproductive habits and infections

An elevated percentage (59.1 percent) of women who currently use some type of contraception work as prostitutes compared to 32.1 percent of those who don't currently use contraception. Of all the contraceptive methods, the condom is used most. and its use in the last years maintains a strong positive association with prostitution (OR=8.3;CI=3.6-19.1). The use of spermicides also has a strong positive association with prostitution (OR=5.1;CI=2.4-10.8).

When the first pregnancy occurred before the age of 18, we found a higher percentage of women who had practiced prostitution (66.2 percent), than when the first pregnancy took place at age 18 or later (42.7 percent).

Those who reported having an STD in the past five years were more than twice as likely to have practiced prostitution.

The association between HIV positivity and prostitution, however, was weak (OR=1.6;CI=1.0-2.6). The same occurred in the case of hepatitis B (OR=1.8;CI=1.0-3.0).

,

٠

Table 6.1 Descriptive analysis of prostitution

Variable	Total	%
Prostitution		
Yes	140	46.
No	164	53.
Number of clients/month last 6 months		
0 (ex-prostitute)	55	39.
< 50	21	15.
50 +	60	42.
No response	4	2.
What came first?		
Drugs	111	79 .
Prostitution	17	12.
Both	9	6.
No response	3	2.
To ask clients about using condom		
Yes	88	62.
No	21	15.
No response	31	22.

.

Variable	Total	Prostitute	OF	R(CI.95%) p	
		Yes(%)	No(%)		
Sociodemographics					
Age		,			
24 or less	73	43.8	56.2	1.2(0.6-2.8)	n.s. ^a
25 -29	110	50.9	49.1	1.6(0.8-3.4)	
30-34	72	45.8	54.2	1.3(0.6-3.0)	
35 or more	49	38.8	61.2	l(reference)	
Age leaving education					
13 or less	88	51.1	48.9	1.7(1.0-3.1)	*a
14-15	82	53.7	46.3	1.9(1.1-3.5)	*a
16 or more	132	37.9	62.1	l(reference)	
Marginalisation variables					
Previously imprisoned					
Yes	98	56.1	43.9	1.8(1.1-3.0)	*
No	206	41.3	58.7		
Abode					
No fixed address	96	72.9	27.1	5.3(3.0-9.4)	***
Fixed address	208	33.7	66.3		
Lives with children					
She has no children	150	40.0	60.0	l(reference)	*** ^a
She lives with	74	31.1	68.9	0.7(0.4-1.3)	
She doesn't live with	80	71.3	28.8	3.7(2.0-7.0)	

Table 6.2 Proportion of women reporting having clients by selected variables

Variable	Total	Prostitute	OI	<u>R(CI.95%) p</u>	
		Yes(%)	No(%)		
Drug-using habits					
Age began injecting drugs					
16 or less	81	48.1	51.9	1.2(0.7-2.2)	n.s.
17 - 19	81	49.4	50.6	1.3(0.7-2.3)	
20 or more	141	43.3	56.7	l(reference)	
Injecting heroin + cocaine					
Yes	151	51.7	48.3	1.6(1.0-2.5)	*
No	153	40.5	59.5		
Use barbiturates					
Yes	71	62.0	38.0	2.3(1.3-4.0)	**
No	233	41.2	58.8	•	
Use hashish					
Yes	126	38.9	61.1	0.6(0.4-1.0)	*
No	178	51.1	48.9		
Ever shared needles/syringes					
Yes	217	51.6	48.4	2.2(1.3-3.7)	**
No	86	32.6	67.4		
Ever shared needles/syringes in	last 6 mo				
Yes	104	55.8	44.2	1.8(1.1-2.9)	*
No	200	41.0	59.0		
Injected in Prison					
Yes	32	71.9	28 .1	2.7(1.1-6.7)	*
No	66	48.5	51.1		

,

Table 6.2 (continued) Proportion of women reporting having clients by selected variables

Table 6.2 (continued) Proportion of women having clients by selected variables

Variable		Total	Prostitute		OR(CI.95%) p	
			Yes(%)	No(%))	
Sexual habits						
Ever had a regular sexua	al partner	r				
Yes	-	293	46.4	53.6	1.3(0.6-2.8)	n.
No		11	36.4	63.6		
Ever had a regular sexua	al HIV+ 1	partner				
Yes		138	49.3	50.7	1.3(0.8-2.1)	n.
No		152	42.8	57.2		
Ever had a regular sexua	al IDU na	artner				
Yes	p	257	45.9	54.1	1.0(0.5-1.8)	n.
No		47	46.8	53.2	()	
Frequency of condom us Vaginal intercourse Never/sometimes Always	122	36.9 35	63.1 48.6	51.4	0.6(0.3-1.3) n.s.	
Ever had a casual sexual	partner					
Ever had a casual sexual Yes	l partner	241	52.7	47.3	4.6(2.3-9.1)	**
	l partner	241 62	52.7 19.4	47.3 80.6	4.6(2.3-9.1)	**
Yes No		62			4.6(2.3-9.1)	**
Yes No		62			4.6(2.3-9.1) 2.5(1.4-4.7)	**
Yes No Ever had a casual sexual		62 artner	19.4	80.6	· · · ·	** **
Yes No Ever had a casual sexual Yes No	∣HIV+ p	62 artner 53 186	19.4 56.6	80.6 43.4	· · · ·	**
Yes No Ever had a casual sexual Yes	∣HIV+ p	62 artner 53 186	19.4 56.6	80.6 43.4	· · · ·	
Yes No Ever had a casual sexual Yes No Ever had a casual sexual	∣HIV+ p	62 artner 53 186 rtner	19.4 56.6 33.9	80.6 43.4 66.1	2.5(1.4-4.7)	
Yes No Ever had a casual sexual Yes No Ever had a casual sexual Yes No Frequency of condom us	l HIV+ p	62 artner 53 186 rtner 156 141	19.4 56.6 33.9 59.0 31.2	80.643.466.141.0	2.5(1.4-4.7)	
Yes No Ever had a casual sexual Yes No Ever had a casual sexual Yes	l HIV+ p	62 artner 53 186 rtner 156 141	19.4 56.6 33.9 59.0 31.2	80.643.466.141.0	2.5(1.4-4.7)	** ** n.s

Variable	Total	Prostitute	OR	R(CI.95%) p	
, 		Yes(%)	No(%)		
Reproductive habits and infect	tions				
Regular periods		•			
Yes	99	35.4	64.6	0.5(0.3-0.8)	**
No	205	51.2	48.8		
Age at first pregnancy					
< 18	74	66.2	33.8	2.6(1.5-4.7)	***
18 or more	150	42.7	57.3	``````````````````````````````````````	
Contraception currently					
Yes	159	59.1	40.9	3.1(1.9-4.9)	***
No	140	32.1	67.9	Х. У.	
IUD used in last 5 years					
Yes	21	33.3	66.7	0.6(0.2-1.4)	n.s.
No	283	47.0	53.0		
Condom used in last 5 years					
Yes	247	53.8	46.2	8.3(3.6-19.1)	***
No	57	12.3	87.7		
Spermicide used in last 5 years					
Yes	45	77.8	22.2	5.1(2.4-10.8)	***
No	259	40.5	59.5		
Contraceptive pill used in last 5	years				
Yes	91	42.9	57.1	0.8(0.5-1.4)	n.s .
No	213	47.4	52.6		
Ever been pregnant					
Yes	230	50.0	50.0	1.9(1.0-3.5)	*
No	74	33.8	66.2		
Abnormal smear test					
Yes	49	57.1	42.9	1.9(1.0-3.5)	*
No	224	41.5	58.5		
STD in last 5 years					
Yes	77	61.0	39.0	2.7(1.5-4.8)	***
No	196	36.7	63.3		

Table 6.2 (continued) Proportion of women reporting having clients by selected variables

•

-HIV +					
Yes	165	53.3	46.7	1.6(1.0-2.7)	*
No	97	41.2	58.8	· · · ·	
-Hepatitis A in last 5 years					
Yes	8	62.5	37.5	2.1(0.5-8.8)	n.s.

•

.

,

Variable	Total	Prostitute	OR(C	CI.95%) p	
••••••••••••••••••••••••••••••••••••••		Yes(%)	No(%)		
Reproductive habits and infe	ctions (cor	tinued)			
HIV+		•			
Yes	165	53.3	46.7	1.6(1.0-2.7)	*
No	97	41.2	58.8		
Hepatitis A in last 5 years					
Yes	8	62.5	37.5	2.1(0.5-8.8)	n.s.
No	276	44.6	55.4		
Hepatitis B in last 5 years					
Yes	72	55.6	44.4	1.8(1.0-3.0)	*
No	212	41.5	58.5		
TT vitis O is last 5 second					
Hepatitis C in last 5 years Yes	89	49.4	50.6	1.3(0.8-2.1)	n.s.
No	195	43.1	56.9	1.5(0.8-2.1)	11.3.
Hepatitis D in last 5 years Yes No	2 282	0.0 45.4	100.0 54.6		n.s.
Tuberculosis in last 5 years	15	53.3	46.7	1.5(0.8-2.8)	n c
Yes	45 239	43.5	56.5	1.5(0.8-2.8)	n.s.
No	237	-J.J	50.5		
Some infection last year					
Yes	148	51.4	48.6	1.7(1.0-2.8)	*
No	136	38.2	61.8		
Type of Treatment					
Methadone maintenance	65	40.0	60.0	1.0(reference)	
Other treatments	143	51.7	48.3	1.6(0.9-3.1)	n.s.
No treatment	96	41.7	58.3	1.1(0.5-2.1)	n.s.

Table 6.2 (continued) Proportion of women reporting having clients by selected variables

.

a: p value of linear trend test

.

5. Results of Hierarchical Logistic Regression Analysis

In the first model, none of the age categories enter significantly into the equation. In the second model, two variables representing the socioeconomic level appear in the equation: place of residence, and living with children. The place of residence maintains a strong association with the dependent variable, with those IDUs who don't have a fixed address presenting a higher probability of having practiced prostitution (OR=4.7;CI=2.7-8.1). This finding hardly varies in subsequent models. To examine the variable "living with children" we compared those "not living with their children" to those "living with children" or "not having practiced prostitution (OR=3.6;CI=2.0-6.4). This probability descends slightly when we incorporate the variables representative of marginalisation in the third model (OR=2.7;CI=1.4-5.1), but suffers few changes in the final model.

Of the six variables selected to study the relationship between the degree of marginalisation and prostitution, only two remain in the equation: 1) Having had a casual IDU partner; and 2) Age at first pregnancy. Having had a casual IDU partner appears to be a relatively strong determinant of prostitution (OR=2.9;CI=1.7-5.0). This finding does not change in subsequent models. Having had a first pregnancy before the age of 17 is also a significant determinant of prostitution (OR=2.0;CI=1.0-4.0) in both the third and fourth models. It is interesting to note that previous imprisonment does not enter into the model, suggesting an indirect relationship with prostitution.

Of the variables selected to represent risk behaviour and infections, only HIV seroprevalence entered the model. Not knowing one's HIV status appears to have a negative relationship with prostitution, and being HIV-positive appears to have no relationship with prostitution.

6. Discussion

The findings of this study are consistent with many previous studies in which the exchange of sex for money or drugs stood out as a common practice among female IDUs (Franceschi, 1988; Lewis, 1991; Saxon, 1991; Van den Hoek, 1988; Astemborski, 1994). In our study, more than 45 percent of the women reported having practiced prostitution at some point in their life. For the great majority of these women, it was the need to pay for their drug habit which drove them to prostitution.

Although in the bivariate analysis we observed a statistically significant association between prostitution and HIV infection, once we control for other variables in the logistic model, this association fails to be significant, indicating an indirect relationship. Furthermore, the negative relationship with unknown HIV status suggests that prostitutes are more likely to have been tested or at least more likely to have returned a physician's report on their HIV status. This finding of no relation between prostitution and seropositivity coincides with the findings of other European studies (Estébanez, 1995; Rhodes, 1994; Van den Hoek, 1988; Tirelli, 1989) in which the transmission of HIV among female prostitutes who also inject drugs is associated more with the use of drugs than with prostitution. Some studies such as Astemborski (1994) find significant

differences in the prevalence of HIV between prostitutes with a high level of promiscuity (50 or more clients per month) and those of low promiscuity (fewer than 50 clients per month), while they don't find a difference in prevalence between the latter group and non-prostitute IDUs. We did not find a significant difference using the promiscuity index.

There are no differences in condom use with regular partners or casual partners. Even though one knows that condoms are a reasonably safe way to prevent HIV infection, they are not particularly popular in private sexual relations. The obstacles to condom use which are typically mentioned are, in general, social factors. 'Prostitutes are reluctant to use condoms in private sexual relations, and not using them constitutes an essential distinction between the two types of relations (private versus commercial). Sexual relations without condoms with a private partner who injects drugs, exposes prostitutes to a greater risk of HIV infection than their relations with clients.

In the final interpretation of the logistic model of our study, the profile of the female prostitute IDU varied from that of a non-prostitute in that the former lacked to a greater degree a fixed address, they were less likely to live with their children, and the median age of the first pregnancy was lower. A higher level of homelessness among prostitute IDUs perhaps reflects a greater level of economic oppression amongst prostitute IDUs and possibly a reason for turning to prostitution as a source of income in the first place. An earlier age of first pregnancy is an interesting finding as earlier studies have linked both early age of first pregnancy and prostitution to childhood sexual abuse (Zierler et al, 1992). Furthermore, early sexual experience has been identified as a factor in prostitution (James and Meyerding, 1977) and age at first coitus as a marker for risky sexual behaviour (Greenberg, Magder and Aral, 1992). The implications of these findings are that early sexual experiences, including childhood sexual abuse, can be used as markers for high-risk sexual behaviour, and interventions aimed at delaying a woman's first sexual experience may lower the likelihood of later engaging in prostitution and other risky sexual practices.

Up until this point, the data are comparable to the possible differences one might find between prostitutes and non-prostitutes in a non-drug using population. But in the case of the IDUs, it stands out that the group of prostitutes is strongly associated with the existence of casual IDU sexual partners. Future investigations should attempt to verify this result in a non-drug-dependent population.

7. Conclusions

- · C This study makes it evident, once again, that the use of drugs and prostitution are directly associated.
 - **C** In the population studied of female IDU prostitutes in Madrid, the prevalence of HIV is slightly higher than non-prostitutes, but this difference disappears when we control for socioeconomic status and marginalisation variables.
 - C Condom use, while relatively frequent with clients, is less common in private sexual relations, which is comparable to what occurs with non-prostitutes.
 - **C** The degree of marginalisation and living conditions of IDUs who work as prostitutes, are even more unfortunate than those of IDUs themselves.
 - **C** This study provides further evidence that the age of first pregnancy or early sexual experience is an important determinant of prostitution.
 - C Those who are homeless, use drugs, and get pregnant at a young age are at higher risk for practicing prostitution.

HIERARCHICAL LOGISTIC REGRESSION OF PROSTITUTION ON RISK FACTORS

-2LL Improvement Chi-Square	RISK BEHAVIOUR and INFECTIONS HIV Seroprevalence HIV negative HIV positive HIV unknown	MARGINALISATION Casual partner IDU No Yes Age first pregnancy 17 or less 18 +	SOCIOECONOMIC STATUS Abode No fixed adress Fixed adress Living with children No Yes or she has not children	AGE GROUP 24 or less 25 - 29 30 - 34 34 or more 0.8	
419.537 -2.243				 1.3 (0.7-2.4) 1.1 (0.6-2.1) 0.8 (0.4-1.7)	HODEL A AGE N=304 OR (I.C. 95%)
356.506 60.787***			4.7 (2.7.8.3)*** 3.6 (2.0-6.6)***	1.5 (0.8-2.9) 1.1 (0.5-2.3) 0.9 (0.4-2.0)	Hodel B Socioeconomic Status N=304 Or (I.C. 95%)
336.824 19.682***		3.1 (1.8-5.2)*** 2.0 (1.0-3.9)*	4 .7 (2.6-8.6)*** 3.0 (1.5-5.8)**	1.5 (0,7-3.0) 0.9 (0.4-2.0) 0.9 (0.4-2.1)	HODEL C MARGINALISATION N=304 OR (1.C. 95%)
331.807 5.017	1.0 (0.6-1,9) 0.4 (0.2-1.0)*	2.6 (1.5-4.6)*** 2.1 (1.1-4.2)*	4.7 (2.6-8.5)*** 2.6 (1.3-5.0)**	1.4 (0.7-2.8) 0.8 (0.4-2.8) 0.8 (0.3-1.9)	HODEL D RISK BEHAVIOUR AND INFECTIONS N=304 OR (1.C. 95\$)

a) The most parsimonious logistic regression were derived for each stage of the analysis using SPSS. For example, for Group 2 the Abode variable was fixed, it could not be eliminated, whilst the new variables were selected rhough process of backward elimination.
b) * p<0.05; ** p<0.01; *** p<0.001.
c) For models B, and C the following variables were excluded from the equation. Stage 1: Age left education: Stage 2: Regular partner IDU. Actual regular partner. Previous imprissonment; Stage 3: Use alcohol. Regular partner HIV, Cocaine injecting/eroine+cocaine injecting. Age first injection. Ever sharing needles and STD last year.

Chapter Seven

Determinants of Previous Imprisonment amongst Female IDUs

1. Introduction

There is a considerable body of evidence to suggest that prisoners have a history of drug use before entering prison. For example, a survey of pre-arrest drug use amongst 1,741 men serving prison sentences in England and Wales, found that 11.0 percent of inmates had injected drugs (Maden, Swinton and Gunn, 1992). Likewise, a series of surveys in Scottish prison have reported previous or current IDU to range between 16.8 (Bird, Gore, Burns and Duggies, 1992a,b) and 33.5 percent (Taylor et al. 1995). Finally, in the USA in 1989, inmates were twice as likely as the general population to have ever used drugs: 78.0 percent of inmates had used drugs during their lifetime; 44.0 percent had used drugs in the month before their current offense; 20.0 percent daily or almost daily; and 27.0 percent were under the influence when they committed their current offense (Polonsky, 1994). Although less evidence is available, data from all female prison populations indicate that female prisoners also have a history of IDU. In a study of 145 female prisoners conducted at HMP (Her Majesty's Prison) Cornton Vale, Scotland, 46 percent of female inmates had injected drugs (Gore et al., 1997). In a larger scale study of the same prison population, 21.3 percent of 3,389 consecutive admissions admitted to drug misuse in the six months prior to custody (Stuart, O'Rourke & Power, 1997). In the USA in 1989, one in three female inmates had been imprisoned for drug-related offenses as compared to one in five male inmates (Polonsky et al., 1994).

Towards the beginning of the acquired immunodeficiency syndrome (AIDS) pandemic, it was realised that the prevalence of the human immunodeficiency virus (HIV) was disproportionately high in prison populations when compared to the population *per se*. Indeed, this is verified in the current study where previous imprisonment is demonstrated to be an important factor in the multivariate logistic regression models presented in Table 5.3 (Chapter 5).

In 1983, a study of an American prison population in New York State demonstrated the importance of intravenous drug use (IDU) in the transmission of HIV (Wormer et al., 1983; Harding and Schaller, 1992). At the time the population represented a convenient sample of drug users and little attention was given to the actual transmission of HIV in prisons (Harding and Schaller, 1992). Since then, however, the prime objective of a number of studies has been to examine the prevalence of HIV in prison populations and to identify relevant risk factors. Many of these studies conclude that the high prevalence of HIV arises because prison populations are predominantly drawn from young, sexually active groups who disproportionately engage in high-risk activities, including, for example, intravenous drug use. Moreover, many people who have a high risk of HIV transmission are also at risk of imprisonment.

If it is assumed that the high incidence of HIV among women were recorded at the outset of a prison sentence, then one can infer that HIV-infected women are more likely to be imprisoned than HIV-infected men. Studies of prisoners suggest that jails are places where a disproportionate number of individuals have a history of risk factors associated with HIV infection (Gaughwin et al., 1990). These include, for example, the injection of illicit drugs (Dolan, Donoghoe and Stimson, 1990), unprotected sexual intercourse (Magura, Shapiro and Kang 1994) and under-provision of

health care (Kendig et al., 1994). Although few studies have been published on risk factors of HIV for women entering prisons, evidence can also be accumulated from data based on male subjects and from studies of women in other high-risk populations.

As partly illustrated in Chapter 6, a further compounding factor in examining the characteristics of female prisoners, HIV prevalence and IDU is the proportion of prisoners who have exchanged sex for drugs or money. In the HMP Corton Vale study by Stuart and colleagues (1997), 16.5 percent of the 616 female drug abusers were in prison on charges of prostitution. Gore et al. (1997) reported that 20 percent of the female injector-inmates in their study had worked as prostitutes. In a multicentre study of female sex workers in Spain, 13.9 percent reported having been previously incarcerated (Estébanez et al., 1995).

In addition to the unrepresentative proportion of IDUs and female sex workers seemingly characteristic of prison entrants, these women are also susceptible to a gamut of diseases associated with both poverty and IDU/prostitution itself. For example, the presence of STDs has been shown to be another major factor that has been associated with sexual transmission of HIV. As previously mentioned, diseases that cause genital ulcers may facilitate both transmission of and susceptibility to HIV.

Other bloodbourne viruses, including hepatitis B and C, have been identified in both male and female prison entrants. In an Australian study the incidence for hepatitis B was 12.6 percent and, for hepatitis C, was 18.3 percent (Crofts et al., 1995). In an examination of 265 male prison inmates in Maryland, USA, intake prevalence for hepatitis C was 38.0 percent and seroincidence was 1.1 percent per person year in prison (Vlahov et al., 1993).

Finally, prison populations have long been considered to be at high risk of tuberculosis (TB) (Darbyshire, 1990). Most studies have either examined the incidence of TB within prison (Darbyshire, 1990) or have examined the prevalence of TB using a cross-section of prisoners (Martin et al., 1994). In the latter study, 56.2 percent of prisoners were infected with Mycobacterium tuberculosis, although the data do not permit the calculation of the prevalence for new entrants. Studies are therefore needed to examine the prevalence of TB, and other bacterial diseases, amongst prison entrants.

To examine the potential for the transmission of HIV and other diseases amongst prisoners, information is needed on risk activities whilst incarcerated. Evidence would then enable the development of effective and equitable policies for the care and treatment of HIV/AIDS prisoners. For example, in the current study nearly a third of those who had been in prison (n=98) reported injecting whilst in prison.

It is the aim of this chapter to look at the determinants of previous imprisonment amongst female IDUs. As with prostitution in Chapter 6, the main justification for so doing is to try to create a better understanding of the causes, and consequences, of a very marginalised environment/lifestyle with respect to HIV. Thus by examining prostitution and previous imprisonment (which were both associated with HIV prevalence in Chapter 5), it might be possible to build a more comprehensive picture. However, it should be noted that - as in Chapter 6 - in the proceeding bivariate and multivariate analyses the characteristics are examined; not risk factors.

Therefore the objectives of this chapter are:

1) To discern the profile of female IDUs that have been in prison, as well as the relationships with other marginalisation factors such as prostitution and HIV;

- 2) To examine the determinants of imprisonment in female IDUs using the data from this study; and
- 3) To facilitate the design of new strategies intended to prevent the transmission of HIV within prisons.

2. Methods

The methods used for this study, including the bivariate and multivariate analyses, are described in Chapter 3. The only variation from those methods occurs in the logistic regression model in which the following four groups of explanatory variables have been identified:

Model 1: Age

In order to evaluate the effect of age on the dependent variable (previous imprisonment) the biological age (coded into four categorical age groups: 24 or less; 25-29; 30-34; and 35+) was the sole variable used in this group.

Model 2: Socioeconomic status and marginalisation

Based on the literature and results of the bivariate analysis, three variables were selected to define the general socioeconomic status of female IDUs: age left full-time education (coded into four categorical age groups: 13 or less; 14-15; 16-18; and 19+); source of income during past six months (illegal and prostitution versus legal); and, place of residence in past six months (no fixed abode versus abode).

In addition to socioeconomic status, two variables were chosen to characterise the extreme marginalised nature of the sample population. These dichotomous variables (yes versus no) were: ever had clients and having a regular partner who was an IDU.

Model 3: Risk behaviours

Specific known risk behaviours of previous imprisonment were included in the third model: ever have regular HIV+ partner; cocaine injecting; heroin+cocaine injecting; age began injecting; ever share needles; and number of clients per month.

Model 4: Gynaecological background and infections.

The variables introduced in this model were: previous pregnancy; some STD in the past year; and hepatitis B during the past year.

3. Results

First we will examine the bivariate relationships with previous imprisonment and selected variables presented in Table 7.1. Then we will present findings from the multivariate logistic regression.

3.1. Sociodemographic characteristics and previous imprisonment

There is a clear positive association between age and prison stay: the older the age the higher the probability of having been in prison. Compared to the youngest stratum (24 years or less), those in the three older strata (25-29; 30-34; over 34) were two to three times more likely to have been in prison.

3.2. Marginalisation variables

The extreme marginalisation of the IDU population which has been to prison becomes evident when we examine the place of residence and source of income. Forty-nine percent of female IDUs that don't have a fixed address had been in prison, compared to 27.2 percent of those that do have one (OR=2.6;CI=1.5-4.5). Furthermore, 42.3 percent of the women whose source of income was illegal or prostitution had been in prison, compared to only 24.0 percent of those who had a legal source of income (OR=2.3;CI=1.4-3.9).

The age of leaving school did not have a significant p-value in the Linear trend test; however, the stratum of those who left school at age 13 or younger had a higher rate of having been in jail (43.2 percent) compared to those who stayed in school until age 16 or older (35.6 percent).

3.3. Drug-using habits

The age of beginning to inject drugs maintains a strong relationship with previous incarceration. Those that began injecting younger (16 years or less) are more than 3 times more likely to have been in prison compared to those who began at age 20 or later. Those who began between ages 17 and 19 are nearly twice as likely to have been to prison.

Those who have ever exchanged injection material are twice as likely to have been incarcerated. However there is not a significant association between having shared injection material in the last 6 months and having been in prison.

No significant associations were found between the use of any specific drugs and imprisonment.

3.4. Sexual habits

No significant associations were found between having a steady sexual partner and having been in prison, even when the partner was an IDU. It is useful to remember that almost all of these women (94 percent) reported having had a steady partner, and in the great majority of cases (85 percent), this partner was also an IDU. However, a positive association does appear between those women who had a steady sexual HIV-positive partner and imprisonment, with those who had an HIV-positive partner presenting an OR=2.1(CI=1.2-3.4) compared to those who hadn't. No relationship was found between condom use with regular partners and prison.

Although no association was found between having had a casual partner and prison, we did find an association when this partner was HIV-positive or an IDU. Having an HIV-positive casual partner produced an OR of 2.7(CI=1.4-5.1), and when the casual partner was an IDU the OR is 2.1(CI=1.2-3.4). No association was found with condom use with casual partners.

Prostitution is also associated with previous imprisonment: the women with a history of prostitution have nearly double the odds compared to those who never practiced prostitution. No association appears between prison and the use of condoms with clients.

3.5 Reproductive habits and infections

We found a significant association between having a regular menstrual period and the dependent variable. When menstruation is regular, 21.2 percent have been in prison whereas the proportion increases to 37.6 percent when it is not regular (OR=0.4;CI=0.3-0.8).

No association was found between imprisonment and the various types of contraception used in the past five years. Nor was their a finding with current use of contraception.

We found a strong relationship between having ever been pregnant and the dependent variable. Those reporting a previous pregnancy were three times more likely to have been imprisoned compared to those who had never been pregnant.

Having had any type of STD in the five years prior to the interview also nearly tripled the odds of previous imprisonment.

As expected and in agreement with data presented in Chapter 5, there is a positive association between HIV and prison. The risk of having been in prison for those who are HIV-positive is 2.8 times greater than for those who are HIV- negative or unaware of their serostatus (CI=1.5-4.9).

Variable	Total	Imprisonment		OR(CI95%)	р
	-	Yes(%)	No(%)		
Sociodemographics					
Age		•			
24 or less	73	17.8	82.2	l(reference)	**3
25 -29	110	33.6	66.4	2.3(1.1-5.1)	
30-34	72	41.7	58.3	3.3(1.5-7.6)	
35 or more	49	36.7	63.3	2.7(1.1-6.7)	
Age left education					
13 or less	88	43.2	56.8	1.4(0.7-2.7)	n.s.ª
14-15	82	28.0	72.0	0.7(0.3-1.5)	
16 or more	73	35.6	64.4	l(reference)	
Abode					
No fixed address	69	49.3	50.7	2.6(1.5-4.5)	***
Fixed address	235	27.2	72.8		
Income					
Illegal+prostitution	137	42.3	57.7	2.3(1.4-3.9)	***
Legal	167	24.0	76.0		
Drug-using habits					
Age began injecting drugs					
16 or less	81	48.1	51.9	3.3(1.8-6.2)	***8
17 - 19	81	33.3	66.7	1.8(0.9-3.4)	
20 or more	141	22.0	78.0	1(reference)	
Ever shared needles/syringes	I.				
Yes	217	36.9	63.1	2.2(1.2-4.0)	**
No	86	20.9	79 .1		
Ever shared needles/syringes	in last 6 r	nonths			
Yes	104	40.4	59.6	1.3(0.8-2.3)	n .s.
No	114	33.3	66.7		

Table 7.1 Proportion of women previously imprisoned by selected variables

Variable	Total	Imprisonment		OR(CI95%)	р
, anabic	Tour	Yes(%)	No(%)	011(02)0747	<u>P</u>
Sexual habits					
Ever had a regular sexual part	ner				
Yes	293	• 32.1	67.9	0.8(0.2-2.9)	n.s.
No	11	36.4	63.6		
	. .				
Ever had a regular sexual HIV		39.1	60.9	2.1(1.2-3.4)	**
Yes	138 152	23.7	76.3	2.1(1.2-3.4)	
No	152	23.1	70.5		
Ever had a regular sexual IDU	partner				
Yes	257	32.7	67.3	1.1(0.6-2.6)	n.s.
No	47	29.8	70.2		
Frequency of condom use with	n regular :	sexual partner			
-Vaginal intercourse	100	25.4	74.6	1.0(0.4-2.3)	n c
Never/sometimes	122 35	25.4 25.7	74.0 74.3	1.0(0.4-2.5)	n.s.
Always	22	23.7	74.5		
Ever had a casual sexual partn	er				
Yes	241	32.8	67.2	1.1(0.6-2.0)	n.s.
No	62	30.6	69.4		
Ever had a casual sexual HIV-		15 7	517	27(1451)	**
Yes	53	45.3 23.7	54.7 76.3	2.7(1.4-5.1)	
No	186	23.1	70.5		
Ever had a casual sexual IDU	nartner				
Yes	156	39.7	60.3	2.1(1.3-3.4)	**
No	141	24.1	75.9		
Frequency of condom use with	n casual s	exual partner			
-Vaginal intercourse	10	29.2	70.8	0.7(0.3-1.6)	n.s.
Never/sometimes	48 50	38.0	62.0	0.7(0.0-1.0)	11.3.
Always	50	56.0	02.0		
Ever had a client					
Yes	140	39.3	60.7	1.8(1.1-3.0)	*
No	164	26.2	73.8		
_		•			
Frequency of condom use with -Vaginal intercourse	h casual c	lient			
Never/sometimes	26	38.5	61.5	0.8(0.3-2.2)	n.s.
Always	20 52	42.3	57.7	、 <i>、 、</i>	

Table 7.1 (continued) Proportion of women previously imprisoned by selected variables

Variable	Total	Imprisonment		OR(CI95%)	р
		Yes(%)	No(%)		
·	c				
Reproductive habits and in	tections				
Regular periods					
Yes	99	· 21.2	78.8	0.4(0.3-0.8)	**
No	205	37.6	62.4		
Age at first pregnancy					
< 18	74	44.6	55.4	1.6(0.9-2.8)	n.s.
18 or more	150	34.0	66.0		
Contracention everently					
Contraception currently Yes	159	30.2	69.8	0.8(0.5-1.3)	n.s.
No	140	34.3	65.7	0.0(0.5~1.5)	11.J.
IUD used in last 5 years		14.2	05 7	0.2(0.1.1.1)	
Yes	21	14.3	85.7 66.4	0.3(0.1-1.1)	n.s.
No	283	33.6	00.4		
Condom used in last 5 years					
Yes	247	33.2	66.8	1.3(0.7-2.4)	n.s.
No	57	28.1	71.9		
Spermicide used in last 5 yea	urs				
Yes	45	42.2	57.8	1.7(0.9-3.2)	n.s.
No	259	30.5	69.5		
Contraceptive pill used in las	t 5 veore				
Yes	91	28.6	71.4	0.8(0.5-1.3)	n .s.
No	213	33.8	66.2	0.0(0.0 1.0)	
	215	55.0			
Ever been pregnant					
Yes	230	37.4	62.6	3.1(1.6-6.1)	***
No	74	16.2	83.8		
Regular smear test					
Yes	134	30.6	69.4	0.9(0.5-1.4)	n.s.
No	168	33.9	66.1		
STD in last 5 years					
Yes	77	49.4	50.6	2.9(1.6-5.1)	***
No	196	25.5	74.5	. ,	

Table 7.1 (continued) Proportion of previously imprisoned women by selected variables

Variable	Total	Imprisonment		OR(CI95%)	n
<u>,</u> ,	TUAL	Yes(%)	No(%)		₽
Donnoductivo hobito oudin	factions (aantinued)			
Reproductive habits and in	itections (continueu)			
HIV +		,			
Yes	165	41.8	58.2	2.8(1.5-4.9)	***
No	97	20.6	79.4		
Hepatitis B in last 5 years					
Yes	72	40.3	59.7	1.7(1.0-3.0)	n.s.
No	212	28.3	71.7		
Hepatitis C in last 5 years					
Yes	89	31.5	68.5	1.0(0.6-1.7)	n.s.
No	195	31.3	68.7		
Tuberculosis in last 5 years					
Yes	45	40.0	60.0	1.6(0.8-3.0)	n.s.
No	239	29.7	70.3		
Herpes last year					
Yes	23	47.8	52.2	2.1(0.9-4.5)	n.s.
No	250	30.8	69.2		
Candidiasis last year					
Yes	43	46.5	53.5	2.1(1.1-4.0)	n.s.
No	230	29.6	70.4		
Genital warts last year					**
Yes	28	57.1	42.9	3.2(1.4-7.1)	**
No	245	29.4	70.6		
Some infection last year			(a a		÷
Yes	148	36.5	63.5 74.2	0.6(0.4-1.0)	*
No	136	25.7	74.3		
Type of Drug Treatment					
Methadone maintenance	65	35.4	64.6	1.0(reference)	n.s.ª
Other treatment	143	36.4	63.6	1.0(0.5-2.0)	
No treatment	96	24.0	76.0	0.6(0.3-4.5)	
		123			

Table 7.1 (continued) Proportion of women previously imprisoned by selected variables

3.6. Results of hierarchical logistical regression analysis

Age: Age appears to be a factor directly associated with the dependent variable, maintaining a significant relationship, with the youngest group presenting the lowest risk of having been in prison. This relationship is maintained throughout the four models, increasing in its statistical significance in the second and third models, and decreasing slightly in the final model. Interestingly, on examination of the different models, we can observe that in the first model the group of women over 35 years present an OR slightly lower than the group of 30- to 34-year-olds (2.7 and 3.3 respectively), thus in the older strata the risk of imprisonment had descended. But if we look closely, this phenomenon disappears in the third model, as we adjust for risk factors, gynecological history and infections. Furthermore, in the final model the direct relationship between age and imprisonment remains clearly evident (the older the age, the higher the risk), with the risk in respect to the group under 24 years varying from 2.9 (CI=1.3-6.6) for the group of 25- to 29-year-olds, to 4.5 (CI=1.9-10.9) for the group of 30- to 34-year olds, and 4.9 (CI=1.8-13.4) for the group of 35 years or older.

Socioeconomic status and marginalisation: The source of income is a factor directly associated with prison, although it loses significance slightly in the final model, where having an illegal source of income presents an OR=2.2 (CI=1.2-4.0) compared to having a legal source of income. Not having a fixed address, though strongly associated to imprisonment in the bivariate analysis, entered only into the second model, and with a weak significance (p=0.08). This association continues to weaken with the introduction of subsequent models. Using the oldest age group for leaving full-time studies (19 years and over) as a reference, two groups (13 or less and 16 to 18) maintain significant associations with imprisonment in the second and third models, but as we proceed, the association weakens and ceases to be significant in the final model. Having had a regular IDU sexual partner and having practiced prostitution remained out of the equation.

Risk behaviour: Of the variables selected to represent risk behaviours, only the age of beginning to inject drugs remains in the equation, with the rest remaining out of the equation. This variable loses significance slightly when we adjust for the variables of previous pregnancy and having had some STD in the previous year. When the age of beginning drug use is younger, the risk of having been in prison increases, such that in the final model, the OR for the group of 16 years or younger is 3.0 (CI=1.5-6.0) and for the 17- to 19-year-olds is 1.6 (0.8-3.1) in comparison to the group of 20 years or older.

Gynaecological background and infections: Those who have ever been pregnant are twice as likely than those who have never been pregnant to have been to prison (OR=2.2; CI=1.0-4.8). The women who have had some STD in the last year are also more likely to have been imprisoned compared to those who had not had an STD in the last year (OR=1.9; 1.1-3.6).

4. Discussion

The main objective of the preceding analysis was to review the determinants of previous imprisonment amongst female IDUs. It will be remembered from Chapter 4 (Table 4.1) that 32.2 percent of the study sample reported that they had been imprisoned. As demonstrated in the bivariate analysis (Table 7.1), the more marginalised the female IDU, the more likely she is to have been to prison. This is demonstrated through the association between imprisonment and having an STD in the past year and having a casual partner who is an IDU or HIV-positive, for example (Table 7.1). It should be stressed that having a partner who is HIV-positive is *not* a cause of imprisonment, but a characteristic of very marginalised female IDUs. In the regression analysis the only marginalisation variable to maintain a direct relationship with previous imprisonment is having had an STD.

Socioeconomic status appears to play an important role in determining previous imprisonment. Claiming one's source of income as illegal or prostitution is directly associated with imprisonment. Age of leaving education and having no fixed address appear to be indirectly linked to imprisonment. Socioeconomic status, as determined by living in a poor district, was also identified as having a direct relationship with previous imprisonment amongst IDUs in a study of AIDS patients in Barcelona (Caylá et al., 1995). Previous pregnancy, which most likely represents the economic burden of children to support, is also directly associated with imprisonment. Perhaps these women are pressed into illegal activities in order to support their families.

Age and age of beginning to inject drugs both have strong, direct relationships with imprisonment. The association with age is most likely a reflection of exposure. The older the woman, the longer she's been involved in IDU and other illegal activities, the more likely it is that she has been to prison. Early injectors have been identified in other studies as more likely to engage in drug-using risk behaviours (e.g. frequency of injection, frequency of needle sharing, and use of shooting galleries) and sexual risk behaviours (amongst females, e.g. prostitution, multiple sex partners) (Battjes, Leukefeld and Pickens, 1992). Thus early injectors may represent a more hard-core group of drug users and are an important target for HIV risk-prevention programmes within prisons.

The positive associations between prison and HIV positivity as well as prison and previous TB infection (not significant) are interesting to consider. The association between HIV and TB with previous imprisonment may - but not necessarily - be the effects of imprisonment. For example, up until the beginning of the AIDS pandemic, TB was considered to be an infectious disease that was all but eliminated from the industrialised countries of Europe. However, in the past decade a number of studies have demonstrated a strong association with marginalised populations, including, for example prisons (Darbyshire, 1990). This association may, in part, be confounded by the prevalence of HIV/AIDS in these populations. For example, HIV infection has been shown to be one of the major risk factors for TB. In Barcelona (Spain), in 1993, 24.2 percent of TB cases were HIV positive and a further 15.4 percent were diagnosed with AIDS (Guelar et al., 1993). Furthermore, studies of the incidence of TB among HIV-infected persons in the United States have revealed rates 100 to 1000 times higher than in the general population (Markowitz et al., 1997). In light of these observations, it is not that surprising that the study population of female IDUs reported to have been infected by TB.

Investigations into HIV amongst female prisoners have occurred infrequently. Nevertheless, the reported prevalence of HIV among female prisoners in North America and Europe ranges from 3.3

percent in the USA (Polonsky et al., 1994) to 26.0 percent in Spain (Estébanez et al., 1990). An anonymous HIV surveillance study at the Corton Vale all female-prison in Scotland found no undisclosed cases and reported an overall HIV prevalence of 1.4 percent (Gore et al., 1997). Data from male or mixed prison populations are more frequently available. Examples include the 3.4 percent of 15,052 male prisoners from the American state of Georgia who were HIV positive (Harding, 1990); the testing of 500 consecutive entries to prison in France that found prevalence to be 13 percent (in Harding, 1990); and the screening of 30,392 Italian prisoners of whom 16.8 percent were HIV positive (in Harding, 1990). Harding (1987) estimated the overall prevalence of HIV in European prisons to be in excess of 10 percent and between 0.1 and 15 percent in 1992 (Harding and Schaller, 1992). Unfortunately, however, these studies are not directly comparable due to different methods of data collection, varying sample sizes and the nature of the prison population from which the samples were obtained (McKee, 1992).

Although Spanish prison officials insist that IDU is not a problem in Spanish prisons, it should be remembered from Table 4.2 that 32.7 percent of those who had been imprisoned reported injecting whilst in prison (Table 4.2). In fact, drug use in prison has been widely reported in other countries. For example, from a study of 559 Scottish inmates (480 males and 79 females) 7.7 percent reported IDU on at least one occasion during a period of imprisonment (Power et al., 1992). A number of surveys have been published on HIV prevalence in male Scottish prisoners, all of which have reported IDU whilst incarcerated. For example, in Edinburgh, nearly one in five prisoners reported IDU, of whom one half had injected whilst in prison (Bird et al., 1992b); in a young offenders institute 17 percent of prisoners admitted misuse of intravenous drugs, of whom a quarter reported use while in prison (Bird et al., 1992a); and, following an outbreak of hepatitis B and HIV at Glenochil Prison, an investigation reported that a quarter of IDUs had first injected inside prison (Gore et al., 1995). Similar evidence was recently reported on female Scottish prisoners, with 46 percent of inmates reporting IDU, 57 percent of these injector-inmates reporting to have injected whilst in prison (Gore et al., 1997).

Unlike other IDUs, prisoners are often denied access to public health programmes, such as needle exchanges (Gore, 1993), and therefore are at more risk of sharing HIV-contaminated injecting equipment (Bath et al., 1993). In fact, Keene (1994) reported that the levels of HIV risk through IDU were higher for subjects in prison than for those out of prison on parole. Furthermore, a multifactorial analysis of 612 IDU from Berlin, Germany, revealed that the most important risk factor for HIV infection was needle-sharing in prison (Müller et al., 1995); and Turnbull, Dolan and Stimson (1992) reported that in the United Kingdom several studies have found that up to a quarter of IDUs who go to prison report that they managed to inject drugs whilst in prison and that most people who inject in prison share syringes. Finally, although no known data is available, it is likely that IDUs who first share needles whilst in prison continue to do so after release and thus imprisonment is associated with a long-term change in risk-taking behaviour.

Thus not only are prisoners who inject at particularly high risk of contracting HIV based on their out-of-prison behaviour and living circumstances, but they also engage in high-risk behaviour whilst in prison. Therefore it is critical to target HIV prevention education in prison populations, not only because there is a "captured" audience, but because this is an audience that might not be accessed otherwise. (Nearly one-quarter of those who had never had drug treatment had been to prison.) Furthermore, it is critical that prison populations also receive harm-reduction interventions, such as needle-exchange programmes, the provision of decontaminants, and the provision of condoms. In fact, in response to the severity and urgency of HIV/AIDS prevalence in prisons, the WHO recommended sweeping reforms including: the availability of condoms; needle-exchange programmes; and the provision of bleach to clean infecting equipment, if such provision is provided

for the community at large (WHO, 1993). By linking the recommendations to that provided for the community, the WHO affirmed the principle of equality, namely that the same prevention and control strategy should be applied in prisons as in other parts of society (Tomaševski, 1991). A similar approach has been recommended by other international organisations including the United Nations and the Council of Europe. Despite this consensus, there is still a difference between provision of preventive programmes in the wider society versus that for female prisoners, or more specifically for the female IDUs.

5. Conclusions

- 'C Those who have been to prison are a more marginalised group of women IDUs.
- C The economic burden of having children, as indicated by having ever been pregnant, increases the likelihood of having been to prison.
- C Early age of first injection has been linked to other risky behaviours and appears to be a determinant of imprisonment in this study.
- C The relationship between HIV seropositivity and previous imprisonment, though indirect in the current study, is important to consider and to study further for reasons of prevention.
- C Contrary to reports by Spanish prison officials, this study shows that injecting drug use does occur in Spanish prisons.

Regular partner HIV. Condom Use.Heroine+cocaine injection, cocaine injection and ever sharing needles: Stage 4: Hepatitis B. mber of clients per month.

a) The most parsimonious logistic regression were derived for each stage of the analysis using SPSS. For example, for Group 2 the Age variable was fixed, it could not be eliminated, whilst the new socioeconomic variables were selected through process of backward elimination. b) * p<0.05; ** p<0.01; *** p<0.001.

R	c) For models B.C and D the following variables were excluded from the equation: Stage 2: Regular sexual partner IDU	
Ľ	0	
only an nartner HIV Condom use Herninetoncaine injection social injection and over charing modules. Star	Ę	
ř	dej	
	SE	
ŗ		
<	anc	
2	0	
2	Ŧ	
3	e T	•
6	0110	
Ę	N.	
n n	DC.	
5	var	
3	iab	
	les	
-	Me.	
Į.	9	
÷	Ř	
5	ธี	
3	ደ	
2	fro	
3	a ct	
5	ਡ	
ļ	දී	
2	ati	
į	3	
1. 2	St	
	ğ	
ł	\sim	
	R	
3	j n	
5	7	
5	Sex.	
2	Jal	
7	ğ	
;	3	
•	4	
5	g	
•	g	
*	P	
3	202	
	Ē	
	È	
	3	
	S	
	g	
	و، د	
	3	
	N N	

312.969 11.190**	324.159 12.059**	337.096 34.034***	371.129 11.085*	-2LL Improvement Chi-Square
1.9 (1.1-3.6)*				STD the past year Yes No
2.2 (1.0-4.8)*				GINECOLOGICAL HISTORY Previous pregnancy Yes
3.0 (1.5-6.0)** 1.6 (0.8-3.2)	3.3 (1.7-6.4)*** 1.7 (0.8-3.4)			RISK BEHAVIOUR Age injecting drugs 16 or less 17 - 19 20 or more
1.6 (0.8-3,1) 2.2 (1.2-4,0)*	1.7 (0.9-3.2) 2.3 (1.3-4.2)**	1.8 (0.9-3.4) 2.3 (1.3-4.1)**		No fixed adress Fixed adress Income Legal Ilegal+prostitution
2.4 (1.0-6.0) 1.3 (0.5-3.4) 2.2 (0.9-5.5)	2.7 (1.1-6.5)* 1.6 (0.6-4.1) 2.5 (1.0-6.2)*	3.8 (1.6-9.0)** 2.0 (0.8-4.9) 3.1 (1.3-7.4)**		SOCIOECONOMIC STATUS Age leaving education 13 or less 14 - 15 16 - 18 19 + Above
2.9 (1.3-6.6)** 4.5 (1.9-10.9)*** 4.9 (1.8-13.4)**	3.5 (1.6-7.8)** 5.8 (2.4-13.7)*** 6.4 (2.4-17.2)***	3.2 (1.5-6.9)** 5.4 (2.3-12.3)*** 4.4 (1.8-11.1)**	2.3 (1.1-4.8)* 3.3 (1.5-7.1)** 2.7 (1.2-6.2)*	AGE Age group 24 or less 25-29 30-34 35+
OR (I.C.95%)	OR (I.C. 95\$)	OR (I.C. 95%)	OR (I.C. 95%)	
HODEL D GINECOLOGICAL HISTORY N-304	HODEL C RISK BEHAVIOUR N-304	HODEL B SOCIDECONOMIC STATUS N=304	HODEL A AGE N=304	

HIERARCHICAL LOGISTIC REGRESSION OF PREVIOUS INPRISONMENT ON RISK FACTORS

Chapter Eight

Determinants of Sharing Needles amongst Female IDUs

1. Introduction

There is a great deal of evidence that drug users in a majority of countries are modifying their risky practices. In a study conducted in the U.K., Hunter et al (1995) reported a decline in the proportion of IDUs that had shared injecting material during the six months prior to the interview over four consecutive years (from 1990 to 1993). Specifically they found a decline from 46.8 percent in 1990 to 39.3 percent in 1993. These figures are lower than those found in Berlin (Stark et al, 1990) where 88.0 percent of the women had at some time shared a needle or those found in Italy (Sasse et al., 1989) where 58.0 percent had engaged in needle sharing.

In our study 71.4 percent of the women had ever shared injection equipment, yet this proportion is reduced to almost half (34.2 percent) for those reporting equipment sharing in the six months prior to the interview. This rate is comparable to rates found by Estébanez, et al. (1996) in other E.E.C. countries: in the U.K. there was a reduction from 69.2 percent that had ever shared injection equipment to 38.5 percent who had shared in the previous six months; the reduction in Germany was from 56.4 percent to 26.9 percent; in Italy from 47.2 percent to 26.3 percent; and in France, where rates were quite a bit lower, from 34.4 percent to 10.0 percent.

Although these data provide encouraging evidence that a decline is underway in the risky behavior of IDUs, we can't forget that more than a third of the IDUs in our study were still continuing to engage in the risky practice of sharing needles. For this reason, the aim of the current chapter is to identify and characterise this residual group of women at risk and attempt to determine the direct and indirect factors associated with continuing to share needles. First, we have conducted a bivariate analysis in which we examine the relationship between the dependent variable "sharing needles in the past 6 months" and a variety of independent variables that could be considered of interest. Later we developed a multilevel logistic regression model to attempt to explain the direct and indirect determinants of our dependent variable.

2. Methods

The methods used for this study, including the bivariate and multivariate analyses, are described in Chapter 3. First, a bivariate analysis is conducted to identify significant relationships between "Sharing needles in the past 6 months" and selected independent variables. Then a multivariate analysis is conducted involving a logistic regression model with the following five groups of explanatory variables entered stepwise:

Model 1: Age

Biological age is the first explanatory variable introduced into the model in order to evaluate the effect of age on the dependent variable (sharing needles) and on the other independent variables. Previous research has identified a strong negative relationship between age and needle sharing.

Model 2: Socioeconomic status

Based on the literature, three variables were selected to define the general socioeconomic status of female IDUs: age left full-time education; source of income during past six months (illegal and prostitution versus legal); and, place of residence in past six months (no fixed abode versus abode).

Model 3: Marginalisation

In addition to socioeconomic status, five variables were chosen to characterise the extreme marginalised nature of the sample population. These variables were: age at first pregnancy (17 or less versus 18+); and four dichotomous variables: ever had clients; previous imprisonment; having a regular partner who was an IDU; and having a casual partner who was an IDU.

Model 4: Risk behaviours

Based on the literature, the following risk behaviours of sharing needles were selected for inclusion in the fourth model: ever have regular HIV-positive sexual partner; injecting in prison; cocaine injecting; heroin+cocaine injecting; age at first injection; actual regular sexual partner; regular PAP smear test; don't always use condom in vaginal intercourse with regular partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vaginal intercourse with casual partner; don't always use condom in vag

3. Results

First, the bivariate relationships with needle sharing in the past six months and selected independent variables are examined (see Table 8.1). Then findings from the regression analyses will be presented.

3.1. Sociodemographic characteristics and sharing needles in the past six months

Several of the sociodemographic characteristics had significant relationships with recent needle sharing. First, we found a strong association between age at the time of interview and having recently shared injection material, with those less than 25 years old more than three times more likely to have recently shared needles than the reference stratum of those 25 years old or older. We also found an association between level of education, reflected by the age of leaving school, and needle sharing. Those who left school before the age of 16 were two times more likely to have recently shared needles than those who left school after the age of 16.

The living situation of these women also proved to have a strong relationship with needle sharing. Those living at home with parents or other family members were much less likely to share needles compared to women who lived alone or with others. Furthermore those who claimed to have no fixed address were four times more likely to have recently shared needles than those with a fixed address.

The source of income is another factor possibly related with the behaviour of sharing needles. Those whose source of income is illegal or prostitution were slightly more likely to have recently shared needles than those with a legal source of income (non-significant finding).

3.2. Marginalisation variables

Amongst marginalisation variables, only previous imprisonment and the absence of drug treatment proved to have significant relationships with needle sharing. Nearly twice as many women who had been in prison engaged in recent needle sharing compared to those who had never been in prison.

The absence of prior treatment is considered to be an indicator of the degree of marginalisation of IDUs. We chose to use as a reference stratum "having received methadone maintenance treatment", as this group had the lowest proportion of having shared needles in recent months (24.6 percent). Results of the linear trend test showed that those that had received no treatment were twice as likely to have engaged in needle sharing in the last six months compared to the methadone group.

HIV status itself is not a factor likely to be associated with needle sharing in the last six months, yet a large number of those who knew of their HIV-positive status reported to have quit sharing injection material. For this reason, being HIV-positive seems to have a protective relationship with needle sharing, although the association is not statistically significant.

3.3. Drug-using habits

When drug use began at an early age (16 years old or less), we found a greater proportion of women that had recently shared injection equipment (45.7 percent) compared to 30.2 percent of those who began using drugs between the ages of 17 and 19 or 24.8 percent of those who began at the age of 20 of later. We used the latter group as reference stratum to conduct a linear trend test, finding that those of the intermediate age group had an OR of 2.0 and the youngest age group an OR of 2.6.

The use of certain non-injectable drugs such as amphetamines and benzodiazepines were associated with the habit of needle sharing with an OR of 3.4 and 1.8 respectively. But the strongest association was with those who had injected heroin in recent months, who were nearly three times more likely to have shared injection material than those who had not injected heroin. We did not find any significant relationships with the users of other types of intravenous drugs.

3.4. Sexual habits

For the analysis of sexual habits in our population, we found it worthwhile to differentiate between sexual habits with a regular partner, with a casual partner and with clients (in the case of prostitution). Having ever had a regular partner, whether HIV-positive or an IDU, did not have a significant relationship with sharing injection material. However, we did find a strong relationship with another risky behavior -- not always using condoms for vaginal penetration. In effect, those women who did not always use condoms with their regular partner were 3.5 times more likely to have shared injection material in the last 6 months.

Amongst those who had ever had a casual sexual partner, we had a similar finding. Though there was no direct association with needle sharing, there was a strong association for those women who did not always use condoms for vaginal penetration with their casual partners. This time the OR was 7.3.

We did find a positive association between having ever had sex with a client and the habit of needle sharing. Those who had ever practiced prostitution were nearly two times more likely to have shared needles than those who had never had clients. Although the age of starting into prostitution did not have a statistically significant association, we did see a clear tendency for the younger a woman began working as a prostitute the more likely she was to share needles. We also found a strong relationship between not always using condoms with clients for vaginal penetration and needle sharing, with an OR of 4.8.

3.5 Reproductive habits and infections

Using less viable methods of contraception such as "withdrawal" was related to the habit of needle sharing. We found that those who practiced withdrawal were two times more likely to have shared needles compared to those who didn't practice withdrawal.

The age of first pregnancy was strongly related to the behaviour of needle sharing. Pregnancy at a young age indicated an important degree of marginalisation and generally was accompanied by other risky behaviours. Those that had their first pregnancy at age 17 or younger were four times more likely to have shared needles than those who had their first pregnancy at the age of 18 or later.

It is evident from these results that not only are risky behaviours frequently associated but protective behaviours as well. For example, those women who had PAP smear tests taken regularly tended to share injection material in a lesser proportion (20.9 percent) than those who didn't have smear tests regularly (45.2 percent). In other words, there appears to be a protective relationship between having a regular smear test and sharing needles (OR=0.3; CI=0.2-0.5).

Having recently had an infection may be an indicator of other risk behaviours. The women who had had an infection during the past year were twice as likely to have shared needles than those who hadn't had an infection.

3.6. Origin of the sample

It appears that more of the women who attend the methadone services were reached by messages to avoid sharing injection equipment than those who don't use the methadone services. When the interview was conducted in a methadone service, the rate of women that had shared needles was 28.0 percent, whereas if the woman was interviewed in another drug service or on the street, the rate of sharing needles was over 35 percent.

3.7. Results of hierarchical logistic regression analysis

Table 8.2 presents the results of the four models. In the first model, we find that being 25 years old or older is a protective factor for the behaviour of sharing needles. In particular the group of 25- to 29-year-olds and the over 35-year-olds presented as significant protective factors, and the group of 30- to 34-year-olds did not reach significance in this first model but did become significant in the third model.

The second model adds variables representing socioeconomic status. The age of leaving school dropped out of the equation while the variables "abode" and "income" remained. The place of residence manifests itself as a variable with a great deal of influence over the behaviour of needle sharing, with those without a fixed address having an OR=5.8 (CI=3.0-11.2) compared to those with a fixed address. When the source of income was illegal or from prostitution, contrary to what was expected, a protective relationship appeared, although this factor did not become significant until the third model. Age maintained a level of significance similar to the first model.

In the third model those variables which express the degree of marginalisation of the women are added. A number of variables dropped out of the equation for lack of significance, including: having had a regular IDU sexual partner; having had a casual sexual IDU partner; not having ever received drug treatment; and having ever been in jail. The age at first pregnancy appears in this model as an influential factor in the habit of needle sharing. Those women who had their first pregnancy before the age of 18 were three times more likely to have engaged in needle sharing, than those who experienced their first pregnancy after 18. This model also introduces "ever practicing prostitution" as a risk factor with an OR=1.7. In this third model, significance is improved for age as well as for the socioeconomic variables.

In the fourth model we added other variables considered to be related to the habit of sharing injection material, including: having had a regular sexual HIV-positive partner; having injected drugs in prison; having injected cocaine; heroin or cocaine+heroin; place where subjects were recruited; and knowing one is HIV-positive. None of these variables were significant and consequently all were dropped out of the final equation.

Also included in this fourth model were variables that imply risky behaviour. These variables are: not getting a PAP smear test with regularity; not always using a condom in vaginal intercourse with a casual sexual partner; not always using a condom in vaginal intercourse with a regular sexual partner; not always using a condom in vaginal intercourse with clients; and not having a current regular sexual partner. Of these five variables, the first two appear to be important determinants. Not getting PAP smear tests with regularity carries an OR=2.3(CI=1.2-4.2) and not always using a condom in vaginal penetration with a casual partner has an OR=4.5(CI=1.6-12.7). The introduction of this new group of variables reinforces and improves the ORs of the variables from the marginalisation group, although it somewhat lowers the OR of the variable "no fixed address", and has no effect on the rest of the variables.

4. Discussion

Our study provides further evidence that IDUs are reducing their unsafe practices of sharing injection equipment. For the third of our sample which continues to share injection equipment, the bivariate and regression analyses presented in this chapter help to characterise this residual risk group by revealing a variety of related characteristics and behaviours. First of all, the younger IDUs appear to be more likely to share needles, a finding also reported by other Spanish studies (Hernández-Aguado et al., 1994; Bravo et al., 1994). One possible explanation for this is that the younger population (adolescents in particular) tend to perceive themselves as invulnerable to many health threats and are therefore more likely to engage in risk-taking behaviours. This might also explain the positive associations with other risk-taking behaviours, including not always using a condom with a casual partner, not having a regular smear test, and the use of amphetamines, benzodiazepines, and heroin injecting. Bravo et al. (1994) also noted less frequent condom use amongst IDUs who shared injecting equipment.

The regression analysis indicated that the residual risk group is quite economically oppressed and marginalised, having no fixed address and tending to have the first pregnancy at an early age. They were also more likely to have been to prison and to be prostitutes, according to the bivariate analysis. A relationship between needle sharing and prostitution was also reported by Bravo et al. (1994).

The relationship between not currently sharing needles and having been in methadone maintenance treatment provides evidence of the potentially positive impact drug treatment programmes can have on HIV risk behaviours. Of course the converse of this is also possible – those most likely to share needles are an oppressed, marginalised subgroup with less access to this type of drug treatment. As other studies have suggested, this residual risk group may be better accessed through needle exchange schemes. For example, Archibald et al. (1998) found that frequent attendance at a needle exchange program was associated with borrowing used needles, and amongst women it was independently associated with frequency of injection of any drug, shooting gallery attendance and having an illegal source of income.

Reducing the needle-sharing practices in this residual risk group of female IDUs poses a major challenge to future HIV prevention programmes. In particular, programme planners may want to target the younger population with strategies to increase their perceived susceptibility to contracting HIV, especially considering that this population may be simultaneously engaged in a variety of high risk behaviours. Although data on the impact of needle exchange programs on HIV incidence are inconsistent, the fact that they attract IDUs who have high risk behaviours or unstable lifestyles makes them a potentially important venue for preventive interventions with this hard-to-reach population.

5. Conclusions

- C Our data confirm that IDUs are making changes in their risky behaviours, particularly in the habit of needle sharing.
- C The residual risk group of female IDUs which continues to engage in needle-sharing practices is characterised by being younger, having a lower socioeconomic status, being more marginalised, and practicing other risk-taking behaviours.
- C The relationship between not always using a condom with a casual partner and recent needle sharing indicates a particularly high risk subgroup and suggests a possible route of transmission into the non-drug-using population.
- **C** The lower rate of sharing needles among those who had received methadone maintenance treatment provides further evidence of the potential benefits of HIV risk-reduction interventions within drug programmes.

.

Variable	Total	Sharing ne	edles	OR(CI.95%)	<u>) p</u>
		Yes(%)	No(%)		
Sociodemographics					
Age		,			
24 or less	73	50.7	49.3	3.6(1.5-8.7)	*** ³
25 -34	182	30.7	69.3	1.5(0.7-3.5)	
35 or more	49	22.4	77.6	1(reference)	
Age leaving education					
15 or less	170	40.6	59.4	1.9(1.2-3.1)	*
16 or more	129	26.4	73.6		
Living with					
Parents or relatives	89	18.5	81.5	0.3(0.2-0.6)	***
Other/alone	212	41.0	59.0		
Abode					
No fixed address	69	60.9	39.1	4.3(2.5-7.6)	***
Fixed address	235	26.4	73.6		
Income					
Illegal+prostitution	137	37.2	62.8	1.3(0.8-2.5)	n.s.
Legal	167	31.7	68.3		
Who finances your expens	es				
Yourself	182	30.2	69.8	0.6(0.4-1.0)	
Your parents	122	40.2	59.8		
Marginalisation					
Prison					
Yes	98	42.9	57.1	1.7(1.1-2.9)	*
No	206	30.1	69.9		
Injected in Prison					
Yes	32	46.9	53.1	1.3(0.5-3.0)	n.s.
No	66	40.9	59.1		
HIV +					
Yes	165	34.5	65.6	0.9(0.5-1.5)	n.s.
No	97	37.1	62.9		

Variable	Total	Sharing ne	edles	OR(CI.95%)) [
		Yes(%)	No(%)		
Drug treatment					
Methadone	65	24.6	75.4	l(reference)	*a
Other treatment	143	34.3	65.7	1.6(0.8-3.3)	
None	96	40.6	59.4	2.1(1.0-4.5)	
Drug-using habits					
Age began injecting drugs				i	
16 or less	81	45.7	54.3	2.6(1.4-4.8)	*
17 - 19	81	39.5	60.5	2.0(1.1-3.7)	
20 or more	141	24.8	75.2	1(reference)	
Use amphetamines					
Yes	22	45.7	54.3	3.4(1.4-8.6)	*
No	222	32.2	67.8		
Use benzodiazepines					
Yes	113	42.5	57.5	1.8(1.1-2.9)	*
No	191	29.3	70.7		
Heroin injecting					
Yes	240	38.3	61.7	2.7(1.4-5.3)	**
No	67	18.8	81.3		
Cocaine injecting					
Yes	160	32.5	67.5	0.8(0.5-1.4)	n.:
No	144	36.1	63.9		
Heroin+Cocaine injecting					
Yes	151	37.7	62.3	1.4(8.4-2.2)	n.
No	153	30.7	69.3		

Variable	Total	Sharing ne	edles	OR(CI.95%)	p
· ····································		Yes(%)	No(%)		
Sexual habits		()			
Ever had a regular sexual	partner				
Yes	293	33.4	66.6	0.4(0.1-1.4)	n.s.
No	11	65.5	45.5		
Ever had a regular sexual	HIV+ partner				
Yes	137	31.9	68.1	0.8(0.5-1.3)	n.s.
No	152	37.5	62.5	0.0(0.0 1.0)	
110		• • • •	-2.0		
Ever had a regular sexual	IDU partner				
Yes	257	34.6	65.4	1.1(0.6-2.2)	n.s.
No	47	31.9	68.1		
Frequency of condom use	with regular sez	xual partner			
-Vaginal intercourse	_				
Never/sometimes	122	36.9	63.1	3.5(1.3-9.7)	*
Always	35	14.3	85.7		
-Anal intercourse			<i></i>		
Never/sometimes	39	38.5	61.5	6.2(0.7-53.9)	n.s.
Always	11	9.1	90.9		
-Oral intercourse	102	38.2	61.8		n.s.
Never/sometimes	123	0.0	100.0		11.5.
Always	3	0.0	100.0		
Ever had a comial control r	artnar				
Ever had a casual sexual p	241	33.2	66.8	0.8(0.4-1.4)	n.s.
Yes No	62	38.7	61.3	0.0(0.11.1)	11.0.
INO	02	50.7	01.5		
Ever had a casual sexual H	-ITV+ partner				
Yes	53	35.8	64.2	1.4(0.7-2.7)	n.s.
No	186	28.5	71.5	、	
Ever had a casual sexual I	DU partner				
Yes	156	36.5	63.5	1.2(0.8-2.0)	n.s.
No	141	31.9	68.1		
Frequency of condom use	with casual sex	ual partner			
-Vaginal intercourse					***
Never/sometimes	48	54.2	45.8	7.3(2.7-19.3)	
Always	50	14.0	86.0		

.

.

Variable	Total	Sharing ne	edles	OR(CI.95%)	р
		Yes(%)	No(%)		
Sexual habits (continued	i)				
-Anal intercourse					
Never/sometimes	9	22.2	77.8		n.s.
Always	6	0.0	100.0		
-Oral intercourse					
Never/sometimes	68	34.8	65.2	1.1(0.2-4.7)	n.s.
Always	9	33.3	66.7		
Ever had a client					
Yes	152	41.4	58.6	1.8(1.1-2.9)	*
No	164	28.0	72.0		
Age at first client					
17 or less	20	55.0	45.0	2.7(0.9-8.7)	* 3
18 - 22	52	46.2	53.8	1.9(0.8-4.4)	
23 +	65	30.7	69.3	1 (reference)	
Frequency of condom use	with client				
-Vaginal intercourse					
Never/sometimes	26	61.5	38.5	4.8(1.7-13.2)	**
Always	52	25.0	75.0		
-Oral intercourse					
Never/sometimes	55	40.0	60.0	1.4(0.5-4.1)	n.s.
Always	22	75.9	68.8		
-Anal intercourse					
Never/sometimes	13	46.2	53.8	1.7(0.3-10-0)	n.s.
Always	9	33.3	66.7		

.

•

Variable	Total	Sharing ne	edles	OR(CI.95%) p
		Yes(%)	No(%)		
Reproductive habits a	and infections				
Ever used condom					
Yes	257	32.3	67.7	0.6(0.3-1.1)	n.s.
No	47	44.7	55.3	· · ·	
Regular periods					
Yes	99	28.3	71.7	0.7(0.4-1.1)	n.s.
No	205	37.1	62.9		
Contraception currently	у				
Yes	159	30.2	69.8	0.7(0.4-1.1)	n.s.
No	140	38.6	61.4		
Used withdrawal last 5	years				
Yes	123	43.1	56.9	1.9(1.2-3.1)	**
No	181	28.2	71.8		
Ever been pregnant					
Yes	230	35.7	64.3	1.3(0.7-2.3)	n.s.
No	74	29.7	70.3		
Age at first pregnancy					
17 or less	74	56.8	43.2	4.2(2.2-7.9)	***
18+	150	24.0	76.0		

.

Variable	Total	Sharing ne	edles	OR(CI.95%)	p
		Yes(%)	No(%)	,, . . ,, .	
Reproductive habits and	infections (con	tinued)			
Ever have miscarriages		,			
Yes	66	47.0	53.0	2.0(1.1-3.5)	*
No	238	30.7	69.3		
Ever had abortions					
Yes	119	28.6	71.4	0.7(4.0-1.1)	n.s.
No	185	37.8	62.2	, , ,	
Regular smear test					
Yes	134	20.9	79.1	0.3(0.2-0.5)	***
No	168	45.2	54.8		
Hepatitis B last year					
Yes	72	38.9	61.1	1.4(0.8-2.4)	n.s.
No	212	31.6	68.4		
TBC last year					
Yes	45	46.7	53.3	1.9(1.0-3.7)	*
No	239	31.0	69.0	、	
Some infection in last year					
Yes	148	40.5	59.5	2.0(1.2-3-4)	**
No	136	25.7	74.3		
Place where the interview	was conducte	ed			
Sample					
Drug services	107	35.5	64.5		n.s.
Methadone services	50	28.0	72.0		
On the street	147	35.4	64.6		

a:p significance of linear trend test

.

a) The most parsimonious logistic regression were derived for each stage of the analysis using SPSS. For example, for broup 2 the Aye variable income in the socioeconomic variables were selected trhough process of backward elimination.
b) () p<0.10; * p<0.05; ** p<0.01; *** p<0.01; *** p<0.01; *** p<0.01; *** p<0.01; *** p<0.001.
c) For models B.C and D the following variables were excluded from the model: Stage 2: Age left education: Stage 3: Casual partner 100. Regular partner 100. Previous imprissomment. Drug treatments: Stage 4: Regular partner HIV. Injecting in prisson. Never used condom. Sample. Cocaine injecting. Heroine injecting. To know HIV positivity. Heroine+cocaine injecting and Age first injection.

1	T	HIERWARCHICAL LOGISTIC REGRESSION OF SHARING NEEDLES LAST 6 MONTHS ON RISK	ST 6 MONTHS ON RISK FACTORS	
	HODEL A AGE N=304	HODEL B SOCIDECON. STATUS N=304		HODEL D RISK BEHAVIOUR N=304
	OR (I.C. 95 X)	OR (I.C. 95%)	OR (1.C. 95%)	OR (1.C.95%)
AGE GROUP 24 or less				
25-29 30-34 35+	0.3 (0.2-0,7)*** 0.6 (0.3-1.1) 0.3 (0,1-0,6)**	0.3 (0.2-0.6)*** 0.5 (0.2-1.0) 0.3 (0.1-0.6)**	0.3 (0.1-0.5)*** 0.5 (0.2-1.0)* 0.3 (0.1-0.6)**	0.2 (0.1-0.5)*** 0.4 (0.2-1.0)* 0.3 (0.1-0.8)**
SOCIOECONOMIC STATUS				0.3 (0,1-0.0)
No fixed adress Fixed adress		5.8 (3.0-11.2)***	5.3 (2.7-10.6)***	3.7 (1.9-7.1)***
[Income				
llegal+prostitution		0,6 (0.3-1.1)	0.4 (0.2-0.8)**	0.4 (0.2-0.8)**
MARGINALISATION Age_first_pregnancy				
10 f 17 or less			3.0 (1,6-5,5)***	3.7 (1.9.7.1)***
Yes			1.7 (1.0-3.2)	2.4 (1.2-4.7)
OTHER RISK BEHAVIOUR Sexo actual regular p				
, No				0.6 (0.3-1.1)
Regular smear tests Yes Mo				
Always condom casual p				2.3 (1.2-4.2)**
No				
Always condom regular p Yes				4,3 (1,0-16,7)
No No				2.6 (0.8-8.3)
Yes No				
-2ll	375 541	346 310	110 120	2.3 (0.9-6,2)
	15.052**	29.322***	18 NON+++	294,543

Chapter Nine

Determinants of Never Used Condoms amongst Female IDUs

1. Introduction

The sexual behaviour of female IDUs is comparable to that of non-IDUs, presenting similar attitudes and having similar problems with condom use. In all developed societies, it is assumed that it is the man who should initiate the use of a condom, making it difficult for a woman to impose her desire for one to be used. It is common to find a condom in a man's wallet, while it's unusual to find one in a women's purse, which would be considered libertine by society and by her own partner. The use of condoms with a steady partner is less frequent than with casual partners or clients, in the case of prostitution (Chapter 4: Tables 4.5, 4.6, and 4.8). Effectively, many women who use condoms regularly with their clients refuse to use them with their habitual partners, even when the partner is an IDU, thus exposing themselves to a higher risk of contracting HIV from their partner than from the practice of prostitution.

Although the last decade has seen an important reduction in certain risk behaviours of IDUs, such as needle sharing (Chapter 8), there has been less success in increasing the conduct of safe sex (Hart, 1989; McKeganey, 1990; Harris, 1990), especially the use of condoms with a steady partner. In some cases, perhaps in response to safe-sex campaigns, there has been a decrease in the number of casual partners, but rarely has there been a parallel increase in condom use (Chitwood, 1990).

In their study of women and AIDS, McCoy and Inciardi (1993) proposed a model of social determinants of sexual risk behaviours. They suggest that the social determinants include such factors as: attitudes about condom use and with whom they should be used; the extent to which drug use by the woman impedes her judgment about participating in risky behaviours; her dependence on an IDU partner for her support and that of her children; and cultural expectations about female partners during sex. The model also includes a number of other factors which have been found to be associated with unsafe sexual practices, including: limited education, financial dependency, and having child care responsibilities. The theoretical model proposed by these authors provides a useful framework for studying sexual risk taking by women.

In this chapter we have chosen to focus our study of sexual risk taking on those women who reported never using condoms in their sexual relations (n=47). It seems to us to be of vital interest to identify this residual group of risk and to analyse the characteristics and circumstances of these women in order to facilitate the design of new strategies intended to increase condom use in this group.

Thus the principal objectives of this chapter are to determine the profile of female IDUs who still have never used condoms and to identify the social determinants of not using condoms.

2. Methods

The methods used for this study, including the bivariate and multivariate analyses, are described in Chapter 3. "Never used condoms" is the dependent variable, and the logistic regression model involves the following four models of explanatory variables:

Model 1: Age

The biological age (coded into four categorical age groups: 24 or less; 25-29; 30-34; and 35+) was the sole variable used in this group in order to evaluate the effect of age on the dependent variable ("never used condoms") and on the other independent variables.

Model 2: Socioeconomic status

Three variables were selected to define the general socioeconomic status of female IDUs: age left full-time education (coded into four categorical age groups: 13 or less; 14-15; 16-18; and 19+); source of income during past six months (illegal and prostitution versus legal); and, place of residence in past six months (no fixed abode versus abode).

Model 3: Marginalisation

In addition to socioeconomic status, six variables were chosen to characterise the extreme marginalised nature of the sample population. These variables were: age at first pregnancy (17 or less versus 18+); drug treatment received (any treatment and none versus methadone maintenance); and four dichotomous variables (yes versus no): ever had clients; previous imprisonment; having a regular partner who was an IDU; and having a casual partner who was an IDU.

Model 4: Reproductive background

Specific known reproductive circumstances that have an influence on condom use were included in the fourth model. These dichotomous variables (yes versus no) are: to have regular periods; use contraception currently; ever been pregnant; and ever had abortions.

Model 5: Risk behaviours

Specific known risk behaviours thought to be related to "never used condoms" were included in this model: ever have regular HIV+ partner; cocaine injecting; heroine+cocaine injecting; age at first injection; actual regular sexual partner; use crack; ever share needles; know HIV serology; and place where interview was conducted.

3. Results of Bivariate Analysis

The bivariate relationships between "Never used condoms" and selected independent variables are examined in Table 9.1.

3. 1. Sociodemographic characteristics and never used condoms

As expected, we found a negative association between age at the time of interview and having never used a condom. In effect, the younger the age, the greater the risk of having never used a condom. Results of the linear trend test were very significant. Those under 25 years were 4.3 times more likely to have never used condoms than those over 35 years. This excess of risk decreased sharply for the other age strata, with both the 25- to 29-year-olds and the 30- to 34-year-olds presenting an OR of 1.6 when compared to those over 34 years.

The relationship with age at the time of leaving school was again negative, but not significant. There was also no clear association with the persons with whom they live, nor the place where they live. Nor did we find a significant relationship when we looked at source of income, comparing illegal+prostitution to legal.

Knowing that it is more common for prostitutes to use condoms than other marginalised groups, we decided it was logical to regroup the source of income variables to compare illegal versus legal+prostitution. This produced a strong positive association, with illegal having an OR of 2.5 compared to the group legal+prostitution.

There was also a significant association with the variable of who finances one's expenses. When the woman is responsible for her own expenses, the risk of having never used condoms was cut in half when compared to being dependent on someone else. In this case, one must be aware of the potential for confounding, either by the age of the woman or by the fact that women who cover their own expenses may be working as prostitutes and as we just mentioned, prostitutes have a higher rate of using condoms.

3.2. Marginalisation variables

Of the three variables which we considered to be indicators of marginalisation, two (having ever been to prison and having injected in prison) were not significantly associated with having never used a condom.

Nevertheless, we found a strongly significant association between knowledge of one's HIV status and condom use. Those female IDUs who knew of their seropositivity were much less likely to have never used a condom than those who didn't know they were seronegative or had never been tested.

3.3. Drug-using habits

No significant relationships were found between condom use and age of initiating drug use.

Having used methadone in the six months prior to the interview lowered the risk of having never used a condom by one-fifth. This might be explained by the fact that the great majority of these women had been enrolled in drug or methadone treatment services during this time, which makes them more likely to have received safe-sex messages as well as to have had better access to condoms. Considering that women who have been in methadone treatment are at a lower risk for having never used a condom (as just mentioned), we compared this group with those that received any other type of treatment and with those who never received any treatment. The group that had received other treatment presented a slight positive association though not significant, while the group that had never received any treatment presented a strong positive association with an odds ratio of 4.5 for having never used a condom. The result of the linear trend test between these groups was significant.

Those women who used crack in recent months were three times more likely to have never used a condom. These data coincide with data obtained in a Spanish study (Bravo, 1996) in which crack users, of both sexes, presented an association with not always using a condom.

Although no relationship was found with having ever shared injection material, there was a positive association with having shared in the past six months. Those who reported recent sharing of injection equipment were nearly two times more likely to have never used condoms.

3.4. Sexual habits

Having a steady sexual partner was positively associated with having never used a condom, even if the partner was an IDU. The corresponding odds ratios for these variables were 1.8 and 1.6, respectively (though non-significant). In other words, there is a greater probability of having never used a condom when one has maintained sexual relations with a stable partner, and this holds true even if the partner is an IDU. One must realise that many of these women share injection material with their partner, and consequently may consider the use of a condom unnecessary (in that they are already exposing themselves to whatever risk their partner poses). The situation changes when the stable partner is HIV-positive. In this case, the odds are halved, suggesting that knowledge of a partner's HIV positivity doubles the likelihood of condom use.

When the woman maintains sexual relations with casual partners, there is a clear increase in condom use, which is maintained for IDU partners as well. In both cases, when one had a casual partner and when this partner was an IDU, the odds ratios for having never used a condom are 0.2. When the casual partner was HIV-positive, the use of condoms was maintained equally high, with an OR=0.4. In a study conducted by Helal et al. (1995) in France, it was found that 33 percent of the subjects had never used condoms with their casual partners. Nevertheless condom use was significantly greater when the partner was seropositive (38 percent versus 12.7 percent).

As we have already mentioned, one decisive factor for the use of condoms is the practice of prostitution, which presented a strong negative association with having never used a condom (OR=0.1), in fact one of the strongest of the study.

3.5 Reproductive habits

Women with regular menstrual periods use condoms less than those who have irregular cycles. The OR for those who have regular periods is 2.8. This association demonstrates one of the principal reasons for condom use - contraception, as condoms are used more amongst women who can't trust the regularity of their cycle. This conclusion is confirmed by evidence that those women who currently use some type of contraception present a strong negative association with having never used a condom. In other words, the condom is a contraceptive method which has been widely diffused in this population.

Those women who have ever been pregnant, as well as those who have ever had an abortion, were half as likely to have never used condoms than those who have never experienced these circumstances.

3.6. Origin of the sample

Of the women recruited in methadone services, only 6 percent had never used a condom. Therefore this stratum was selected as a reference for the linear trend test, which indicated that those recruited in other drug services were more than twice as likely to have never used a condom, while those recruited from the street were four times more likely.

Variable	Total	Ever	used condom	OR(CI.95%)	<u>p</u>
		No (%)	Yes(%)		
Sociodemographics					
Age		,			
24 or less	73	27.4	72.6	4.2(1.2-15.9)	**
25 - 29	110	12.7	87.3	1.6(0.5-6.3)	
30 - 34	72	12.5	87.5	1.6(0.4-6.7)	
35 +	49	8.2	91.8	1 (reference)	
Age leaving education					
13 or less	88	19.3	80.7	1.6(0.7-3.6)	n .:
14 - 15	82	15.9	84.1	1.3(0.5-2.9)	
16 +	132	12.9	87.1	l(reference)	•
Living with					
Parents or relatives	92	14.1	85.9	0.9(0.4-1.7)	n.s
Other/alone	212	16.0	84.0		
Abode					
No fixed address	69	20.3	79.7	1.6(0.8-3.1)	n .:
Fixed address	235	14.0	86.0		
Income					**
Illegal	74	25.7	74.3	2.5(1.3-4.8)	**
Legal + prostitution	230	12.2	87.8		
Who finances your expense					-
Yourself	182	12.1	87.9	0.5(0.3-1.0)	*
Your parents	122	20.5	79.5		
Marginalisation					
Prison				0.0(0.4.1.5)	
Yes	98	13.3	86.7	0.8(0.4-1.5)	n.s
No	206	16.5	83.5		
Injected in Prison				1 2(0 4 4 5)	
Yes	32	15.6	84.4	1.3(0.4-4.5)	n.s
No	66	12.1	87.9		
HIV+	• • • •	0.0	01.0	0.2(0.2,0.7)	**
Yes	166	9.0	91.0	0.3(0.2-0.7)	
No	138	23.2	76.8		

Table 9.1 Proportion of women who never used condom by selected variables

Variable	<u> </u>		used condom	OR(CI.95%)	p
		No(%) Yes(%)		
Drug-using habits					
Age started injecting dru	ıgs				
16 or less	81	16.0	84.0	1.0(0.4-2.3)	n.s.
17 -19	81	14.8	85.2	0,9(0,4-2,1)	
19 or more	141	15.6	84.4	l(reference)	
Use methadone					
Yes	72	4.2	95.8	0.2(0.1-0.6)	**
No	232	19.0	81.0		
Use crack					
Yes	46	30.4	69.6	3.0(1.4-6.1)	**
No	258	12.8	87.2		•
Ever shared needles					
Yes	217	12.9	87.1	0.6(0.3-1.1)	n.s.
No	86	20.9	79.1		
Shared needles in last 6	months				
Yes	104	20.2	79.8	1.7(0.9-3.2)	n.s
No	200	13.0	87.0		
Drug Treatment					
Methadone	65	7.7	92.3	l(reference)	***
Other treatment	143	11.2	88.8	1.5(0.5-4.9)	
None	96	27.1	72.9	4.5(1.5-14.2)	
Sexual habits					
Ever had a regular sexua	l partner				
Yes	293	15.7	84.3	1.8(0.2-14.9)	n.s.
No	11	9.1	90.9		
Ever had a regular sexua	l IDU partner				
Yes	257	16.3	83.7	1.6(0.6-4.4)	n.s.
No	47	10.6	89.4		
Ever had a regular sexua	•				×
Yes	138	10.9	89.1	0.5(0.2-0.9)	*
No	152	21.1	78.9		
Ever had a casual sexual	-				***
Yes	241	10.0	90.0	0.2(0.1-0.4)	ጥጥቸ
No	62	37.1	62.9		

Table 9.1 (continued) Proportion of women who never used condom by selected variables

Table 9.1 (continued) Proportion of women who never used condom by selected variables

Variable	Total	Ever u	ised condom	OR(CI.95%)	_p
) Yes(%)		
Sexual habits (continued	l)				
Ever had a casual sexual	DU partner	•			
Yes	156	6.4	93.6	0.2(0.1-0.4)	***
No	141	26.2	73.8	· · ·	
Ever had a casual sexual	HIV+ partner				
Yes	53	9.4	90.6	0.4(0.1-1.0)	*
No	186	21.0	79 .0		
Ever had a client					
Yes	140	4.3	95.7	0.1(0.1-0.3)	***
No	164	25.0	75.0		•
Place where the intervie	www.was.conducte	ન			
					
Methadone services	50	6.0	94.0	1(reference)	
Drug services	107	13.1	86.9	2.4(0.6-10.9)	
On the street	147	20.4	79.6	4.0(1.1-17.4)	** ^a
Reproductive habits					
Regular periods					
Yes	99	25.3	74.7	2.8(1.5-5.3)	***
No	205	10.7	89.3		
Contraception currently					***
Yes	159	8.2	91.8	0.3(0.2-0.6)	***
No	140	22.9	77.1		
Ever been pregnant		10.6	07.4	0.5(0.2.0.0)	*
Yes	230	12.6	87.4 75.7	0.5(0.2-0.9)	-
No	74	24.3	75.7		
Ever had abortions	110	10.1	90.0	0.5(0.2.0.0)	*
Yes	119	10.1 18.9	89.9 81.1	0.5(0.2-0.9)	-
No	185	10.7	01.1		

^a p significance of Chi-square of Linear trend test

4. Results of Hierarchical Logistic Regression Analysis

Table 9.2 presents the odds ratios and confidence intervals of each of the variables entered into the regression model and the new odds ratios, as subsequent groups of variables are entered into the regression.

Age (Introduction of Model 1)

We found a negative association between having never used condoms and age at the time of the interview. It was the youngest group (24 or less) who were most likely to have never used condoms, with an OR of 4.2 (CI=1.4-13.3). For the other age groups the association was weaker. As we proceed to add the other models, we see that this association loses its intensity, becoming non-significant by the time the reproductive variables of Model 4 are introduced. When the risk behaviours and interaction factor are incorporated in Model 5, the value of the OR for the youngest group decreases to 1.3 (CI=0.3-5.6), no longer significant.

Socioeconomic Status (Introduction of Model 2)

For those who don't have a fixed address, the risk of having never used a condom is higher than for those who do. This association becomes stronger with successive models, such that by the final model the OR has increased more than 5 points (from 2.1 in Model 2 to 7.2 in Model 5).

The source of income (illegal+prostitution versus legal) presents a protective OR (0.5; CI=0.2-1.0), which is explained by the more frequent use of condoms among those practicing prostitution, although this association is weak and continues to weaken with successive models, becoming practically non-existent in the final model (OR=0.9;CI=0.3-2.7). The age of leaving school never entered the equation in the second model.

Marginalisation (Introduction of Model 3)

Never having had a casual IDU partner presented as a variable positively associated with the dependent variable (OR=4.4;CI=1.9-10.6). This association is maintained in subsequent models, though weakening slightly in the final one (OR=3.7;CI=1.4-9.8). We know from previous studies, that even if condom use is infrequent with a steady partner, it is more frequent with casual partners, and even more frequent with clients. (In our study condoms were always used for vaginal intercourse with the habitual partner by 22.3 percent, with the casual partner by 51.0 percent and with clients by 66.7 percent).

The variable with the strongest association with the dependent variable in the various models is the practice of prostitution. Those that don't practice prostitution present an OR of 9.3 (CI=3.3-26.7) in Model 3, increasing to 12.1 (CI=3.1-47.2) in the final model. The following variables remained out of the equation: habitual IDU partner; previous incarceration; age at first pregnancy; and having received any drug treatment.

Reproductive Background (Introduction of Model 4)

It seems evident that the regularity of the menstrual cycle creates a certain confidence in women which leads them to risk having sexual relations without contraception. Thus we found that those women with regular menstrual cycles presented an OR=2.5 (CI=1.1-5.6) for the dependent variable, compared to those who have irregular cycles. In the subsequent and final model, the level of association of this variable hardly varied (OR=2.6;CI=1.0-6.9).

Those that were not using any methods of contraception at the time of the interview presented a positive association with having never used a condom (OR=2.5;CI=1.1-5.6). This association is duplicated in the subsequent model, when we control for risk behaviours (OR=5.2;CI=1.9-14.1). This finding suggests that the condom plays an important part in the contraceptive methods utilized by this population.

Previous pregnancies appear to predispose one to use condoms, perhaps because many of these women are all too familiar with unwanted pregnancies resulting from a lack of contraceptive planning. This would also explain the positive association found between having never been pregnant and having never used condoms (OR=2.2;CI=1.0-5.0), which is maintained in the subsequent model (OR=2.1;CI=0.8-5.2).

The variable of having had a previous abortion did not enter the equation.

Risk Behaviour (Introduction of Model 5)

The proportion of women interviewed in methadone centres who never used condoms was the lowest of the three groups (6.0 percent), which is why we used this group as a reference stratum. A significant association was not found between the group of women recruited in other drug centres and the dependent variable when compared to those recruited in the methadone centres (OR=1.1;CI=0.2-5.4).

However, a strong positive association was found between those recruited in the street and the dependent variable (OR=4.5;CI=1.5-13.0). Although the women recruited in the street probably belong to a more marginalised group than those recruited in the centres, one can't discard the possibility that those recruited in the treatment centres may feel morally coerced to over-report the practice of protective behaviours (or conversely, under-report the practice of risky behaviours) either for fear of rejection by the centre, or with the intention of pleasing the centre.

The groups which began injecting drugs earlier are the most marginalised and with risk behaviours more evident at every level. Thus when the age of the first injection was 16 years or less, we found a noticeably higher risk of having never used condoms if we compare it with the group who began injecting themselves at age 20 or later (OR=2.7;CI=0.9-8.0). However, the group who began injecting between ages 17 and 19, did not present a clear association compared to the reference group (OR=0.8;CI=0.2-2.3).

Amongst possible interaction effects, a significant interaction effect was identified between the variables "knowing one's serostatus" and "having ever had a steady partner". This interaction effect was entered into the final model and maintained a significant relationship with "having never used condoms". Specifically, those who knew of their <u>seronegativity</u> and who had <u>never</u>

had a steady partner presented an important risk of never having used condoms (OR=5.0;CI=1.9-13.5), compared to the rest.

Remaining out of the final equation were the following variables: use of crack in the previous six months; having ever exchanged injection material; having a steady HIV-positive partner; having injected cocaine or heroin+cocaine in the last six months; having a current steady partner and knowing (or believing to know) one's HIV status.

5. Discussion

The results of the bivariate and regression analyses help us to characterise the residual risk group of female IDUs who continue to be at high risk for HTV due to their failure to use condoms. One of the most important factors appears to be not practicing prostitution. The use of condoms seems to be a well-diffused behaviour among prostitutes in our study, at least in their relations with clients. Other studies have also noted that condom use tends to be more frequent with clients and casual partners than regular partners (Rhodes et al., 1994; Paone et al., 1995).

Another striking feature of this residual risk group who has never used condoms is that they are quite marginalised. They tend not to have a fixed address, they began injecting drugs at an early age, and they were recruited from the street. In other words, this is a subgroup which may have less access to HIV prevention information as well as less access to condoms themselves. Other studies have also documented low rates of condom use among homeless women (Nyamathi et al., 1997; Fisher et al., 1995). Future educational campaigns should be aware of the important challenge of reaching this subgroup.

Conversely, we can also interpret from the above results that those who were enrolled in any drug treatment programme were more likely to use condoms, suggesting the potential positive effect these programmes can have on other risk behaviours. Of course, this finding should be qualified by the possibility of over-reporting bias at the drug centres, as mentioned previously. Still, the potential for drug treatment programmes to be used as a venue for implementing HIV risk reduction cannot be overstated. In fact, Eldridge et al. (1997) have reported success in reducing sexual risk for HIV infection among female entrants to an inpatient substance abuse treatment centre using a brief skills training intervention.

Analysis of the reproductive history provides further insights into this subgroup. Those who have never used condoms tend not to be using any contraception, are perhaps relying on the rhythm method due to their regular periods, and have never been pregnant. Moreover condom use would appear to be a widely diffused (91.8 percent) method of contraception amongst those who reported to be currently using contraception.

Younger women originally appeared to be at higher risk for having never used condoms, but the regression analysis revealed that age was merely a marker for other more important variables, such as prostitution, ever being pregnant, irregular menstrual cycles, frequenting of drug/methadone centers, etc., which become more common over time.

A number of studies have noted that condom use is more frequent or consistent among those aware of their seropositivity when compared to those who were not aware or were seronegative (Van den Hoek, 1990; Wilson et al., 1998). Further evidence of this relationship is provided by a study carried out in Europe by Desenclos (1993), who found that IDUs who knew of their seropositivity were three times more likely to always use a condom than those who had never been tested. They also conclude that the seronegative IDUs tend to be less accepting of condom use than other risk-reducing behaviours, such as not sharing needles. A study carried out amongst female IDUs participating in New York City syringe exchange programmes indicated that knowing one's HIV status was one of three independent predictors of consistent condom use with primary partners (Paone et al., 1995). Shedding further light on this relationship, Friedman (1994) found that condom use was quite frequent among seropositive IDUs, especially if the partner was not an IDU. The authors offered two possible explanations: the first being an altruistic desire to protect one's partner; and the second being a strong predisposition on the part of the partner to use condoms knowing that he/she is maintaining relations with someone who presents a high risk.

In our study, we also found a significant bivariate relationship between knowing one's serostatus and condom use. However, the regression analysis revealed that the interaction between knowing one's serostatus and having ever had a steady partner was a more important direct determinant of condom use. In other words, it was not merely knowing one's serostatus that determined condom use, but knowing of one's seronegativity combined with having never had a steady partner which had a direct effect on having never used a condom. This finding suggests that women who knew they were seropositive and had a steady partner were more likely to have used condoms probably to protect the steady partner.

Another important determining factor revealed in the regression analysis was never having had a casual IDU partner. Although this is not particularly useful information as a descriptor of the subgroup who has never used condoms, we can make some inferences. As the bivariate analysis revealed, condoms are more likely to be used with casual partners or clients than regular partners. The use of condoms in these situations, particularly with IDU or HIV-positive casual partners, appears to be driven by perception of risk. Unfortunately, perception of risk is lower with a known partner, even if the known partner is an IDU. Future interventions should help these women to more accurately assess the risk of not using condoms with steady partners. Furthermore, future studies could elaborate on the reasons why these women don't use condoms with a regular partner and propose appropriate strategies for correcting this problem.

Ideally, women need a form of barrier contraception which they can control, not requiring the man's participation or initiative. The female condom has emerged in recent years as a possible solution. Studies have indicated that it provides contraceptive efficacy in the same range as other barrier methods when used correctly and has shown to be acceptable amongst women and their partners, though cost remains an issue (Farr et al., 1994; Ray et al., 1995). Of course, the female condom still requires male willingness to use it making safer sex negotiation skills as important as ever. Furthermore, future studies should compare the acceptability of female condoms versus male condoms amongst steady partners.

6. Conclusions

From the simultaneous observation of the bivariate analysis and the multilevel logistic regression model we can extract the following conclusions:

- **C** Although the use of condoms is relatively accepted with clients and casual partners, it is still not as common among stable partners, even when the partner is high risk (IDU or HIV +).
- C The visit by a female IDU to a drug/methadone treatment centre apparently has a positive effect on her risk behaviours.
- **C** The residual risk group of female IDUs which has never used condoms tends to be a marginalised group having no fixed address, having begun injecting drugs at an early age, and having been recruited from the street.
- **C** The female IDU who knows she is seropositive and has a steady partner is more likely to use condoms.

	H	ERARCHICAL LOGISTIC REGRESSI	HIERARCHICAL LOGISTIC REGRESSION OF NEVER USED CONDOM ON RISK FACTORS	ISK FACTORS	
	HODEL A AGE N-304	Hodel B Socioecon, status N=304	HODEL C MARGINALISATION N-304	HODEL D REPRODUCTIVE	NODEL E RISK BEHAVIOUR
	OR (I.C. 95%)	OR (1.C. 95 X)			
AGE GROUP 24 or less			UN (1.C. 304)	(1. U. 1. U. 934)	(\$C6.7.1) MO
24 or less 25-29	4.2 (1.4-13.3)* 1.6 (0.5-5.3)	4.8 (1.5-15,6)** 1.7 (0.5-5.4)	4.2 (1.2-14.4)*	2.7 (0.7-9.6)	1.3 (0.3-5.6)
30-34 35+	1.6 (0.5-5.5)	1.6 (0,4-5,4)	1.8 (0.5-6,8)	1.7 (0.4-6.5)	1,1 (0,3-4,0) 1,1 (0,3-4,7)
SOCIDECONOMIC STATUS					
No fixed adress		2.1 (0.9-4.7)	4.9 (1.8-13.6)**	5.6 (1.8-17.3)**	7.2 (1.8-28.2)**
Income Income		1			
Legal + prostitution		0.5 (0.2-1,0)	0.9 (0.4-2.1)	0.8 (0.3-2.1)	0 9 (0.3-2.7)
MARGINALISATION Casual partner IDU					
NO			4.4 (1.9-10,2)***	4.7 (1.9-11.7)***	3.7 (1.4-9.8)**
Ever had clients					
Yes			9,3 (3,3-26,7)***	8,7 (2,7-28,2)***	12.1(3.1-47.2)***
REPRODUCTIVE BACKGROUND Regular periods					
No				2.5 (1.1-5.6)*	2.6 (1.0-6.9)
Contracepcion currently Yes					
No Ever been pregnant				2.5 (1.1-5.6)*	5.2 (1.9-14.1)**
Yes No				2.2 (1.0-5.0)	2.1 (0.8-5.2)
RISK BEHAVIOUR					
Methadone services					
On the street					1.1 (0.2-5.4)
Age injecting drugs					
17 - 19					0.8 (0.2-2.3)
20 or more INTERACTION					
To be seronegative and No actual regular partmer					5.0 (1.9-13.5)***
-211 Improvement Chi-Souare	251.547	246.640	202.451	178.880	154 182
a) The most parsimonious logistic regression were derived for each stage of the analysis using			94.120	11.010	74, 22, 72

uv = p-v.vs; == p-v.vs; == p-v.vi; === p-v.vi; === p-v.vs; === p-v.vs; === p-v.vs; == p-v.vs; === p-v.vs; == p-

Chapter Ten

Determinants of Never Received Treatment amongst Female IDUs

1. Introduction

This chapter introduces the topic of drug treatment. The primary issues which we wish to consider include: access to and utilisation of treatment amongst IDUs, success in treatment; and the effect of treatment on risk behaviours.

There exists little documentation of the utilisation of drug-dependency services by IDUs. although it seems that uptake is low. Of the 104 confirmed seropositive IDUs examined by Rhodes et al. (1993), 43.1 percent had never received help or treatment for drug dependency and 52.2 percent had not been formerly tested for HIV. It was also observed in London that 42 percent of heroin addicts and 75 percent of cocaine users had never received treatment for drug dependency (Strang et al., 1994). In a study carried out in Madrid amongst IDUs, only 42.0 percent of those interviewed in the street had received some type of help or treatment for drug dependency. Of these: 21.6 percent had received treatment in a municipal centre for drug dependence; 21.1 percent had received methadone treatment; 22.9 percent had been to a hospital detoxification unit; 22.9 percent had received ambulatory detoxification; and 30.0 percent had received care in the community (Zunzunegui et al., 1993). According to the results of a European study (conducted between 1989 and 1990 in 12 countries) on the knowledge, attitudes and beliefs of IDUs about measures taken for HIV prevention, the low utilisation of drugdependency services could be due to the low opinion of these services amongst IDUs. By comparison, the direct provision of sterile injection equipment was generally welcomed (Richardson et al., 1994).

Inequalities between men and women also occur with regard to attending services and to assimilating information. A multi-site U.S. study on access to care for drug abusers with HIV reported that multiple barriers in access to service exist for drug-using women (Weissman et al., 1995). The most important barriers were cost and waiting times. The authors also noted that significantly fewer women than men received pre- and post-test counseling and were advised to get medical services after their first positive HIV test.

According to the data recorded amongst drug addicts interviewed in Madrid in 1990, women had a lower rate of uptake into drug-dependency services, although the differences were not statistically significant (Zunzunegui, 1993). About 50 percent of men and 40 percent of women in the study were recruited from drug-dependency centres, suggesting a lower rate of access and/or utilisation for these women. Although data on the use of social services amongst these women was not recorded, it is probable that drug-dependent women have lower rates of access due, in part, to feelings of guilt and/or the fear that the state might consider removing their children.

Gender differences in the process and response to drug treatment have also been examined. As men have traditionally accounted for a higher proportion of those addicted to heroin and therefore a higher proportion of those entering treatment facilities, it has only recently been questioned whether treatment intervention methods and goals are too "male-oriented" and not appropriately geared for women. In general, women are found to be more likely to drop out of treatment than men, although when they do carry through with treatment they have similar rates of success (reviewed in Anglin et al., 1987). Studies of the factors which contribute to positive treatment outcome for women, though few, have consistently indicated that a partner's involvement in entry into treatment and continued support in attending treatment are critical (reviewed in Anglin et al., 1987).

Furthermore, a more recent study into the apparent gender paradox in drug treatment that although women have more "risk factors" associated with drug use relapse they are no more likely to relapse than men, found that women become more engaged in treatment specifically by participating more in group counseling (Fiorentine et al., 1997). They did not find support for the drug severity hypothesis (women are less likely to relapse because their drug-use histories are less severe) or the social support hypothesis (women are less likely to relapse because they have higher levels of social support, facilitating recovery). The authors attribute the differences in treatment engagement to gender norms concerning help-seeking, personal independence, strength, and control.

The relationship of drug treatment to HIV risk behaviours has not been clearly defined. First of all, most studies have failed to adequately control for differences due to selection bias between those seeking treatment and not seeking treatment as well as between those recruited at different sites of treatment. In fact, few studies have examined the differences between those who seek treatment and those who do not. Ross et al. (1993), comparing those who had never been in treatment with those who had previously been in treatment and with those currently in treatment, found that those who had never been in treatment were younger, used and injected drugs less frequently, had a lower level of HIV risk-related injecting behaviours (though no difference in risk-related sexual behaviours), and reported less involvement in the drug subculture.

Most research examining the relationship between treatment and risk behaviour has come from evaluations of methadone maintenance treatment (MMT). In general, these studies have found that MMT has a positive effect on risky drug-taking practices, though no effect on risky sexual practices. For example, Baker et al. (1995) found that those currently enrolled in MMT reported a lower frequency of injecting and a higher frequency of cleaning injection equipment with bleach than those previously or never enrolled in MMT. No differences were found in needle sharing practices or sexual risk-taking behaviour.

The overall aim of this chapter is to shed further light on the relationship between drug treatment and HIV risk-taking behaviours. First, we will examine determinants of having never attended any treatment to identify differences between those who seek or have access to treatment and 159

those who do not. Secondly, we will look for direct relationships between treatment and risk behaviours to estimate the potential impact of treatment on HIV risk.

2. Methods

The methods used for this study, including the bivariate and multivariate analyses, are described in Chapter 3. Drug treatment is defined broadly, including services ranging from short-term counseling to residential detoxification and rehabilitation. We have also included methadone treatment and any harm-reduction programmes, such as those which supply needles or condoms. "Never had any treatment" is the dependent variable in the multilevel logistic regression model which involves the following four models of explanatory variables:

Model 1: Age

In order to evaluate the effect of age on the dependent variable ("never had treatment") and on the other independent variables, the biological age (coded into four categorical age groups: 24 or less; 25-29; 30-34; and 35+) was the sole variable used in this group.

Model 2: Socioeconomic status

Three variables were selected to define the general socioeconomic status of female IDUs: age left full-time education (coded into four categorical age groups: 13 or less; 14-15; 16-18; and 19+); place of residence in past six months (no fixed abode versus abode); and living with children (coded into three groups: she lives with her children; she doesn't live with her children; and she has no children).

Model 3: Marginalisation

In addition to socioeconomic status, three variables were chosen to characterise the extreme marginalised nature of the sample population. These variables were: age at first pregnancy (17 or less versus 18+); and two dichotomous variables (yes versus no): having a current regular sexual partner; and having a casual partner who was an IDU.

Model 4: Risk behaviours

Specific known risk behaviours that had significant associations with "never had treatment" in the bivariate analysis were included in the fourth model. Six dichotomous variables (yes versus no): ever have regular HIV-positive partner; use alcohol; cocaine injecting; heroine+cocaine injecting; never use condom; and ever sharing needles in last six months; age at first injection (coded into three groups: 16 or less, 17 - 19, and 20 or more); and HIV result (coded into three groups: positive, negative, and unknown).

3. Results of the Bivariate Analysis

The bivariate relationships with drug treatment (never versus ever) and selected independent variables are examined in Table 10.1. A summary of the results of Table 10.1 are presented in the following sections.

3.1. Sociodemographic characteristics

Age is a factor strongly associated with having never had drug treatment (the linear trend test is very significant). The youngest group (age 24 or younger) was more than ten times more likely to have never had treatment when compared to the oldest group (35 years or more). The other two age groups do not present an elevated risk, both having non-significant associations. So it is the youngest group in which almost half (47.9 percent) have never received any treatment. This percentage drops sharply for subsequent age groups, with 18.2 percent for 25- to 29-year-olds, 11.1 percent for 30- to 34-year-olds, and 8.9 percent for 35 and over.

The age of leaving full-time studies did not present a clear association with the dependent variable. The linear trend test was not significant; although the highest percentage of no treatment corresponds to the group who had left their studies earliest (31.8 percent).

3.2. Marginalisation variables

In general, we observed that the group of characteristics reflecting greater marginalisation has a higher risk of having never had treatment for drug addiction. In effect, those that don't have a fixed address are twice as likely to have had no treatment compared to those who do have a fixed address. On the other hand, if a woman lives with her children, the percentage of cases without anti-drug treatment is lower (12.2 percent), than if she lives without her children (18.8 percent), and still higher if she doesn't have children (28.7 percent). It could be that having children, especially when they live at home, provides a stimulus for a woman to decide to seek treatment.

No significant associations were found between source of income or previous imprisonment and the dependent variable.

3.3. Drug-using habits

The age of beginning drug use is significantly associated with having received drug treatment. The linear trend test is significant, with the women who started earlier less likely to have never received treatment. Of the group who began using drugs at age 16 or younger, 14.8 percent had never received treatment, compared to 22.2 percent for those who began between ages 17 and 19, and 26.2 percent for those who began at age 20 or later.

Although the association between having ever shared injection material and having never had treatment is not significant, we do find a clear association when this risk behaviour was undertaken in the six months prior to the interview. Those who had recently shared needles were over two times more likely to have never received treatment.

Having injected heroin+cocaine in the six months prior to the interview nearly doubles the odds of having never attended treatment though the association doesn't quite reach significance (p=0.06).

3.4. Sexual habits

Women that have never had a regular sexual partner have a much higher rate of having never had drug treatment (54.5 percent) than those who have had a regular sexual partner (26.5 percent). Interestingly, when the regular partner is HIV-positive, the proportion that have not received treatment is almost one-third (10.9 percent) that of when the partner is not HIV-positive (34.2 percent).

A similar phenomenon occurs with casual partners, where we see that when the partner is HIVpositive or an IDU, the proportion of women who have never received treatment is significantly less.

We did not find a significant association between prostitution and the dependent variable.

3.5 Reproductive habits and infections

Having regular menstrual cycles was strongly associated with having never received treatment. Those women reporting regular periods were three times more likely to have never received treatment. This can be interpreted that those IDUs who still have regular cycles are most likely in less advanced stages of drug addiction and have experienced little physical deterioration (it is well known that the use of certain drugs, especially heroin, can alter the menstrual cycle), such that they haven't yet considered it necessary to seek treatment.

A strong relationship was also found with having ever used a condom. Those who had never used a condom were five times more likely to have never been in treatment. Of the female IDUs who had ever used a condom, 16.7 percent had never been in treatment, compared to 51.1 percent of those who have never used a condom.

No association was found between having suffered from any type of STD and not having received drug treatment.

Having had hepatitis B and hepatitis C in the past year both presented as protective factors against the dependent variable, with an OR=0.5 and 0.2 respectively. This most likely reflects the fact that in many cases the centres are detecting the hepatitis through routine physical exams while those who haven't sought treatment are probably going undetected.

As far as the relationship between HIV serology and never having had treatment, no significant differences were found between HIV-positives and HIV-negatives, while these differences are evident between both groups and those who don't know their HIV serologic status.

4. Results of the Hierarchical Logistic Regression Analysis

Table 10.2 presents the odds ratios and confidence intervals of each of the variables entered into the regression model, and the new odds ratios as subsequent groups of variables are entered into the regression.

Age (Introduction of Model 1)

Using the oldest age group as a reference, the youngest age group (24 years or less) presents significant associations with having never had treatment. This association strengthens with the introduction of successive models.

Socioeconomic status (Introduction of Model 2)

The age of leaving full-time studies does not maintain a clear association with the dependent variable until the final model, where after adjusting the model for variables representing risky behaviours, the group who quit studying at age 13 or younger presents an OR=4.0 (CI=1.2-12.6) for never having received treatment, in comparison to the age group who quit school latest.

Not having a fixed address is also a factor directly associated with the dependent variable. This association continues to strengthen in subsequent models, resulting in an OR of 3.3 (CI=1.5-7.3) in the final model.

Living with children remained out of the equation indicating that it is not directly associated with never having received treatment.

Marginalisation variables (Introduction of Model 3)

Not having had a casual IDU partner presented, inexplicably, as a factor directly associated with the dependent variable. However, in the final model, after adjusting for the variables representing risk behaviours, this association disappears.

The most marginalised group of women, who had their first pregnancy at age 17 or younger, are, paradoxically, at less risk of having never received drug treatment (OR=0.3; CI=0.1-0.9).

Risk behaviour variables (Introduction of Model 4)

As we found in the bivariate analysis, the group of women with a regular HIV-positive sexual partner have a lower probability of having never received treatment in relation to the group that had not had a regular HIV-positive partner (OR=0.4;CI=0.1-1.0).

The age of beginning to inject drugs is also directly associated with the dependent variable. Those who began injecting later (over age 20) were three times more likely to have never received treatment than those who began injecting earliest (before age 17). Other risk behaviours, such as injecting heroin+cocaine and never having used a condom, also present a higher probability of having never received treatment (OR= 2.9 and 4.2 respectively).

Those women who don't know their HIV serostatus have a higher probability of having never received treatment than those who are HIV-positive (OR=2.9;CI=0.9-9.8).

5. Discussion

In this chapter we have examined the relationship between treatment and risk behaviours and we have attempted to identify the characteristics or factors directly and indirectly associated with having never attended treatment. Unfortunately, the limitations of the study design preclude us from making any strong conclusions, because we are unable to firmly establish the direction of the relationships we are examining. For example, in this type of study it is impossible to establish whether treatment has reduced risk behaviours or people with fewer risk behaviours have better access to treatment. So it is with these limitations in mind that we present the following findings.

In attempting to define the subpopulation which treatment programmes apparently don't reach, we found that this subgroup tended to be younger, less educated, homeless, young in their drugusing careers, and more likely to engage in recent needle sharing, less condom use and more injecting of heroin+cocaine. In other words, it would appear that the IDU women who are not being reached by drug treatment programmes are at very high risk for contracting HIV. Clearly, this is an urgent public health problem for Madrid that needs to be addressed.

Moreover, the characteristics identified in this study are not altogether different from characteristics identified in previous studies. For example, Ross et al. (1993) found that those never in treatment were younger and reported lower drug use and less involvement in the drug subculture. We also observed indirect relationships with having regular menstrual periods and not having a casual IDU partner, which might indicate lower drug use and less involvement in the drug subculture.

Baker et al. (1995) also found that those never enrolled in methadone treatment were younger (non-significant trend), started injecting later (non-significant trend), and were more likely to be homeless. Thus, on the one hand, those who have never received treatment appear to be those who are young in their drug-using careers, and their habit has not become severe enough to lead them to seek treatment. On the other hand, never having received treatment is also associated with a more marginalised lifestyle, suggesting that some of these women may not have access to treatment.

We also found that those who had never attended any treatment were more likely to have had their first pregnancy at a later age and to have not had a regular HIV-positive sexual partner. In other words, those who attended treatment were more likely to have had their first pregnancy at a young age, and the early pregnancy may have led them into treatment in an effort to protect their unborn child or to take control of their life for the sake of the child. Having had a regular HIV-positive sexual partner may indicate that these women are recognizing the risks to their health and are seeking drug treatment in attempt to save their lives. On the other hand, it could indicate

that an HIV-positive partner is concerned for his partner's health and more supportive of entry into drug treatment. Clearly male sexual partners play an important role in the drug treatment process, as has been demonstrated in previous studies (reviewed in Anglin et al., 1987).

The risk behaviours observed to have a significant relationship with treatment were: injecting heroin+cocaine; recent needle sharing; and having ever used a condom. Those who had been in treatment were less likely to report injecting heroin+cocaine in the previous six months. As the direction of this relationship cannot be determined, it is difficult to say whether drug treatment is reducing the tendency to inject heroin+cocaine or treatment programmes are simply less likely to reach those who inject heroin+cocaine. Of course, the latter needs to be considered, as injectors of heroin+cocaine have been found to engage in other HIV risk behaviours. Chitwood and Comerford (1990) found that injectors of cocaine used in combination with opiates were more likely to have multiple sex partners and high-risk injecting behaviour than cocaine-only or opiate-only injectors. Kowalewski (1990) reported that the disinhibiting effects of cocaine may predispose users to HIV infection through high-risk sexual behaviour. These findings are supported by the work of Joe & Simpson (1995) who reported that HIV risks increased as a direct function of cocaine usage level, though only for males. Among females, low-to-intermediate levels of cocaine use (less than weekly) were associated with high-risk behaviours.

Needle sharing in the previous six months was not a significant factor in the regression model, though many studies have reported reductions in needle sharing after exposure to treatment. This study may not have been able to detect a direct effect of treatment on needle sharing because of a pre-existing trend towards reducing needle sharing.

Unlike other studies, which have not found changes in sexual risk-taking, especially increases in condom use, we found that treatment was directly related to condom use. Although the study design precludes us from making assumptions about the directionality of this relationship, there is evidence from other studies that treatment programmes which implement specific sexual risk reduction interventions (such as skills training) can have success in reducing sexual risk for HIV infection among female entrants (Eldridge et al., 1997). In other words, drug treatment centres as venues for HIV risk reduction interventions can potentially have an important impact on the prevention of the spread of AIDS.

The relationship between HIV status and treatment is a bit difficult to interpret without knowing the direction of causality. In the bivariate analysis we observed a non-significant trend between being HIV-positive and ever receiving treatment. This could be explained by the fact that HIV-positives frequently begin drug treatment in the hope of improving their physical state and ability to combat the deterioration of their immune system. There was also a weaker, though positive relationship between treatment and being HIV-negative. This may indicate that many of those who have tested negative consider it a good time to quit using drugs, now while there is still time. This variable was later recoded to reflect "knowledge of HIV status" and a strong positive relationship was detected, suggesting either that knowledge of one's status prompts drug users to seek treatment or that drug treatment successfully encourages HIV testing. Of course, this relationship may also be reflecting an attitude of apathy in which the same individual who doesn't seek treatment also failed to provide a medical report on their HIV status.

Finally, although this study did not address gender differences in treatment, it seems apparent that women have an array of different needs that ought to be considered by drug treatment programmes. For example, studies have indicated that women in treatment report more medical, psychiatric, family-social, and employment problems than men (McLellan et al., 1992; Brown et al., 1993). More specifically, these needs and concerns include: reproductive health needs, such as STD detection and treatment and family planning; child care responsibilities; assertiveness training for negotiating safer sex and drug use with their partners; the involvement of male sexual partners in the treatment programme; and vocational skills training to help them re-integrate into society without having to rely on trading sex for money or financial assistance from drug-using partners.

6. Conclusions

From the simultaneous observation of the bivariate analysis and the multilevel logistic regression model we can extract the following conclusions:

- C IDU women who are not being reached by drug treatment programmes tend to be younger, less-educated, homeless, less involved in the drug subculture and engaging in more high risk behaviour, such as needle sharing, injecting heroin+cocaine, and not using condoms.
- C Exposure to treatment appears to be associated with more condom use, suggesting that drug treatment centres as venues for HIV risk reduction interventions can potentially have an important impact on the prevention of the spread of AIDS.
- **C** Drug treatment is reaching IDU women who get pregnant at a young age.
- C A woman's decision to seek treatment is influenced by her male partner.

Table 10.1.

Proportion of women reporting never had drug treatment by selected variables

Variable	Total	Had drug tr	eatment	OR(CL95%) p
		Never(%)	Ever(%)		
Sociodemographics					
Age					
24 or less	73	47.9	52.1	10.4(3.1-37.9)	***
25 - 29	110	. 18.2	81.8	2.5(0.7-9.2)	
30-34	72	11.1	88.9	1.4(0.4-6.0)	
35 or more	49	8.2	91.8	l(reference)	
Age leaving education					
13 or less	88	31.8	68.2	1.7(0.9-3.2)	n.s.ª
14-15	89	12.2	87.8	0.5(0.2-1.1)	
16 or more	132	22.0	78.0	l(reference)	
Abode					
No fixed address	69	33.3	66.7	2.2(1.2-4.0)	**
Fixed address	235	18.7	81.3		
Income					
Illegal+prostitution	137	23.4	76.6	0.9(0.5-1.5)	n.s.
Legal	167	21.0	79.0		
Ever been in prison					
Yes	98	17.3	82.7	0.7(0.4-1.2)	n.s.
No	206	24.3	75.7		
Drug-using habits					
Age began injecting drugs	-				
16 or less	81	14.8	85.2	l(reference)	* a
17 - 19	81	22.2	77.8	1.6(0.7-4.0)	
20 or more	141	26.2	73.8	2.1(1.0-4.5)	
Ever shared needles/syring	zes				
Yes	217	19.4	80.6	0.6(0.3-1.1)	n.s.
No	86	27.9	72.1		
Ever shared needles/syring	ges in last 6 moi	nths			
Yes	104	26.9	73.1	2.4(1.2-4.9)	**
No	114	13.2	86.8		

 Table 10.1 (continued)

 Proportion of women reporting never had drug treatment by selected variables

Variable	Total	Had drug treatment		OR(CI.95%)	
· · · · · · · · · · · · · · · · · · ·		Never(%)	Ever(%)		
Injected Heroin+Cocaine					
Yes	151	26.5	73.5	1.7(1.0-2.9)	*
No	153	17.6	82.4	· · · ·	
Ever used condom					
Yes	257	16.7	83.3	0.2(0.1-0.4)	**:
No	47	51.1	48.9		
Sexual habits					
Ever had a regular sexual	partner				
Yes	293	20.8	79.2	0.2(0.1-0.7)	*
No	11	54.5	45.5	· · · ·	
Ever had a regular sexual	HIV+ partner				
Yes	138	10.9	89.1	0.2(0.1-0.4)	***
No	152	34.2	65.8	· · ·	
Ever had a regular sexual	IDU partner				
Yes	257	21.0	79 .0	0.7(0.3-1.4)	n.s.
No	47	27.7	72.3		
Frequency of condom use Vaginal intercourse	with regular sex	cual partner			
Never/sometimes	122	28.7	71.3	6.6(1.5-29.2)	**
Always	35	5.7	94.3	0.0(1.0 2).2)	
Ever had a casual sexual p	artner				
Yes	241	20.7	79.3	0.7(0.4-1.3)	n.s.
No	62	27.4	72.6	、 /	
Ever had a casual sexual H	IIV+ partner				
Yes	138	10.9	89.1	0.2(0.1-0.4)	***
No	152	34.2	65.8		
Ever had a casual sexual I	DU partner				
Ever had a casual sexual D Yes	DU partner 156	14.7	85.3	0.4(0.2-0.7)	***

•

Table 10.1 (continued) Proportion of women reporting never had drug treatment by selected variables

.

Variable	Total	Had drug treatment		OR(CI.95%) p	
		Never(%)	Ever(%)		
Frequency of condom use	e with casual sex	cual partner			
Vaginal intercourse		•			
Never/sometimes	48	· 18.8	81.3	0.9(0.3-2.5)	n.s.
Always	50	20.0	80.0		
Ever had a client					
Yes	140	20.0	80.0	0.8(0.5-1.4)	n.s.
No	164	23.8	76.2		
Frequency of condom use Vaginal intercourse	e with client				
Never/sometimes	26	7.7	92.3	0.2(0.0-1.0)	n.s.
Always	52	28.8	71.2	0.2(0.0 1.0)	
Reproductive habits and	l infections				
Regular periods					
Yes	99	35.4	64.6	3.0(1.7-5.2)	***
No	205	15.6	84.4		
Age at first pregnancy			.		
< 18	74	18.9	81.1	1.3(0.6-2.6)	n.s.
18 or more	230	23.0	77.0		
Contraception currently					
Yes	159	26.4	73.6	1.7(1.0-3.0)	n.s.
No	140	17.1	82.9		
IUD used in last 5 years					
Yes	21	14.3	85.7	0.6(0.2-2.0)	n.s.
No	283	22.6	77.4		
Condom used in last 5 year					ماد مال وال
Yes	247	16.6	83.4	0.2(0.1-0.4)	***
No	57	45.6	54.4		

Variable	Total	Had drug treatment		OR(CI.95%) p	
		Never(%)	Ever(%)		<u>. P</u>
Reproductive habits and	infections (cor	· · ·			
Spermicide used in last 5 y		,			
Yes	45	31.1	68.9	1.8(0.9-3.5)	n.s.
No	259	· 20.5	79.5		
Contraceptive pill used in I	last 5 years				
Yes	91	19.8	80.2	0.8(0.5-1.5)	n.s.
No	213	23.0	77.0		
Ever been pregnant					
Yes	230	19.6	- 80.4	0.6(0.3-1.0)	n.s.
No	74	29.7	70.3		
Regular smear test					
Yes	134	23.1	76.9	1.1(0.6-1.9)	n.s.
No	168	21.4	78.6		
STD in last 5 years					
Yes	77	11.7	88.3	0.4(0.2-0.9)	*
No	196	25.0	75.0		
HIV					
Positive	165	13.9	86.1	l(reference)	***0
Negative	97	22.7	77.3	1.8(0.9-3.6)	
Unknown	42	52.4	47.6	6.8(3.0-15.4)	
Hepatitis A in last 5 years					
Yes	8	12.5	87.5	0.5(0.1-4.5)	n.s.
No	276	20.7	79.3		
Hepatitis B in last 5 years					
Yes	72	12.5	87.5	0.5(0.2-1.0)	*
No	212	23.1	76.9		
Hepatitis C in last 5 years					ala ala ala
Yes	89	5.6	94.4	0.2(0.1-0.4)	***
No	195	27.2	72.8		
Tuberculosis in last 5 years					
Yes	45	20.0	80.0	1.0(0.4-2.1)	n.s.
No	239	20.5 172	79.5		

Table 10.1 (cont) Proportion of women reporting never had drug treatment by selected variables

٠

C) for models B.C and D the following variables were excluded from the equation. Stage 2:Live with children: Stag 3: Previous imprissonment. Regular partner son Regular sexual partner IOU and prostitution: Stage 4: Ever sharing needles. Cocaine injecting. HIV positivity

2	Ð	æ	a
5	*	limi	=
5	Å	inat	ц С
2	.05	ed.	son
	*	4	R
2	*	i 1s	Irsi
	<u></u>	+	
5	01:	he	g
÷	**	new	, L
2	Å	ŝ	gigo
ī	ö.0	ijoe	tic
ŝ	b) * p<0.05; ** p<0.01; *** p<0.001.	ŝ	гe
) mic	gres
5		×a	sic
ĩ		ria	S V
Ş		ble	ere
5		S M	ਉ
Ś		re	rive
Į.		sel	đ
2		ect	or
1		2	eac
Ī		Th	ls u
		ĝ	age
-		pro	of
i -		Ces	H
		S	e ar
ו נ		ğ	ыly
		ickw	sis
i		ard	USI
-		eli	B
		Ĩ	SeS
		ati	S
c) For models R (and D the following warishing wore evaluated from the pointing (that differ with addition of		eliminated. whilst the new socioeconomic variables were selected trhough process of backward elimination.	a) The most parsimonious logistic regression were derived for each stage of the analysis using SPSS. For example
2			2
ן ג			ŝ
2			<u>"</u>
			힉
			S
Ì			ទ
•			2
			<u>s</u>
-			for Group 2 the Age va
,			≨ا
			ā
			le i
			SPA
			fix
			ëd.
			it
			xample, for Group 2 the Age variable was fixed, it could not be
			Ē
			not
			፳

	HIERARCHI	HIERARCHICAL LOGISTIC REGRESSION OF EVER HAD DRUG TREATMENT	ever had drug treatment on	on risk factors
	HODEL A AGE N=304	HODEL B SOCIOECONOMIC STATUS N-304	HODEL C MARGINALISATION N=304	HODEL D RISK BEHAVIOUR N=304
	OR (I.C. 95%)	OR (1.C. 95%)	OR (1.C. 95%)	OR (1.C.95%)
AGE group				
25-29 30-34 35+	0.2 (0.1-0.5)*** 0.1 (0.1-0.3)*** 0.1 (0.0-0.3)***	0.2 (0,1-0,4)*** 0,1 (0,0-0,3)*** 0,1 (0,0-0,2)***	0.3 (0,1-0,6)*** 0.2 (0,1-0,4) 0.0 (0.0-0.2)***	0.2 (0.1-0.4)*** 0.1 (0.0-0.4)*** 0.0 (0.0-0.2)***
SOCIDECONDMIC STATUS Age leaving education				
13 or 1ess 14 - 15 16 - 18		1,2 (0,5-2,9) 0,3 (0,1-0,8)* 1,4 (0,5-3,5)	1.9 (0.7-5.2) 0.3 (0.1-1.2) 1.8 (0.6-5.1)	3.7 (1.2-11.2)* 0.5 (0.1-1.6) 2.3 (0.8-6.5)
19 + Fixed Adress				
No			0.0 (1.177.4)	3.6 (1.3-0.0)
HARGINALISATION Casual partner IDU				
Yes Actual regular partner			0.4 (0.2-0.9)*	0,5 (0.3-1.1)
Yes .			2.2 (1.0-4.7)*	1.8 (0.8-3.7)
Age first pregnancy 17 or less 18 +			0.5 (0.2-1.1)	0.4 (0.2-0.9)*
RISK BEHAVIOUR Regular partner HIV				
Yes .		<u>.</u>		0.3 (0.1-0.7)**
Age injecting drugs 16 or less				2 1 2 2
17 - 19 20 +				1.3 (0.5-3.6) 3.4 (1.3-9.1)*
Injecting heroine+cocaine Yes				2.5 (1.2-5.2)
- 71	787 177	016 696	2726 EA7	1
Improvement Chi-Square	37.336***	20.008***	13.425**	27.226***
a) The most parsimonious logistic	regression were derived for	each stage of the	ysis using SPSS. For exam	analysis using SPSS. For example, for Group 2 the Age val

Chapter Eleven

Conclusion and Policy Recommendations

1. Introduction

This chapter is divided into four sections.

The first section will provide a summary of the major research findings which have been found in the study of determinants or predictors of HIV and HIV risk behaviour in female IDUs in Madrid. The study of HIV risk behaviour included: examination of the determinants of risky sexual behaviour (lack of condom use); and risky injection practices (needle and syringe sharing). Having experienced prison and exchanged sex for drugs or money will also be summarized in relation with other risks and sociodemographic characteristics. Finally, we will summarise the role of treatment.

The second section will consider the methodological issues regarding this research and their implications.

As a consequence of this study and based on the limitations of this research, the third section will present a number of issues worthy of further research and propose new methods which should be implemented.

The final section, perhaps the most beneficial to the women IDUs and consequently to society, will consider the implications for public health and briefly outline a policy agenda based on the more relevant findings of this work, some of which may call for urgent implementation.

2. Summary of Findings

We have studied the social determinants of the risks of HIV and other diseases among a sample of IDUs in Madrid. First, considerable support was found for the relationship between poor lifestyle and risk behaviour. Even though in many cases we didn't find strongly significant statistics due to the relationships among variables that have influenced the dependent variable both directly and indirectly, the general conclusion would be that this population is in a state of high risk for both health and psychosocial problems.

Overall, many of these women were poorly educated and unemployed, had jail or prison experience, were without a fixed address, and a considerable number had worked as street-workers. The majority of them live with an IDU partner and live in an unstable situation, with a multitude of risk factors for ill health, particularly for HIV and STDs. Thus our sample indicates. in comparison with other studies, that women who continue to inject drugs are a higher risk population, in general.

One of the most pervasive factors identified by this study was having no fixed address. It was directly related to prostitution, recent needle sharing, lack of condom use, no exposure to treatment, and even maintained a direct relationship with HIV prevalence. Furthermore, we found an indirect association with previous imprisonment. Previous research has found elevated levels of HIV risk practices in homeless populations, including IDU, crack-cocaine use, sexual relations with an IDU, exchange of sex for money or drugs, multiple sexual partners, and inconsistent condom use (Fisher et al., 1995; Nyamathi et al., 1997), but little information is available on the specific risks associated with homelessness in an IDU population. Clearly this subgroup of IDUs appears to be at higher risk for HIV while being more socially isolated, with less access to drug treatment. Certainly the interpretation of these findings must be qualified by the limitations imposed by the design of this study; however, they do have important implications for future research and policy (to be discussed in subsequent sections.)

On nearly all of the characteristics examined, even when all variables were controlled in the multivariate analysis, findings indicated that first pregnancy at a younger age was a relevant predictor for risk behaviour in IDU women. There was a relationship to prostitution and sharing needles; ever being pregnant was correlated positively with previous imprisonment and negatively with high-risk sexual activity. First pregnancy at a younger age is also positively related to treatment. The IDU women that experienced their first pregnancy before age 18 may also have a lower socioeconomic status and need money to support themselves as well as their children, which may lead them into a life of prostitution or crime just to survive. Earlier studies have linked both early age of first pregnancy and prostitution to childhood sexual abuse (Zierler et al, 1992). Furthermore, early sexual experience has been identified as a factor in prostitution (James and Meyerding, 1977), and age at first coitus as a marker for risky sexual behaviour (Greenberg, Magder and Aral, 1992). The implications of these findings are that early sexual experiences, including childhood sexual abuse, can be used as markers for high-risk sexual behaviour and interventions aimed at delaying a woman's first sexual experience may lower the likelihood of later engaging in prostitution and other risky sexual practices. We did not look at the relationship with rape or sexual assault in this study, but many authors have considered this issue.

More than half (53.4 percent) started injecting drugs before the age of 20. This, without a doubt, will have an important impact on their psychological development and social behaviour as it is in this stage of life when the basis of their adult personality is established. Therefore, in these women, future behavioural problems are found with greater frequency than in those who started injecting drugs later in life.

The preceding chapters identified a number of risk factors associated with HIV infection, including the observation that female IDUs were at greatest risk of HIV infection through sexual partners who were either drug addicts and/or HIV-positive. The partner also has a great influence in the introduction of drug use to the woman. Almost half of the women questioned (49 percent) injected drugs for the first time with their regular partner. Having a steady HIV-positive partner also played a significant role in seeking treatment. This demonstrates the great influence a regular male partner has over a woman, and this influence must be taken into account when developing strategies against drug addiction. Perhaps one of the most effective measures to prevent drug addiction in women would be to persuade the drug-using partner to discourage the women from taking drugs and to inform him of the enormous influence he can have in this regard and the consequent responsibility. Furthermore, it is imperative for treatment programmes to involve the regular partner in the treatment process.

When we studied the HIV risk behaviour of sharing injection equipment, we found that those women who reported sharing habitually with partners were clearly taking high risks, just because of the frequency with which they used unsterile needles and syringes, particularly when we know that a great proportion of these partners are HIV-positive. Most studies indicate that women exchange needles more than men when both partners are at risk of HIV transmission and when needles and syringes are shared between them. However, given that the majority of women are in relationships with injectors whereas the same is not true among men, one would have to conclude that in this respect women are at higher risk of HIV.

Unlike men, very few women (1 percent) participate in drug-taking behaviour alone. This implies that women, more than men, need to feel supported by someone in making decisions of this kind, involving the transgression of a behavioural norm. This is another indication of the great difference between men and women in the drug-injecting experience. Women are seen to be more affected by prejudice than men and thus need to have someone reaffirm their behavior and push them into action. The explanation for this can be found in the roots of a society which establishes different roles according to one's sex, producing gender differences.

Women most commonly shared needles with a regular sexual partner (47.5 percent), whilst 17.8 percent shared with a male friend and 10.9 percent with a female friend. The main source of injecting equipment for women was the pharmacy or chemist, although 56.6 percent of women reported the use of needle-exchange schemes. This observation, however, probably reflects the sample characteristics of recruitment and thus any comparison with other (associated or independent) studies would be meaningless.

This leads to relevant observation that the risk for HIV among female heroin addicts is influenced by their relationship with male sexual partners. Women who were living with a drug user were less likely to use condoms. Furthermore, having an HIV-positive partner was one of the strongest determinants of HIV for the population studied, despite the fact that we found an interaction indicating that women who know that their partner is HIV-positive use more condoms than other IDU women, and this issue will be considered later.

It should also be noted that in the multivariate analysis, having shared a needle in the past six months was negatively associated with HIV prevalence, regardless of the type of partner, which may imply that those women who knew they were HIV-positive had stopped sharing needles, although such a hypothesis cannot be confirmed by these data.

When we studied the predictive factors for continuing to share needles, we found that this residual risk group was quite marginalised, as indicated by: not having a fixed address; younger age of first injecting drugs; younger age at first pregnancy; prison; and prostitution. Thus it appears that female heroin addicts who continue to engage in risky injection practices are embedded in a high-risk lifestyle.

Sexual behaviour in this population reflects the net effect of almost all of the variables. As unequivocally demonstrated (Table 5.3, Chapter 5), having a partner who is HIV-positive increases the likelihood of HIV infection. Given that nearly 50 percent of this population of female IDUs reported having a regular partner who is HIV-positive, the importance of this factor cannot be underestimated. Clearly the most important conclusion to be drawn from this study is that the sexual partners (and therefore sexual behaviour) of female IDUs are as important a

component in explaining the HIV epidemic in this population as other risk factors, including high-risk drug-taking behaviour.

As it was noted by Barnard (1992), in the sphere of sex and the negotiation of sexual encounters, with regard to heterosexual relationships, expectations of behaviour appropriate to a woman place the woman in a more disadvantaged position. First of all, both men and women are socialised into a culture which on the whole encounters difficulties in the discussion of sexual matters. This situation is further exacerbated by expectations of male dominance and female modesty. In sexual issues, women are not expected to take the lead over men. Given an existing power balance in favour of men, it may be extremely difficult for a woman to assert her desire that a condom be used. Thus in general, it is quite unusual for women to carry condoms even when they are working as a prostitute. The low rate of condom use in sexual relations with long-term partners raises a different set of issues, including the idea that condom use decreases emotional intimacy and physical closeness, and the social construction of the condom as a barrier in physical terms. Another issue is that requesting the use of a condom may imply mistrust in the relationship. Finally, as Day et al. (1993) discovered, many female sex workers find that they need to make clear the distinction between personal and commercial sex, causing them to shun the use of condoms in private relationships.

In summary, the female IDUs in this study are particularly vulnerable to HIV infection because of their higher level of risk practices as well as their sexual behaviour with regular sexual partners. As it has been shown in the outcome of regression, the strongest determinant of HIV infection in women is to have an HIV-positive sexual partner.

When we look at the social determinants of sexual risk behaviour among female IDUs, studying in particular never using condoms, we found that, once again, the most important contributors to the model were the social marginalisation factors. In fact, the residual risk group who has never used condoms tends not to have a fixed address, began injecting drugs at an early age, and were recruited from the street. In other words, this is a subgroup which may have less access to HIV prevention information as well as less access to condoms themselves. Those who began injecting drugs earlier are the most marginalised and have risk behaviours more evident at every level. Another important determinant of never using condoms was not practicing prostitution. The use of condoms seems to be a well-diffused behaviour among prostitutes in our study, at least in their relations with clients. Other studies have also noted that condom use tends to be more frequent with clients and casual partners than with regular partners (Rhodes et al., 1994; Paone et.al, 1995).

We also detected a link between sexual risk behaviour and drug-taking risk behaviour. There are significant associations between not always using condoms with casual partners, regular partners, or clients, and having shared needles in the last six months. In fact, the strong relationship between needle sharing and lack of condom use with casual partners suggests not only a particularly high-risk subgroup, but also a potential bridge for HIV transmission to the general population.

Among sociodemographic factors, previous imprisonment was a relevant determinant of HIV status. Although it didn't emerge as a significant direct determinant in the regression analysis, prison experience did show a strong relationship with HIV in the bivariate analysis and has been identified consistently as a direct determinant in earlier studies. It is evident that in the current study imprisonment is strongly associated with other factors, such as having an HIV-positive partner and history of an STD, which were significant determinants of HIV in the model. In other words, prison is an indirect determinant of HIV and is an important marker for a high-risk

population due to both their behaviour in and out of prison as well as their living situation out of prison.

In general, the environmental and socioeconomic conditions of women who engage in drug use may result in a variety of interrelated risk factors for HIV infection which result in a greater risk than the sum of the risks for each individual factor. In addition, such women usually come from socially and economically deprived backgrounds and may suffer from generally poor health due to inadequate nutrition caused by drug addiction, lower access to health services, frequent abortions, use of various types of drugs, high stress levels and exposure to violence, to name but a few. Their health may be further compromised by other social situations, including prostitution and prison.

Moreover, the relationship between drug users and prostitution is of great interest as prostitutes constitute a possible source for the heterosexual transmission of HIV from female IDUs to the general population. These data, however, did not clarify the relationship between prostitutes and drug use and, if anything, indicate that IDU prostitutes do not act as a bridge to the general population given the high rate of condom use. However, there was a significant relationship between prostitution and HIV, albeit indirect. Clearly the IDUs in this study who engaged in trading sex for drugs were more "down and out" than their counterparts who had never had sex with clients. They were more likely to report no fixed address, to have been incarcerated, to have been younger at their first pregnancy, and to lack social support.

Likewise, we have also confirmed what is already well documented: the connection between syphilis and other sexually transmitted diseases and the prevalence of HIV. This relationship is most likely a consequence of risky sexual behaviour, although there is also evidence that physical damage caused by sexually transmitted diseases, such as lesions, promote the transmission of HIV. It is also interesting to note that in the current study the rate of sexually transmitted disease was extremely high, probably as a result of poverty, social disintegration, and kind of partner, but the rate of condom use was also quite low.

3. Methodological Issues

The objective of this section is to examine the limitations of the study and to discuss the development of a new model explaining risk behaviour amongst female IDUs. The foremost conclusion of the study was that there was no apparent association between HIV status and some risk behaviours. It is likely that this is a consequence of the study's cross-sectional design, as HIV infection may have occurred before the "study period". For example, the implementation of campaigns in Spain in 1992, including testing for HIV, would have meant that women who were measured in 1995 as being HIV-positive had past risk behaviours which were changed as a result of public health intervention before and during the course of the study.

Second, the study is subject to both selection and measurement biases due to the atypical characteristics of the population. Whilst continued high-risk behaviour of the population makes it important that the results are not generalised to other groups of IDUs, it is nevertheless important to study this residual group of women as an understanding of their behaviour could result in significant changes in public health policy. Consequently, to reduce biases in the study, it was considered important to use a multi-site / multi-sampling design. This would mean that the effects of any outliers would be reduced, and that the combined data would provide insights into the behaviour of female IDUs.

Third, regression analysis in this group is limited by several factors. First, as noted above, the limitations of a cross-sectional design may result in spurious findings. This is likely in this population as behaviour has been changing in recent years. Moreover, the population is not isolated, and its characteristics are determined by a range of other factors, all of which may themselves influence behaviour. Furthermore, the correlational nature of multiple regression does not allow us to make causal inferences on the relationships between background factors and risk behaviour or dependent variables.

Given the above, although initially we analysed the model using a simple parsimonious logistic regression, it became clear that this was unsuitable in these circumstances. The objective of a parsimonious model is to identify the most *efficient* explanatory variables, which will fail to quantify socially important pathways. This is illustrated by X and Y who demonstrate that regression analysis has two objectives. The first is to be predictive, i.e., to know the best way to determine the effect of independent variables on the outcome. The second is to be estimative, i.e., to identify the relationship between the independent and dependent variables. Our aim was to use a predictive model, therefore we used a hierarchical stepwise regression because we wish to understand the relationship of <u>all</u> variables. In this model, we can look at the complexity of many variables, and examine whether they have a direct or indirect effect on the dependent variable, which would not be possible using a parsimonious model. In fact, the change in our thinking during the course of the study demonstrates the importance of using a pragmatic and flexible approach when examining outcomes in highly marginalised populations.

Finally, and as an illustration of the above, following the discovery that most of the women had changed their behaviour, we decided to introduce new research questions to identify the reasons why some had not changed their behaviour. For example, we have studied the unsafe behaviours (e.g., never use condoms). Likewise, after showing that a large number of women had previously been in prison and/or were prostitutes, we also looked at these outcomes as dependent variables.

In summary, this study has presented a methodological challenge in planning, implementing and analysing epidemiological studies of marginalised populations. In such circumstances it is important not to restrict our mind-set to purely statistical power, and be willing to explore the social context of the data that are being examined. This would include developing an understanding of the lifestyle of female IDUs and the networks within which they live.

4. Future Research Implications

As discussed above, there are major difficulties in analysing data resulting from cross-sectional studies. Indeed the particular characteristics of IDUs are further complicated by recent changes in behaviour, thereby suggesting that this design is now redundant. Furthermore, this is confounded by the problems of identifying an appropriate model that adequately describes the lifestyle of different populations of IDUs, prostitutes, prisoners, etc. It is therefore important that further methodological advances be made. For example, the use of path analysis (using LISREL) may be of help. Similarly there is a need to develop multi-dimensional models which accommodate the subtlety and complexity of the data. It is important that such models are able to take into account temporal changes in behaviour, etc.

We have found in this study that the real risk for IDU women is when their partner is an IDU. It therefore makes sense that future studies focus on the relationship between a male IDU and his

partner. For example, non-IDU female partners of male IDUs have the risk of becoming an IDU and therefore would make an excellent population to target.

Future studies should also seek to explain and further explore the relationships identified in this study between not having a fixed address, risk behaviours, and HIV prevalence in an IDU population. IDUs who are also homeless may prove to be an important target population for future HIV prevention programmes. Furthermore, researchers should also study the relationship between psychological problems and HIV risk behaviour in this population.

As we have demonstrated that imprisonment is an important predictive variable for a number of outcomes, it is important that further research be implemented to examine HIV prevalence and IDU in female prison populations. Likewise, similar studies should look at the relationship between IDU and prostitution.

Further research should examine other health issues in female IDUs, such as their nutritional status, health habits, experiences with regard to births and abortions and gynaecological health. More knowledge is needed in these areas so that appropriate strategies can be designed to improve the health status of women.

Despite extensive research into IDU, there have been very few attempts to study in an integrated way the different life events of IDUs which directly affect their behaviour. The particular nature of these studies makes it hard to assess the characteristics of the circumstances of IDU and how one interferes with the other.

There is also a need to study the relationship between sexuality and power in decision making. Such research could provide valuable insights as to how women might gain a more balanced position when it comes to making decisions about sex. Qualitative research could also be useful to discover the motives and reasons for the failure of many women to use condoms in their sexual relations even with partners they know to be HIV-positive.

More research is needed regarding the relative efficacy of different models of outreach and intervention with IDU women and the non-IDU partners of IDU men.

Additional research is required to assess the extent to which risk behaviour among identified groups of women with IDU partners or ex-addicted partners vary over time and to identify those factors affecting change.

We need more research on developing and testing protective methods that are controlled by women, including the recently introduced female condoms, spermicides, etc., and further research is needed in this area to examine the role of spermicides with petroleum in facilitating HIV transmission especially among prostitutes, despite the fact that we have not studied this issue in the current study.

The paucity of studies in gender-differences in treatment and the recent conclusions by Hodgins et al (1997) about the efficacy of single-gender treatment groups indicate the urgent need to conduct research on gender differences in treatment process as well as outcome.

More research from prospective studies could provide useful information on the nature of the relationship between, for example, pregnancy and HIV infection. Furthermore, more studies are needed to identify appropriate methods of family planning and HIV prevention for female IDUs.

Finally, it would be worthwhile to study the relationship between teenage pregnancies and socioeconomic stressors as they combine to create a higher risk environment in which risk behaviours are more readily engaged in.

5. Public Health Implications and Policy Agenda

Our study has implications both for the prevention and control of the spreading of AIDS as well as in the area of prevention of drug use and harm-reduction programmes for IDU women.

We have examined the contexts within which risk behaviour took place and the determinants of each different situation, and we have found relevant factors. First, we found that age was a critical factor in several contexts. In particular, the younger women (adolescents) were more likely to engage in risky behaviour (they continued to share needles and were more likely to have never used condoms), and they were less likely to have received any treatment. Early age of first injection (less than 16) was also a direct determinant of prison and never using condoms. Early age of first pregnancy (less than 18) was a direct determinant of prostitution and needle sharing. Thus adolescence appears to be a time of high-risk behaviour, and if certain high-risk behaviours can be delayed or avoided, women might lower their risk of future problems. Clearly intervention with this age group of female IDUs is imperative. Adolescent females, particularly those with a low level of education are in need of specific strategies to prevent drug use and Ideally these strategies should be implemented in schools as treatment-based pregnancy. strategies don't appear to reach this sub-group. Furthermore, school-based strategies need to be targeted to young teenagers, before they begin engaging in risky behaviour and before they leave school.

Second, the sexual partners appear to exert an important influence on the risk behaviours of IDU women. Almost half of the women questioned (49 percent) injected drugs for the first time with their regular partner. And, unlike men, very few women (1 percent) participate in drug- taking behaviour alone. Women most commonly shared needles with a regular sexual partner (47.5 percent), whilst 17.8 percent shared with a male friend, and 10.9 percent with a female friend.

Hence future health education should be targeted in three directions:

1) toward the sexual partners of IDU women to increase awareness of risks of HIV transmission as well as their potential influence in reducing risk behaviour of their partner;

2) toward the IDU woman herself to improve her assertiveness and safe sex negotiation skills; and

3) toward IDU couples to improve awareness about the risk of HIV transmission via heterosexual relations.

Third, the IDU women who claimed to have no fixed address may prove to be an important target group for future HIV prevention programmes. This subgroup appeared to be less likely to be reached by drug treatment programmes and exhibited a number of high-risk practices as well as a higher prevalence of HIV. Clearly, public health policy makers in Spain need to focus on finding ways to access this hard-to-reach and high-risk subgroup. Furthermore, the planners of HIV prevention programmes for this population need to take into account the psychological problems which may be prevalent in this population and which can act as barriers to adopting safe behaviours.

Another important target group for HIV prevention education and harm-reduction programmes is prison populations. As was observed in this study, injecting drug use does occur in Spanish prisons. Administrators need to acknowledge the high-risk behaviour which occurs and initiate programmes to both educate prisoners about the risks and to reduce the spread of HIV within prisons.

Another point is that the message about risk in the area of drug-taking shouldn't be focused only on heroin. As it is carried out in Spain, AIDS prevention and drug-use prevention have to begin by educating IDUs about the additional risks associated with drugs other than heroin, such as amphetamine, cocaine, etc., because heroin use is decreasing and its social image is diminishing, so an attempt should be made to prevent other epidemics of injected drugs even if they are legal.

The gender-distinct nature of many of the social determinants of risk behaviour is also reflected in sharing patterns, sexual behaviour, and gender-related attitudes. This suggests that strategies of public health should be targeted according to women's experiences of a drug-injecting lifestyle and, on a related note, their particular risks of HIV transmission. So it is necessary to: increase and improve drug-abuse treatment programmes, especially with regard to single-gender programmes; make sterile syringes and needles freely available to those who need them, together with instructions on how to clean them; and promote the use of condoms. This should include interventions in prisons where many high-risk individuals can be reached and where high-risk activities occur.

Moreover, several types of female condoms have been developed, but are not yet widely available due to a high cost; these should be made freely available as quickly as possible so that women have another option for barrier protection which they can control.

So we must recognize that the more immediate goal of HIV prevention may not be realized without addressing the broader social issues affecting these women's lives. More intervention programmes need to be carried out among female IDU prostitutes with specific reference to the use of spermicides.

Given the importance of social services to IDU and HIV-positive women, efforts should be made in improving their provision to this population. An effort must be made to try to do away with the administrative barriers and constraints. These efforts should include peer education, since health education programmes are more likely to be successful if the person delivering the intervention is someone whom the target population can identify with.

Based on the findings of this study as well as those of earlier studies, women have a range of special needs, such as reproductive health, and require more resources than do men. Unfortunately, as IDUs and therefore drug-treatment recipients have been predominately men, women's needs have gone unnoticed until now. Treatment programmes usually fall short in providing adequate resources for issues of particular concern to women, such as: meeting female gynecological and prenatal health needs; providing assistance in finding child care facilities; offering appropriate methods in family planning; offering objective counselling on options for HIV-positive pregnant IDUs--providing the most up-to-date information on possible risks for the mother and child, but always respecting the right of each woman to make the final decision; and eliminating sex stereotyping in career counseling which is detrimental to the rehabilitation process for women.

In the area of TB, hepatitis, and other STDs, clinical strategies for their control should be implemented more strongly. That is, if possible, all diagnosed hepatitis and TB patients should be monitored and studied to examine all risks associated with these diseases including the particular needs of IDUs with regard to treatment and the implementation of public health initiatives to control the spread of hepatitis and tuberculosis amongst this marginalised community of female IDUs. This should include health education, screening and vaccination programmes.

BIBLIOGRAPHY

•••

.

ALCABES, P., VLAHOV, D., & ANTHONY, J.C. (1992). Correlates of human immunodeficiency virus infection in intravenous drug users: are treatment-program samples misleading? <u>Brit J Addiction</u> 87:47-54.

ALEGRÍA, M., VERA, M., FREEMAN Jr., D.H., ROBLES, R. et al. (1994). HIV infection, risk behaviors, and depressive symptoms among Puerto Rican sex workers. <u>Am J Public Health</u> 84:2000-2002.

ANDERSON, D., HILL, J. (1991). Cellular and soluble factors in semen and the vaginal environment that may influence the heterosexual transmission of HIV type 1. Horowitz, B., Mardh, P., eds. <u>Vaginitis and vaginosis</u>. New York: Wiley-Liss, pp. 69-76.

ANGLIN, M.D., HSER, Y., BOOTH, M.W. (1987). Sex Differences in Addict Careers. 4. Treatment. <u>Am</u> JDrug Alcohol Abuse 13(3):253-280.

ARCHIBALD. C.P., OFNER, M., STRATHDEE, S.A., PATRICK, D.M. et al. (1998). Factors associated with frequent needle exchange program attendance in injection drug users in Vancouver, Canada. J Acquir Immune Defic Syndr & Hum Retrovirol 17:160-166.

ASTEMBORSKI, J., VLAHOV, D., WARREN, D. et al. (1994). The trading of sex for drugs or money and HIV seropositivity among female intravenous drug users. <u>Am J Public Health</u> (March) 84(3):382-387.

BAKER, A., KOCHAN, N., DIXON J., WODAK, A. & HEATHER, N. (1995). HIV risk-taking behaviour among injecting drug users currently, previously and never enrolled in methadone treatment. <u>Addiction</u> 90, 545-554.

BARNARD, M.A. <u>Gender Differences in HIV-Related Risk Behaviour among a Sample of Glascow</u> <u>Drug Injectors</u>. Doctoral Thesis: Public Health Research Unit, University of Glasgow, Scotland, 1992.

BATH, G.E., DAVIES, A.G., DOMINY, N.J. et al. (1993). Imprisonment and HIV prevalence. <u>The Lancet</u> 342(27 November): 1368.

BATTJES, R.J., LEUKFELD, C.G., & PICKENS, R.W. (1992). Age at first injection and HIV risk among intravenous drug users. <u>Am J Drug Alcohol Abuse</u> 18(3):263-273

BAI, J., GREENWALD, E., CATERINI, H. & KAMINETZKY, H. (1989). Drug related menstrual aberrations. <u>Obstetrics and Gynaecology</u> 44: 713-9.

BECK, E.J., MANDALIA, S., LEONARD, K., GRIFFITH, R.J. et al. (1996). Case-control study of sexually transmitted diseases as co-factors for HIV-1 transmission. Int J STD AIDS 7(1):34-38.
 BERREY, M.M. & SHEA, T. (1997). Oral sex and HIV transmission. J Acquir Immune Defic Syndr & Hum Retrovirol 14:475-477.

BIRD, A.G., GORE, S.M., BURNS, S.M. & DUGGIE, J.G. (1992a). Study of infection with HIV and related risk factors in young offenders' institute. <u>BMJ</u>, 307.

BIRD, A.G., GORE, S.M., JOLLIFFE, D.W. & BURNS, S.M. (1992b). Anonymous HIV surveillance in Saughton Prison, Edinburgh. <u>AIDS</u>, 6, 725-733.

BRAVO PORTELA, M.J., BARRIO ANTA, G., DE LA FUENTE DE HOZ, L., COLOMO GÓMEZ, C. et al. (1996). Risk behaviors in HIV transmission among recent clients of a syringe exchange program in Madrid, 1993. <u>Gac Sanit</u> (November-December) 10(57):261-273.

BRAVO, J.M., DELGADO-RODRÍGUEZ, M., DE LA FUENTE, L., LARDELLI, P. et al. (1994). Sharing injecting equipment and sexual behaviour in ambulatory intravenous drug users: A national survey (Spain). Int J Addictions 29(4):1893-1907.

BROWN, B.S. (1989). Presentation of NADR national data. In: <u>Proceedings of the conference on AIDS</u> intervention strategies for female sexual partners. Berkeley, California, June.

BROWN, L.S., ALTERMAN, A.I., RUTHERFORD, M.J., CACCIOLA, J.S. & ZABALLERO, A.R. (1993). Addiction severity index scores of four racial/ethnic and gender groups of methadone maintenance patients. J Subst Abuse 5:269-279.

CAIN, V.S. & WILLIAMSON, N.E. (1994). Issues of acceptability and behaviour working group report. In: Barrier contraceptives and current status and future prospects. <u>Proceedings of the 4th contraceptive</u> <u>research and development program international workshop</u> (Edited by Mauck et al.). New York: Wiley Liss, p. 323-337.

CARLIN, E.M. & BOAG, F.C. (1995). Women, contraception and STDs including HIV. Int J STD AIDS 6:373-86.

CASTILLA, J., POLLÁN, M., & LÓPEZ-ABENTE. (1997). The AIDS epidemic among Spanish drug users: a birth cohort-associated phenomenon. <u>Am J Public Health</u> 87:770-774.

CAYLÁ, J.A., MARCO, A., BEDOYA, A., GUERRERO. R. et al. (1995). Differential characteristics of AIDS patients with a history of imprisonment. Int J Epidemiol 24(6):1188-1196.

CENTRO NACIONAL DE EPIDEMIOLOGIA. REGISTRO NACIONAL DEL SIDA. (1997) <u>Vigilancia</u> <u>del SIDA en España. Informe trimestral No. 4, 1997</u>. Ministerio de Sanidad y Consumo, Instituto de Salud "Carlos III", Subdirección de Salud, Madrid.

CHIN, J., SATO, P.A., MANN, J.M. (1990). Projections of HIV infections and AIDS cases to the year 2000. Bull WHO 68:1-11.

CHIRGWIN, K.D., FELDMAN, J., MUNEYYIRCI-DELALE, O., LANDESMAN, S. et al. (1996). Menstrual function in human immunodeficiency virus-infected women without acquired immunodeficiency syndrome. J Acquir Immune Defic Syndr Hum Retrovirol, Aug 15; 12(5):489-94.

CHITWOOD, D.D. & COMERFORD, M. (1990). Drugs, sex and AIDS risk. Am Behav Sci 33:465-477.

CLARK, R.A., KISSINGER, P., BEDIMO, A.L., DUNN, P. & ALBERTIN, H. (1997). Determination of factors associated with condom use among women infected with human immunodeficiency virus. Int J STD & AIDS 8:229-233.

COHEN, J., HAUER, L., WOFSY, C. (1989). Women and IV Drugs: Parenteral and Heterosexual Transmission of Human Immunodeficiency Virus. <u>J Drug Issues</u> 19,1: 39-56.

COLOMO, C., ESTÉBANEZ, P., ALBERDI, J., BRU. F. et al. (1990). Spread of infection among prostitutes of Madrid. Sixth International Conference on AIDS. San Francisco, June 1990 [abstract 3144].

COMMITTEE ON AIDS RESEARCH AND THE BEHAVIOURAL, SOCIAL AND STATISTICAL SCIENCES (1990) Interventions for female prostitutes. In: Miller, H.G., Turner, C.F., Moses, L.E., eds. AIDS: The second decade. Washington, D.C.: National Academy Press: 253-288.

CROFTS, N., STEWART, T., HEARNE, P. et al. (1995). Spread of bloodbourne viruses among Australian prison entrants. <u>BMJ</u> 310:285-288.

DARBYSHIRE, J.H. (1990). Tuberculosis en las prisiones. Posibles vínculos de unión con la infección por VIH. BMJ 5: 22.

DAWE, S., GERADA, C. & STRANG, J. (1992). Establishment of a liaison service for pregnant opiatedependent women. <u>Brit J Addiction</u> 87:886-871.

DAY, S., WARD, H., & PERROT, A. (1993). Prostitution and risk of HIV: male partners of female prostitutes. BMJ 307:359-361.

DE LA FUENTE, L., BARRIO, G., VINCENTE, J., BRAVO, M.J. & LARDELLI, P. (1994). Intravenous administration among heroin users having treatment in Spain. Int J Epidemiol 23(4): 805-811.

DELGADO-RODRIGUEZ, M., DE LA FUENTE. L., BRAVO, M.J., LARDELLI, P. & BARRIO, G. (1994). IV drug users: changes in risk behaviour according to HIV status in a national survey in Spain. J Epidemiol and Comm Health 48:459-463

DEMOGRAFIA Y SALUD (September 1996) Indicadores demográficos.

DESENCLOS, J.C., PAPAEVANGELOU, G., ANCELLE-PARK et al. (1993). Knowledge of HIV serostatus and preventive behaviour among European injecting drug users. <u>AIDS</u> 7:1371-1377.

DES JARLAIS, D.C. (1992). The first and second decades of AIDS among injecting drug users. Brit J Addictions 87:374-353. DÍAZ DE TUESTA, M.J. (1997). Seis de cada diez trabajadores de 25 a 30 años viven todavía en la casa de sus padres. <u>El País</u> (28 April) p. 29.

DOERR, H.W., ENZENSBERGER, R., BOLENDER, C., VAN LAERE-FISCHER, S. & PETERS, M. (1990). Prevalence of HIV infection in prostitutes from Frankfurt, West Germany. <u>Sixth International</u> <u>Conference on AIDS</u>, San Francisco, June 1990 [abstract FC 626].

DOLAN, K.A., DONOGHOE, M.C., JONES, S., STIMSON, G.V. (1991). <u>A Cohort Study of Syringe-Exchange Clients and Other Drug Injectors in England, 1989/1990</u>. Monitoring Research Group, Centre for Research on Drugs and Health Behaviour, Charing Cross and Westminster Medical School: London.

DOLAN, K., DONOGHOE, M., STIMSON, G. (1990). Drug injecting and syringe sharing in custody and in the community: an exploratory survey of HIV risk behaviour. <u>The Howard Journal</u> 29(3):177-186.

DONOGHOE, M.C. (1992). Sex, HIV and the injecting drug user. Brit J Addiction 87:405-416.

DONOGHOE, M.C., DOLAN, K. & STIMSON, G.V. (1991). The 1989-1990 National Syringe-Exchange Monitoring Study: syringe exchange schemes in England: service delivery and organisation; client characteristics and HIV risk behaviour. Centre for Research on Drugs and Health Behaviour, Charing Cross and Westminster Medical School: London.

DONOGHOE, M., STIMSON, G., & DOLAN, K. (1989). Sexual behaviour of injecting drug users and associated risks of HIV infection for non-injecting sexual partners. <u>AIDS Care</u> 1(1):51-58.

EGEA J.M., TOR, J., MUGA, R., ROCA, J. et al. (1996). Rates of human immunodeficiency virus (HIV) infection in intravenous drug addicts from the Barcelona area, according to sex and age of drug consumption onset. <u>Med Clín (Barc)</u> 106(3):87-90.

EL-BASSEL, N., SCHILLING, R.F., IRWIN, K.L., FARUQUE, S. et al. (1997). Sex trading and psychological distress among women recruited from the streets of Harlem. <u>Am J Public Health</u> 87(1):66-70.

ELDRIDGE, G.D., ST. LAWRENCE, J.S., LITTLE, C.E., SHELBY, M.C. et al. (1997). Evaluation of an HIV risk reduction intervention for women entering inpatient substance abuse treatment. <u>AIDS Education</u> and Prevention 9 (supplement A): 62-76.

ERICKSON, P. I., BASTANI, R., MAXWELL, A.E., MARCUS, A.C. et al. (1995). Prevalence of anal sex among heterosexuals in California and its relationship to other AIDS risk behaviors. <u>AIDS Education</u> and Prevention 7(6): 477-493.

ESTÉBANEZ, P. Estudio de los factores de riesgo para salud y para las ETS particularmente el SIDA en el trabajo sexual femenino en España. Universidad Autonoma de Madrid, 1995.

ESTÉBANEZ, P., COLOMO, C., ZUNZUNEGUI, M.V. et al. (1990). Cárceles y SIDA. Factores de riesgo de infección por el VIH en las cárceles de Madrid. <u>Gac Sanit</u> (May-June) 4(18):100-105.

ESTÉBANEZ, P., MARTÍNEZ, L., AGUILAR, D. & GRANT, J. (1996). <u>Risk behaviour in female</u> injecting drug users in France, Spain, Italy, England and Germany. London School of Hygiene and Tropical Medicine/European Commission.

ESTÉBANEZ, P., NÁJERA, R., RUA-FIGUEROA, M. et al. (1991). Prevalence HIV, HTLV1, HIV2, syphilis, hepatitis B in female sex-workers. <u>Seventh International Conference on AIDS</u>, Florence [Abstract W.C 3131].

ESTÉBANEZ P., RUA-FIGUEROA AGUILAR, D. et al. The prevalence of HIV and associated risk factors amongst Spanish sex workers. Submitted for publication, 1995.

ESTÉBANEZ, P., SARASQUETA, C., FITCH, K. et al. (1992). Prevalencia del VIH-1 y otras enfermedades de transmisión sexual en prostitutas españolas. <u>Med Clín (Barc)</u> 99(5):161-167.

EUROPEAN CENTRE FOR THE EPIDEMIOLOGICAL MONITORING OF AIDS. <u>AIDS surveillance</u> in Europe. Quarterly report No. 35, 30th September 1992.

EUROPEAN CENTRE FOR THE EPIDEMIOLOGICAL MONITORING OF AIDS. <u>AIDS surveillance</u> in Europe. Quarterly report No. 38, 30th June 1993.

EUROPEAN CENTRE FOR THE EPIDEMIOLOGICAL MONITORING OF AIDS. <u>AIDS surveillance</u> in Europe. Quarterly report No. 49, 31st March 1996.

EUROPEAN CENTRE FOR THE EPIDEMIOLOGICAL MONITORING OF AIDS. <u>AIDS surveillance</u> in Europe. Quarterly report No. 52, 31st December 1996.

EUROPEAN CENTRE FOR THE EPIDEMIOLOGICAL MONITORING OF AIDS. <u>HIV/AIDS</u> surveillance in Europe. Quarterly report No. 55, 30th September 1997.

EUROPEAN CENTRE FOR THE EPIDEMIOLOGICAL MONITORING OF AIDS. <u>HIV/AIDS</u> surveillance in Europe. Quarterly report No. 56, 31st December 1997.

EUROPEAN WORKING GROUP ON HIV INFECTION IN FEMALE PROSTITUTES (1993) HIV infection in European female sex workers: epidemiological link with use of petroleum-based lubricants. <u>AIDS</u> 7(3)401-408.

FARR, G., GABELNICK, H., STURGEN, K., DORFLINGER, L. (1994). Contraceptive efficacy and acceptability of the female condom. <u>Am J Public Health</u> 84(12):1960-1964.

FEINGOLD, A.R., VERMUND, S.H., BURK, R.D., KELLEY, K.F. et al. (1990). Cervical cytologic abnormalities and papillomavirus in women infected with human immunodeficiency virus. <u>J Acquir</u> <u>Immune Defic Syndr</u> 3(9):896-903

FENNEMA ,J.S.A., VAN AMEIJDEN, E.J.C., CONTINHO, R.A. et al. (1995). HIV, sexually transmitted diseases and gynaecological disorders in women: increased risk for genital herpes and warts among HIV-infected prostitutes in Amsterdam. <u>AIDS</u> 9: 1071-1078.

FIORENTINE, R., ANGLIN, M.D., GIL-RIVAS', V., & TAYLOR, E. (1997). Drug treatment: Explaining the gender paradox. Subst Use & Misuse 32(6):653-678.

FISHER, B., HOVELL, M., HOFSTETTER, C.R., & HOUGH, R. (1995). Risks associated with long-term homelessness among women: battery, rape, and HIV infection. Int J Health Services 25(2):351-359.

FRANCESCHI, S., TIRELLI, U., VACCHER, E. et al. (1988). Risk factors for HIV infection in drug addicts from the northeast of Italy. Int J Epidemiol. 17:162-167.

FREEMAN, R., RODRÍGUEZ, G. & FRENCH, J. (1994). A comparison of male and female intravenous drug users' risk behaviours for HIV infection. <u>Am J Drug Alcohol Abuse</u> 20(2):129-157.

FRIEDMAN, S., JOSE, B., NEAIGUS, A. et al. (1994). Consistent condom use in relationships between seropositive injecting drug users and sex partners who do not inject drugs. (Short communication). <u>AIDS</u> 8:357-361.

FRISCHER, M., HAW, S., BLOOR, M., GOLDBERG, D. et al. (1993). Modeling the behavior and attributes of injecting drug users: a new approach to identifying HIV risk practices. Int J Addictions, 28(2):129-152.

GAUGHWIN, M.D., DOUGLAS, R.M., WODAK, A.D. (1990). Behind bars - risk behaviours for HIV transmission in prisons, a review. Presented at the <u>HIV/AIDS and Prisons Conference</u>, Melbourne, Australia.

GLASS, R.M. (1993). Methadone maintenance. New research on a controversial treatment, JAMA 269:1995-1996.

GOEDERT, J., EYSTER, M., RAGNI, M. et al. (1988). Rate of heterosexual transmission and associated risk with HIV antigen Fourth International Conference on AIDS, (Abstract 4019).

GOLDSMITH, M.F. (1988). Sex tied to drugs = STD spread. JAMA 260:2009.

GOLDSTEIN, P.J. (1979). Prostitution and Drugs. Lexington: Lexington Books.

GORE, S.M., BIRD, A.G., BURNS, S.M., GOLDBERG, D.J. et al. (1995). Drug injection and HIV prevalence. <u>BMJ</u>, 310:293-296.

GORE, S.M., BIRD, A.G., BURNS, S.M., ROSS, A.J. & GOLDBERG, D.J. (1997). Anonymous HIV

surveillance with risk-factor elicitation: at Perth (for men) and Cornton Vale (for women) Prisons in Scotland. Int J STD & AIDS 8:166-175.

GORE ,S., GRAHAM, A. (1993). No escape: HIV transmission in jail. BMJ 307(17 July):147-148.

GOSSOP, M., POWIS, B., GRIFFITHS, P., & STRANG, J. (1995). Female prostitutes in south London: use of heroin, cocaine and alcohol, and their relationship to health risk behaviours. <u>AIDS Care</u> 7(3):253-260.

GREEN, S., GOLDBERG, D., NATHAWANI et al. (1990). Intercourse during menstruation among prostitutes. JAMA 264: 333.

GREENBERG, J., MAGDER, L., & ARAL, S. (1992) Age at first coitus: a marker for risky sexual behavior in women. <u>Sexually Transmitted Diseases</u>. 19(6): 331-334.

GUELAR, A., GATELL, J.M., VERDEJO, J. et al. (1993). A prospective study of tuberculosis among HIV infected patients. <u>AIDS</u> 7:1345-1349.

HAMERS, F.F., BATTER, V., DOWNS, A.M., ALIX, J. et al. (1997). The HIV epidemic associated with injecting drug use in Europe: geographic and time trends. <u>AIDS</u> 11:1365-1374.

HANKINS, C., GENDRON, S., ST. PIERRE, S. et al. (1989). Risk factors associated with HIV infection among women incarcerated in a medium security prison. <u>Fifth International Conference on AIDS</u>, Montreal, June (Abstract M.A.P.23:81).

HANKINS, C., GENDRON, S., HANDLEY, M., RICHARD, C. et al. (1994). HIV infection among women in prison: an assessment of risk factors using a nonnominal methodology. <u>Am J Public Health</u> 84(10):1637-1640.

HARDING, T. (1987). AIDS in prison. The Lancet, (November)1260-1263.

HARDING, T., MANGHI, R., SANCHEZ, G. (1990). HIV/AIDS and prisons: a survey covering 54 prison systems in 45 countries. <u>Report commssioned by the World Health Organization Global Programme on AIDS</u>, Geneva.

HARDING, T. & SCHALLER, G. (1992). HIV/AIDS and prisons: Updating and Policy review. <u>Report</u> for the World Health Organisation Global Programme on AIDS, Geneva.

HARRIS, R.E., LANGROD, J., HERBERT, J.R., ZANGE, E. & WYNDER, E.L. (1990). Changes in AIDS risk behaviour among intravenous drug abusers in New York City. <u>New York State J of Med</u> 90:123-126.

HARRISON, P.A. (1989). Women in treatment: Changing over time. Int J Addictions 24: 655-673.

HART, G.J., SONNEX, C., PETHERICK, A., JOHNSON, A.M. et al. (1989). Risk behaviours for HIV infection among injecting drug users attending a drug dependency clinic. <u>BMJ</u> 298: 1081-1083.

HEITJAN, D.F. (1997). Annotation: What can be done about missing data? Approaches to Imputation. <u>Am J Public Health</u> 87(4):548-550.

HELAL, H., MOMAS, I., PRÉTET, S., MARSAL, L. & POINSARD, R. (1995). HIV prevalence and risk behaviour among intravenous drug users attending HIV counselling and testing centres in Paris. <u>Addiction</u> 90:1627-1633.

HELLMANN, N., DESMOND-HELLMANN, S., NSUBIGA, P. et al. (1991). Genital trauma during sex is a risk factor for HIV infection in Uganda. <u>Seventh International Conference on AIDS, Florence, 1991</u> (Abstract M.C. 3079).

HENNEKENS, C.H. & BURING, J.E. (1987). Epidemiology in Medicine. Little, Brown & Company: Boston.

HERNÁNDEZ-AGUADO, I., & BOLUMAR, F. (1993). Determinants of HIV-1 Infection in Intravenous Drug Users in Valencia, Spain, 1987-1991. Int J Epidemiol 22(3):537-542.

HERNÁNDEZ-AGUADO, I., RUIZ, I., BOLUMAR, F., PÉREZ-HOYOS, S. et al. (1994). Sharing of injection equipment among 3755 intravenous drug users in Valencia, Spain, 1987-1992. <u>Int J Epidemiol</u> 23(3):602-607.

HODGINS, D.C., EL-GUEBALY, N. & ADDINGTON, J. (1997). Treatment of substance abusers: single or mixed gender programs? <u>Addiction</u> 9297):805-812.

HOLST, E., HOVELIUS, B., MARDH, P. (1987). Bacterial vaginosis: microbiology and clinical findings. Eur. J. Clin. Bacteriol. 6: 536-541.

HOUWELING, H., HAMERS, F.F., TERMORSHUIZEN, F., GILL, O.N. et al. (1998). A birth cohort analysis of AIDS in Europe: high incidence among young persons at risk. <u>AIDS</u> 12:85-93.

HUNTER, G.M., DONOGHOE, M.C., STIMSON, G.V., RHODES, T. & CHALMERS, C.P. (1995). Changes in the injecting risk behaviour of injecting drug users in London, 1990-1993. <u>AIDS</u> 9:493-501.

INSTITUTO DE LA MUJER (1992) La mujer en cifras. Ministerio de Asuntos Sociales, Madrid.

JAMES, J. & MEYERDING, J. (1977). Early sexual experience as a factor in prostitution. <u>Arch Sexual</u> <u>Behav</u> 7(1):31-42.

JOE, G.W. & SIMPSON, D.D. (1995). HIV risks, gender and cocaine use among opiate users. Drug & Alcohol Dependence 37:23-28.

JOHNSEN, L.W. & HARLOW, L.L. (1996). Childhood sexual abuse linked with adult substance use, victimization, and AIDS-risk. <u>AIDS Education & Prevention</u> 8(1):44-57.

JOHNSON, T.P., FREELS, S.A., PARSONS, J.A. & VANGEEST, J.B. (1997). Substance abuse and homelessness: social selection or social adaption? Addiction 92(4):437-445.

JOHNSTON, D., SMITH, K. & STALL, R. (1994). A comparison of public health care utilisation by gay men and intravenous drug users with AIDS in San Francisco. <u>AIDS Care</u> 6:303-316

JOHNSTONE, F.D., MACCALLUM, L., BRETTLE, R. et al. (1988). Does infection with HIV affect the outcome of pregnancy? <u>BM J</u> 296: 467.

KEENE, J. (1994). High-risk groups and prison policies. Int J Drug Pol 5(3):142-146.

KENDIG, N., STOUGH, T., AUSTIN, P., KUMMER, L. et al. (1994). Profile of HIV seropositive inmates diagnosed in Maryland's state correctional system. <u>Public Health Rep</u> (Nov.-Dec.) 109(6):756-760

KLEE, H. (1993). HIV risks for women drug injectors: heroin and amphetamine users compared. Addiction 88:1055-1062.

KLEE, H., FAUGIER, J., HAYES, C., BOULTON, T. & MORRIS, J. (1990). Factors associated with risk behaviour among injecting drug users, <u>AIDS Care</u>, 2, 133-154.

KLINE, A. & STRICKLER, J. (1993). Perceptions of risk for AIDS among women in drug treatment. Health Psychol 12(4):313-323.

KOWALEWSKI, M.R., KHALSA, H.K., ANGLIN, M.D. (1990). HIV risk behaviour among cocaine users, <u>Sixth International Conference on AIDS</u>, San Francisco, June 1990 [abstract SC 561].

KREISS, J., CARAEL, M., MEHEUS, A. (1988). Role of sexually transmitted diseases in transmitting human immunodeficiency virus [editorial]. <u>Genitourin Med</u> (Feb) 64(1):1-2.

LAMPINEN, T.M., JOO, E., SEWERYN, S., HERSHOW, C., WIEBEL, W. (1992). HIV seropositivity in community-recruited and drug treatment samples of injecting drug users; <u>AIDS</u> 6;123-126

LARDELLI, P., DE LA FUENTE, L., ALONSO, J.M., LÓPEZ, R. et al. (1993). Geographical variations in the prevalence of HIV infection among drug users receiving ambulatory treatment in Spain. Int J Epidemiol 22(2):306-314.

LEHMAN, J., ALLEN, D., GREEN, T. et al. (1994). HIV infection among non-injecting drug users entering drug treatment, United States, 1989-1992. <u>AIDS</u> 8(10):1465-1469.

LEWIS, D.K., WATTERS, J.K. (1991). Sexual risk behavior among heterosexual intravenous drug users: ethnic and gender variations. <u>AIDS</u> 5:77-83.

MADEN, A., SWINTON, M.. & GUNN, J. (1992). A survey of pre-arrest drug use in sentenced

prisoners. Brit J Addict 87:27-33.

MAGURA, S., SHAPIRO, J.L., KANG, S.Y. (1994). Condom use among criminally-involved adolescents. <u>AIDS Care</u> 6(5):595-603.

MARKOWITZ, N., HANSEN, N.I., HOPEWELL, P.C., GLASSROTH, J. et al. (1997). Incidence of tuberculosis in the United States among HIV-infected persons. <u>Ann Intern Med</u> 126:123-132.

MARMOR, M., DES JARLAIS, D.C., COHEN, H. et al. (1987). Risk factors of infection with Human Immunodeficiency Virus among intravenous drug users in New York City. <u>AIDS</u> 1:39-41.

MARTÍN, V., GONZALEZ, P., CAYLA, J. et al. (1994). Case-finding of pulmonary tuberculosis on admission to a penitentiary centre. <u>Tubercle and Lung Disease</u> 74:49-53.

MATI, J., MAGGWA, A., CHEWE, D. et al. (1990). Contraceptive use and HIV infection among women attending family clinics in Nairobi, Kenya <u>Sixth International Conference on AIDS</u>, San Francisco, [Abstract Th. C.99].

MCCOY, H.V. & INCIARDI, J.A. (1993). Women and AIDS: social determinants of sex-related activities. <u>Women & Health</u> 20(1):69-86.

MCCOY, H.V., INICIARDI, J.A., METSCH, L.R., POTTIEGER, A.E. & SAUM, C.A. (1995). Women, crack, and crime: gender comparisons of criminal activity among crack cocaine users. Federal Legal Publications, Inc.

MCCOY, H.V. & KHOURY, E. (1990). Drug use and risk of AIDS. Am Behav Sci 33:419-431.

MCELRATH, K., CHITWOOD, D.D., GRIFFIN, D.K., & COMERFORD, M. (1994). The consistency of self-reported HIV risk among injection drug users. <u>Am J Public Health</u> 84(12):1965-1970.

MCKEE, K.J. & POWER, K.G. (1992). HIV/AIDS in prisons. Scot Med J 37:132-137.

MCKEGANEY, N.P. et al. (1990). Injecting drug use and female street -working prostitution in Glasgow. AIDS 4: 1153-1155.

MCKEGANEY, N.P. (1990). Being positive: drug injectors' experiences of HIV infection. <u>Brit J</u> <u>Addiction</u> 85:1113-1124.

MCLELLAN, A.T., KUSHNER, H., METZGER, D., PETERS, R. et al. (1992). The fifth edition of the Addiction Severity Index: Historical critique and normative data. J Subst Abuse Treatment 9:199-213.

MELNICK, S., RENSLOW, S., LOUIS, T. et al. (1994). The Terry Beim Community Programs for Clinical Research on AIDS. Survival and disease progression according to gender of patients with HIV infection. JAMA 272(24):1915-1921.

MONDANARO, J. (1987). Strategies for AIDS prevention: Motivating health behaviour in drug

dependent women. J Psychoactive Drugs 19:143-149.

MOR, V., FLEISHMAN, J., DRESSER, M. et al. (1992). Variation in health services use among HIVinfected patients. <u>Med Care</u> (January) 30(1):17-29.

MORRISON, C.L., RUBEN, S.M. & BEECHING, N.J. (1995). Female sexual health problems in a drug dependency unit. Int J STD AIDS 6: 201-3.

MULLEADY, G., WITE, D., PHILIPS, K. & CUPITT, C. (1990). Reducing sexual transmission of HIV for injecting drug users: the challenge for counselling. <u>Counselling Psychol Q</u> 3:325-341.

MÜLLER, R., STARK, K., GUGGENMOOS-HOLZMANN, I. et al. (1995). Imprisonment: a risk factor for HIV infection counteracting education and prevention programmes for intravenous drug users. <u>AIDS</u> 9:183-190.

MURPHY, D. (1988). Heterosexual contacts of intravenous drug abusers: implications for the next spread of the AIDS epidemic. <u>AIDS and Substance Abuse</u> p. 89-97.

MUSICCO, M., NICOLOSI, A., COSTIGLIOLA, P. et al. (1994) The role of contraceptive practices in HIV sexual transmission among heterosexual couples. <u>Nicolosi, A., ed. HIV Epidemiology: Models and</u> <u>Methods. New York: Raven Press</u>, pp. 121-136.

NÁJERA, R. (1990). SIDA, de la biomedicina a la sociedad. Madrid: EUDEMA, S.A.,

NATIONAL PLAN ON DRUGS (1995) Ministerio de Sanidad y Consumo, Madrid.

NEAL, J.J., FLEMING, P.L., GREEN, T.A. & WARD, J.W. (1997). Trends in heterosexually acquired AIDS in the United States, 1988 through 1995. <u>J Acquir Immune Defic Syndr Human Retrovirol</u> 14:465-474.

NELSON, K., VLAHOV, D., COHN, S. et al. (1991). Sexually transmitted diseases in a population of intravenous drug users: association with seropositivity to the Human Immunodeficiency Virus (HIV). J Infectious Diseases 164 (September):457-463.

NICOLOSI, A., CORRÊA LEITE, M.L., MOLINARI, S., MUSICCO, M. et al. (1992a). Incidence and prevalence of HIV infection in intravenous drug users attending treatment centers in Milan and northern Italy, 1986-1990. J Acquir Immune Defic Syndr. 5(4):365-373.

NICOLOSI, A., CORRÊA LEITE, M.L., MUSICCO, M. et al. (1992b). Parenteral and sexual transmission of Human Immunodeficiency Virus in intravenous drug users: a study of seroconversion. <u>Am J</u> <u>Epidemiol</u> 135(3):225-233.

NICOLOSI A., CORRÊA LEITE M.L., MUSICCO M. et al.(1994a). The efficiency of male-to-female and female-to-male sexual transmission of the Human Immunodeficiency Virus: a study of 730 stable couples. <u>Epidemiol</u> 5(6):570-575.

NICOLOSI. A., MUSICCO, M., SARACCO, A., & LAZZARIN, A. FOR THE ITALIAN STUDY GROUP ON HIV HETEROSEXUAL TRANSMISSION. (1994b). Risk factors for women-to-man sexual transmission of the human immunodeficiency virus. <u>J Acquir Immune Defic Syndr</u> 7(3): 296-300.

NYAMATHI, A., FLASKERUD, J., & LEAKE, B. (1997). HIV-risk behaviors and mental health characteristics among homeless or drug-recovering women and their closest sources of social support. Nursing Research 46(3):133-137.

PAONE, D., CALOIR, S., SHI, Q., DES JARLAIS, D.C. (1995). Sex, drugs, and syringe exchange in New York City: women's experiences. J Am Med Women's Assoc (May-Aug) 50(3-4):109-114.

PAPAEVANGELOU, G., ROUMELIOTOU, A., STERGIOY, G. et al. (1991). HIV infection in Greek intravenous drug users. <u>Eur J Epidemiol</u> 7(1):88-90.

PICKERING, H. (1988). Asking questions on sexual behaviour...testing methods from the social sciences. Health Policy and Planning, 3(3):237-244.

PLUMMER, F., SIMONSEN, J., CAMERON, W. et al. (1991). Cofactors in male-female transmission of human immunodeficiency virus type 1. J Infect Dis 339: 785-788.

POLK, B. (1985). Female-to-male transmission of AIDS (Letter). JAMA 254: 3177.

POLONSKY, S., KERR, S., HARRIS, B., GAITER, J. et al. (1994) HIV prevention in prisons and jails: obstacles and opportunities. <u>Public Health Rep</u> (Sept-Oct) 109(5):615-625.

POWER, K., MARKOVA, I., ROWLANDS, A. et al. Intravenous drug use and HIV transmission amongst inmates in Scottish prisons. <u>Brit J Addiction</u> (1992) 87:35-45.

POWIS, B., GRIFFITHS, P., GOSSOP, M. & STRANG, J. (1996). The differences between male and female drug users: Community Samples of Heroin and Cocaine Users Compared. <u>Substance Use & Misuse</u> 31(5): 529-543.

PUBLIC HEALTH LABORATORY SERVICE (PHLS) (1991) AIDS and HIV1 antibody reports--United Kingdom. Communicable Disease Report (April) 1:15.

RALPH, N. & SPIGNER, C. (1986). Contraception practices among female heroin addicts. <u>Am J Public</u> <u>Health</u> 76:1016-7.

RAY, S., BASSETT, M., MAPOSHERE, C., MANANGAZIRA, P. et al. (1995). Acceptability of the Female Condom in Zimbabwe: positive but male-centered responses. <u>Reprod Health Matters</u> (May) 5:68-79.

RHODES, T., DONOGHOE, M., HUNTER, G. et al. (1993). Continued risk behaviour among positive drug injectors in London: implications for intervention. <u>Addiction</u> 88:1552-1560.

RHODES, T., DONOGHOE, M., HUNTER, G. et al. (1994) Sexual behaviour of drug injectors in London: implications for HIV transmission and HIV prevention. <u>Addiction</u> 89:1085-1096.

RICHARDSON, S., PAPAEVANGELOU, G. & ANCELLE-PARK, R. (1994). Knowledge, attitudes and beliefs of European injecting drug users concerning preventive measures for HIV. <u>Eur J Epidemiol</u> 10: 135-142.

RODDY, R.E. (1994). Four family health international studies on barrier methods and sexually transmitted infections. In: Barrier contraceptives and current status and future prospects. <u>Proceedings of the 4th contraceptive research and development program international workshop (Edited by Mauck et al.)</u>. New York: Wiley Liss, p. 245-248.

ROSS, M.W., GOLD, J., WODAK, A., MILLER, M.E. (1991). Sexually transmissible diseases in injecting drug users . <u>Genitourin Med</u> 67:32-36

ROSS, M.W., STOWE, A., WODAK, A., MILLER, M.E. & GOLD, J. (1993). A comparison of drug use and HIV infection risk behavior between injecting drug users currently in treatment, previously in treatment, and never in treatment. J Acquir Immune Defic Syndr 6(5):518-528.

RUIZ, M.T. & VERBRUGGE, L.M. (1997). A two way view of gender bias in medicine. <u>J Epidemiol &</u> <u>Comm Health</u> 51:106-109.

SALMASO, S., CONTI, S., SASSE, H. et al. (1991), Drug use and HIV-1 infection: report from the second Italian multicentre study. <u>J Acquir Immune Defic Syndr</u> 4(6):607-613.

SASSE, H., SALMASO, S., CONTI, S. et al. (1989). Risk behaviour for HIV-1 infection in Italian drug users: report from a multicentre study. <u>J Acquir Immune Defic Syndr</u> Vol. 2, No. 5, p. 486-496.

SAXON, A,J,, CALSYN, D,A,, WHITTAKER, S,, FREEMAN, G. (1991). Sexual behaviors of intravenous drug users in treatment. J Acquir Immune Defic Syndr 4:938-944.

SCHERING ESPAÑA (1997). Encuesta Schering anticoncepción en España 1997 - uso de anticonceptivos en España: April 1997.

SCHOENBAUM, E., HARTEL, D., SELWYN, P. et al. (1989). Risk factors for Human Immunodeficiency Virus infection in intravenous drug users. <u>New Engl J Med</u> 321(13):874-879.

SEAMARK, R. & GAUGHWIN, M. (1994). Jabs in the dark: injecting equipment found in prisons, and the risks of viral transmission. <u>Austral J Public Health</u> 18(1):113-116.

SERRAINO, D., FRANCESCHI, S., VACCHER, E. et al. (1991). Risk factors for Human Immunodeficiency Virus infection in 581 intravenous drug users, Northeast Italy, 1984-1988. <u>Int J</u> Epidemiol 20(1):264-270. SERRAINO, D., NAPOLI, P.A., ZACCARELLI, M., ALLIEGRO, M.B., PEZZOTTI, P. & REZZA, G. (1996). High frequency of invasive cervical cancer among female injecting drug users with AIDS in Italy. <u>AIDS</u> 10:1041-1055.

SHERR, L. (1991). Women and children. AIDS Care 3(4):423-432.

SMYTH, B., KEENAN, E., DORMAN, A, & O'CONNOR, J.J. (1994). Gender differences in needle sharing behaviour patterns. <u>Addiction</u> 89(1):96-97.

SIMONSEN, J., PLUMMER, F. & NGUGI, E. (1990). HIV infection among lower socioeconomic strata prostitutes in Nairobi. <u>AIDS</u> 4; 139-144.

SOLOMON, L., ASTEMBORSKI, J., WARREN, D., MUÑOZ, A. et al. (1993). Differences in risk factors for human immunodeficiency virus type 1 seroconversion among male and female intravenous drug users. <u>Am J Epidemiol</u> 137(8):892-898.

SOPELANA, P., DIÉGUEZ, A. & BAUTISTA, L. (1991). Prevalencia de la infección por el virus de la inmunodeficiencia humana tipo 1 en drogodependientes de Madrid, durante un periódo de 5 años. <u>Med Clín</u> (<u>Barc</u>) 97(16):601-603.

SOTHERAN, J., WENSTON, J., ROCKWELL, R. et al. (1992). Gender differences in the social context of syringe sharing among New York IDUs. <u>Eighth International Conference on AIDS</u>, 19-24 July 1992, Amsterdam.

SPINELLI, A., GRANDOLFO, M., DONATI, S., MEDDA, E. (1993). Family planning in Italy. <u>Adv</u> <u>Contracept</u> 9(2): 153-160.

SPIRA, A. (1993). Sexual behaviour [letter]. BMJ (Feb. 27) 306(6877):583.

STALL, R. & LEIGH, B. (1994). Understanding the relationship between drug or alcohol use and high risk sexual activity for HIV transmission: where do we go from here? [editorial] Addiction 89(2):131-134.

STARK, K. & MÜLLER, R. (1993). HIV prevalence and risk behaviour in injecting drug users in Berlin. Forensic Sciences International 62:73-81.

STARK, K., MÜLLER, R., GUGGENMOOS-HOLZMANN, I. et al. (1990). HIV infection in intravenous drug abusers in Berlin: risk factors and time trends. <u>Klin Wochenschr</u> 68:415-420.

STATISTICHE DELLA SANITA (1995). Ministerio della Sanita, Rome, Italy.

STEPHENSON, J.M., GRIFFIOEN, A. and the Study Group for the Medical Research Council Collaborative Study of Women with HIV. (1996). The effect of HIV diagnosis on reproductive experience. <u>AIDS</u> 10:1683-1687.

STERK, C. (1988). Cocaine and HIV seropositivity [letter]. Lancet (May) 1052-1053.

STIMSON, G. (1992). Drug injecting and HIV infection: New directions for social science research. Int J Addictions, 27(2):147-163.

STIMSON, G.V., HUNTER, G.M., DONOGHOE, M.C., RHODES, T. et al. (1996). HIV-1 prevalence in community-wide samples of injecting drug users in London, 1990-1993. <u>AIDS</u> 10:657-666.

STRANG, J., POWIS, B., GRIFFITHS, P. et al. (1994). Heterosexual vaginal and anal intercourse amongst London heroin and cocaine users. (Short paper). Int J STD AIDS (March/April 5(368):1-4.

STUART, A.B., O'ROURKE, S. & POWER, K.G. (1997). Regional variations in pre-imprisonment drug use among female inmates of a Scottish prison. Addiction Res 5(2):83-94.

SYMONDS, E.M. (1987). Essential obstetrics and gynaecology. Churchill Livingstone: Edinburgh.66 Brown BS, 1989. Presentation of NADR national data. In: <u>Proceedings of the conference on AIDS</u> intervention strategies for female sexual partners. Berkeley, California, June.

TAYLOR, A., GOLDBERG, D., EMSLIE, J. et al. (1995). Outbreak of HIV infection in a Scottish prison. BMJ (4 February):289-292.

TIRELLI, U., VACCHEV, E., CARBONE, A., DEPAELI, P. et al. (1986). Heterosexual contact is not the predominant mode of HTLV-III transmission among drug users. JAMA. 255:2289 (letter).

TOMASEVSKI, K. (1991). AIDS and prisons. AIDS 5:S245-S251.

TURNBULL, P., DOLAN, K., STIMSON, G. (1991). <u>Prisons, HIV and AIDS: Risks and experiences in</u> custodial care. Horsham, England: AVERT(The AIDS education and Research Trust).

TURNBULL, P.J., STIMSON, G.V., DOLAN, K.A. (1992). Prevalence of HIV infection among exprisoners in England. <u>BMJ</u>, 304:90-91.

UN, AIDS, AND WHO, GENEVA. The HIV/AIDS situation in mid 1996. Fact sheet 1, July 1996.

UN, AIDS, AND WHO WORKING GROUP ON THE WORLD SURVEILLANCE OF HIV/AIDS AND STDS. <u>Report on the world epidemic of HIV/AIDS</u>. December 1997.

USIETO, R. (1992). Conductas sexuales de riesgo: sociogenesis heterosexual de SIDA en España. <u>SIDA:</u> <u>Transmisión Heterosexual del VIH</u>. Centro de Estudios Sociales Aplicados, p. 103.

VAN AMEIJDEN, E., VAN DEN HOEK, A., VAN HAASTRECHT, H. et al. (1994). Trends in sexual behaviour and the incidence of sexually transmitted diseases and HIV among drug-using prostitutes, Amsterdam 1986-1992. <u>AIDS</u> 8(2):213-221.

VAN DEN HOEK, J.A.R., VAN HAASTRECHT, H.J.A., COUTINHO, R.A. (1990). Heterosexual behaviour of intravenous drug users in Amsterdam: implications for the AIDS epidemic. <u>AIDS</u> 4:449-453.

VAN DEN HOEK, J.A.R., VAN HAASTRECHT, H.J.A., COUTINHO, R.A. (1992). Little change in sexual behavior in injecting drug users in Amsterdam. J Acquir Immune Defic Syndr 5(5):518-522.

VAN DEN HOEK, J.A.R., VAN HAASTRECHT, H.J.A., SCHERINGA-TROOST, B. et al. (1989). HIV infection and STD in drug-addicted prostitutes in Amsterdam: potential for heterosexual HIV transmission. <u>Genitourin Med</u>. 65:146-150.

VAN DEN HOEK, J.A.R., COUTINHO, R.A., VAN HAASTRECHT, H.J.A., ZADELHOFF, A.W. et al. (1988). Prevalence and risk factors of HIV infection among drug users and drug-using prostitutes in Amsterdam. <u>AIDS</u>, 2:55-60.

VLAHOV, D., BREWER, F., CASTRO, K. et al.(1991). Prevalence of antibody to HIV-1 among entrants to US correctional facilities. JAMA 265(9):1129-1132.

VLAHOV, D., MUÑOZ, A., ANTHONY, J. et al. (1990). Association of drug injection patterns with antibody to Human Immunodeficiency Virus type 1 among intravenous drug users in Baltimore, Maryland. <u>Am J Epidemiol</u> 132(5):847-856.

VLAHOV, D., NELSON, K.E., QUINN, T.C. et al. (1993). Prevalence and incidence of hepatitis C virus infection among male prison inmates in Maryland. <u>Eur J Epidemiol</u> 9(5):566-569.

VOELLER, B. & ANDERSON, D. (1992). Heterosexual transmission of HIV (Letter). JAMA 267:1917-1918.

WARD, H., DAY, S., MEZZONE, J., DUNLOP, L. et al. (1993). Prostitution and risk of HIV: female prostitutes in London. <u>BMJ</u> 307:356-358.

WATTERS, J.K. (1994a). Trends in risk behavior and HIV seroprevalence in heterosexual injection drug users in San Francisco, 1986-1992. J Acquir Immune Defic Syndr 7(12):1276-1281.

WATTERS, J., ESTILO, M., KRAL, A. et al. (1994b). HIV infection among female injection-drug users recruited in community settings. <u>Sexually Transmitted Diseases</u>. (November-December) 21(6):321-328.

WEISNER, C. & SCHMIDT, L. (1992). Gender disparities in treatment for alcohol problems. JAMA 268:1872-1876.

WEISNER, C. & SCHMIDT, L. (1993). Alcohol and drug problems among diverse health and social service populations. <u>Am J Public Health</u> 83(6):824-829.

WEISSMAN, G., MELCHIOR, L., HUBA, G., ALTICE, F. et al. (1995). Women living with substance abuse and HIV disease: medical care access issues. J Am Med Women's Assoc 50(3-4):115-120.

WELLINGS, K., FIELD, J., JOHNSON, A.M. & WADSWORTH, J. (1994). <u>Sexual behaviour in</u> Britain. The National Survey of Sexual Attitudes and Lifestyles. London: Penguin Books, p 330.

WELLS, D.V.B. & JACKSON, J.F. (1992). HIV and chemically dependent women: Recommendations for appropriate health care services and drug services. Int J Addictions, 27, 571-85.

WHO (1993). WHO urges prison reform to help control the spread of HIV. IXth International Conference on AIDS, Berlin, 10 June.(press release)

WHO COLLABORATIVE STUDY GROUP. An international comparative study of HIV prevalence and risk behaviour among drug injectors in 13 cities. <u>Bulletin on Narcotics</u> (1993); Vol. XLV, No. 1, p. 19-46.

WILLIAMSON, N.E. & JOANIS, C. (1994). Acceptability of barrier methods for prevention of unwanted pregnancy and infection. In: Barrier contraceptives and current status and future prospects. <u>Proceedings of the 4th contraceptive research and development program international workshop (Edited by</u> Mauck et al.). New York: Wiley Liss, p. 53-67.

WILSON, T.E., MINKOFF, H., DEHOVITZ, J., FELDMAN, J. & LANDESMAN, S. (1998). The relationship of cocaine use and human immunodeficiency virus serostatus to incident sexually transmitted diseases among women. <u>Sexually Transmitted Diseases</u> 25(2):70-74.

ZANGERLE, R., FUCHS, D., RÖSSLER, H. et al. (1992). Trends in HIV infection among intravenous drug users in Innsbruck, Austria. J Acquir Immune Defic Syndr 5(9):865-871.

ZIERLER, S., WITBECK, B., MAYER, K. (1996). Sexual violence against women living with or at risk for HIV infection. <u>Am J Prev Med</u> 12(5):304-310.

ZUNZUNEGUI, V., CASABONA, J., LAGUNA, J. et al. (1992). Factores de riesgo para la transmisión heterosexual del VIH de varón a mujer: un estudio multicéntrico español. <u>Med Clín (Barc)</u> 98(19):721-725.

ZUNZUNEGUI, M.V., RODRÍGUEZ, M.A. & SARASQUETA, C. (1993). Drogadicción intravenosa riesgo de infección por VIH en Madrid 1990. Gac Sanit (Enero-Febrero) 7(34):2-11.